The Determinants of Public Service and Environmental Performance of Local Governments in Indonesia

Muhammad Arif

This thesis is presented for the degree of Doctor of Philosophy College of Arts, Business, Law and Social Sciences Murdoch University

Declaration

I, Muhammad Arif, declare that this thesis is my own account of my research and contains as its main content work that has not previously been submitted for a degree at any tertiary education institution.

Name: Muhammad Arif

Abstract

Local governments have significant roles to play with respect to providing highquality public services and maintaining environmental sustainability. However, local communities in developing countries, including Indonesia, still have less access than those in developed countries to effective public services; and are facing increasing risks from environmental degradation. Given these issues for developing countries, this study focuses on the performance of the Indonesian public sector with respect to both the provision of public services and environmental sustainability.

This study has two objectives. First, it aims to examine the influences of public sector governance and financial management factors on Indonesian districts' service performance, mediated by financial performance. Second, this study aims to investigate the impacts of public sector governance and financial management factors on districts' environmental performance through the mediating role of financial performance. Indonesia was chosen as an exemplar of developing countries. In this study, public sector governance is represented by financial transparency, accountability, and internal control, while financial management factors are measured by financial flexibility, financial independence, and resource management. Several conceptual models are developed, from which the theoretical relationship between public sector governance, financial management factors, and service and environmental performance are hypothesised. These hypotheses are then tested using a quantitative approach and secondary data. The main findings of this study indicate that public sector governance and financial management factors are positively related to service and environmental performance. The study also confirms the mediation effect of financial performance on service and environmental

iii

performance. This study enriches the literature related to the public sectors since it provides incremental evidence regarding the determinant factors of service and environmental performance and develops the theoretical models that investigate the mediating role of financial performance on the performance of local government in a developing country like Indonesia. Future studies in this area could include investigating the effects of other potentially relevant measures, such as innovation and managerial styles.

Acknowledgements

All praise belongs to Allah, Lord of all the worlds. The long my PhD journey comes to an end. I would like to express my deepest gratitude to my principal supervisor, Professor Manzurul Alam, and my co-supervisor, Dr Terri Trireksani, for their invaluable guidance, advice, and encouragement throughout the period of my PhD study. My sincere thanks also go to my postgraduate research chair, Dr Megan Paull, for her support.

I would like to thank the Australia Awards Scholarships and the Ministry of Finance, the Republic of Indonesia for providing me the priceless opportunity to pursue this research degree at Murdoch University. I am also grateful to Australia Awards – Student Contact Officer, Leslie Theaker, for her support. Many thanks to Brother Mahbub for his valuable help, feedback, and generosity. I would like to thank Bec Donaldson for proofreading and editing my thesis. My sincere thank also goes to Mbak Yuditha, Brother Amer, Brother Krisna, and my fellow PhD students in the Murdoch Business School for their suggestions and support.

My special thanks go to my beloved mother Hajjah Eem Halimah, my mother-in-law Hajjah Yayat Maryati, my wife Sri Gustini, and my children Alia Nailatul Kiyasah, Nisrina Indah Taqiyyah, and Muhammad Fatih Najmulhaq. Their prayers, endless love, understanding, and patience have become huge energy for me to complete my PhD course. To my father Haji Abidin and my father-in-law Haji Sukamajaya who could not see the culmination of my study completed, your love and inspiration taught me to achieve my life goals in this world and hereafter.

Table of Contents

Dec	clarati	วท	ii
Abs	stract		iii
Acl	knowle	edgements	v
Tak	ole of (Contents	vi
Lis	t of Fig	jures	x
Lis	t of Ta	bles	xi
Cha	apter 1	Introduction	1
1.	. Res	earch Background	1
2.	. Gap	s in the Existing Research	3
3.	. Res	earch Problems	7
4.	. Sco	be of the Study	11
5.	. Res	earch Objectives	12
6.	. Res	earch Questions	13
7.	. Orga	inisation of the Thesis	
Cha	apter	2 Literature Review, Theoretical Framework, and Hypo	otheses
Dev	velopn	1ent	17
1.	. Intro	duction	17
2.	. Serv	ice Provision and Protection of Environment	
3. G	. Dete		
	overnn	rminants of Public Service and Environmental Performance of Loc	
	overnn 3.1	erminants of Public Service and Environmental Performance of Loc nents Public Sector Governance	
	overnn 3.1 3.1. ²	erminants of Public Service and Environmental Performance of Loc nents Public Sector Governance Financial Transparency	
	3.1 3.1. 3.1.2 3.1.2	erminants of Public Service and Environmental Performance of Loc nents Public Sector Governance Financial Transparency Accountability	20 21 24 25
	3.1 3.1. 3.1.2 3.1.2 3.1.3	erminants of Public Service and Environmental Performance of Loc nents Public Sector Governance Financial Transparency Accountability Internal Control	20 21 24 25 27
	3.1 3.1.2 3.1.2 3.1.3 3.2	erminants of Public Service and Environmental Performance of Loc nents Public Sector Governance Financial Transparency Accountability Internal Control Financial Management Factors	20 21 24 25 27 31
	3.1 3.1.2 3.1.2 3.1.3 3.2 3.2.2	erminants of Public Service and Environmental Performance of Loc nents Public Sector Governance Financial Transparency Accountability Internal Control Financial Management Factors Financial Flexibility	20 21 24 25 27 31 33
	3.1 3.1.2 3.1.2 3.1.3 3.2 3.2.2 3.2.2	erminants of Public Service and Environmental Performance of Loc nents Public Sector Governance Financial Transparency Accountability Internal Control Financial Management Factors Financial Flexibility Financial Independence	20 21 24 25 27 31 33 34
	3.1 3.1.2 3.1.2 3.1.3 3.2 3.2.2 3.2.2 3.2.2	erminants of Public Service and Environmental Performance of Loc nents Public Sector Governance Financial Transparency Accountability Internal Control Financial Management Factors Financial Flexibility Financial Flexibility Resource Management	20 21 24 25 27 31 33 34 35
	3.1 3.1.2 3.1.2 3.1.3 3.2.2 3.2.2 3.2.2 3.3	erminants of Public Service and Environmental Performance of Loc nents	20 21 24 25 27 31 33 34 34 35 36
	3.1 3.1.2 3.1.2 3.1.3 3.2.2 3.2.2 3.2.2 3.2.3 3.2.3 3.4	 rminants of Public Service and Environmental Performance of Loc nents Public Sector Governance Financial Transparency Accountability Internal Control Internal Control Financial Management Factors Financial Flexibility Financial Independence Resource Management Vediating Role of Financial Performance Control Variables 	20 21 24 25 27 31 33 34 35 36 38

4.1	l The	e Link Between Institutional Theory and the Attributes of the S	tudy 40
4.2	2 The	e Conceptualisation of Institutions	43
5. H	Hypothe	eses Development	
5.1 Go	l Hyp overnan	ootheses Related to the Relationship between Publi ice and Local Government's Performance	c Sector 48
5.2 Fa	2 Hyp ctors a	ootheses Related to the Relationship between Financial Ma nd Local Government's Performance	nagement 53
5.3 En	3 The vironm	e Mediating Role of Financial Performance on Serrental Performance	vice and 56
6. (Conclus	sion	59
Chapt	er 3 In	donesian Local Governments	61
1. I	ntroduc	ction	61
2. F	Public S	Sector Reforms at the District Level in Indonesia	63
2.1	I Loc	al Government Decentralisation	66
2.2	2 Org	anisational Structure of a Local Government	67
2.3	B Pub	blic Sector Governance of Districts	69
2.4	4 Fina	ancial Management of Districts	70
3. 1	The Sei	rvice Performance of Districts	73
4. E	Environ	mental Performance of Districts	75
5. (Conclus	sion	78
Chapt	er 4 R	esearch Methodology	80
1. I	ntroduc	ction	80
2. F	Resear	ch Methods	82
2.1	l Dat	a Sources	
2.2	2 Res	search Samples	
2.3	B Dat	a Analysis	88
	2.3.1	Testing Validity and Reliability	
2	2.3.2	Descriptive Statistics and Correlation Tests	
2	2.3.3	Multiple Linear Regression and Bootstrapping Analysis	
2.4	1 Ass	sumptions of the OLS regression and Goodness of Fit of the M	1odel 93
2	2.4.1	Multicollinearity Test	
2	2.4.2	Autocorrelation Test	
2	2.4.3	Tests of Linearity and Homoscedasticity	
2	2.4.4	Test of Normality of Residuals	
2	2.4.5	Assessing the Goodness of Fit of the Model	

3.	Rob	oustness Checks	96
4.	Defi	initions and Measurements of Variables	98
4	.1	Developing Composite Indices	98
4	.2	Service Performance 1	00
4	.3	Environmental Performance1	103
4	.4	Financial Performance1	07
4	.5	Financial Flexibility1	110
4	.6	Financial Independence 1	111
4	.7	Resource Management1	112
4	.8	Financial Transparency1	14
4	.9	Internal Control 1	117
4	.10	Accountability1	121
5.	Con	nclusion1	23
Chap	oter	5 Service Performance of Local Governments: Data Analys	sis
and	Нур	otheses Testing1	24
1.	Intro	oduction1	124
2.	Data	a Preparation and Analysis1	126
3.	Des	criptive Statistics and Correlations 1	27
4.	The	Impacts of Public Sector Governance 1	130
5.	The	Impacts of Financial Management Factors1	132
6.	The	Mediating Role of Financial Performance on Local Government's Service	ce
7	Doh		130
7.	ROD		140
8.	Con		140
Cnaj Test	oter ina	6 Local Government's Environmental Performance: Hypothes	48 48
1	Intro	aduction	148
2.	Des	criptive Statistics and Correlations	149
3.	Mul	tiple Regression Analysis	150
3	5.1	Effects of Public Sector Governance on Environmental Performance 1	150
3	.2	Effects of Financial Management on Environmental Performance	152
4.	The	Mediating Role of Financial Performance on Local Government's	
En	vironi	mental Performance1	154
5.	Rob	oustness Checks1	158
6.	Con	nclusion1	62

Chapter 7 Discussion and Conclusion163		
1.	Introduction1	63
2.	Service Performance of Local Government1	64
3.	Financial Performance of Local Government1	70
4.	Mediating Role of Financial Performance on Service Performance	76
5.	Environmental Performance of Local Government1	78
6.	Mediating Role of Financial Performance on Environmental Performance 1	82
7.	Theoretical and Practical Contributions of the Study1	86
8.	Limitations of the Study 1	91
9.	Recommendations for Future Research1	92
10.	Conclusions1	93
References198		
Appendix243		

List of Figures

Figure 1.1: The Integrated Construct of Chapters 14
Figure 2.1: The Integration of Chapter 2 with the Research Objectives
Figure 2.2: Theoretical Framework of Public Sector Governance Factors Affecting Local Government Performance
Figure 2.3: Theoretical Framework of Financial Management Factors Affecting Local Government Performance
Figure 3.1: The Integration of Chapter 3 with the Research Objectives
Figure 3.2: Level of Government in the Republic of Indonesia
Figure 3.3: The Impacts of Public Sector Reforms on Districts in Indonesia 66
Figure 3.4: Organisational Structure of a District68
Figure 4.1: The Integration of Chapter 4 with the Research Objectives
Figure 4.2: Structure of the Indonesian Local Governments
Figure 5.1: Structure of Chapter 5 124
Figure 5.2: The Linkage of Chapter 5 with the Research Objectives 125
Figure 6.1: The Linkage of Chapter 6 with the Research Objectives

List of Tables

Table 2.1: Determinants of Public Service and Environmental Performance 21
Table 2.2: The Three Pillars of Institutions 47
Table 2.3: Summary of Research Hypotheses 59
Table 4.1: The Number of Samples Used in the Study
Table 4.2: Models and Equations
Table 4.3: Methods of Measurement
Table 4.4: Dimensions of the Financial Performance of Local Governments 108
Table 4.5: Indicators of Financial Transparency
Table 4.6: The Measurements of Internal Control 119
Table 4.7: Administrative Accountability of Local Government in Indonesia 122
Table 5.1: Descriptive Statistics and Correlations 128
Table 5.2: Multiple Regression Analysis of Public Sector Governance on Financialand Service Performance131
Table 5.3: Multiple Regression Analysis of Financial Management Factors onService and Financial Performance133
Table 5.4: Mediating Role of Financial Performance on Service Performance (B,95% CI). Standardised Coefficients Results
Table 5.5: Stepwise Multiple Regression Analysis of Public Sector Governance onService and Financial Performance141
Table 5.6: Stepwise Multiple Regression Analysis of Financial Management Factorson Service and Financial Performance
Table 5.7: Monte Carlo Simulation of the Indirect Effects of Financial Performance on Service Performance 146
Table 6.1: Pearson's Correlation Matrix
Table 6.2: The Effects of Public Sector Governance on Environmental Performance

Table 6.3: Multiple Regression Analysis of Financial Management Factors on			
Environmental Performance153			
Table 6.4: Mediating Role of Financial Performance on Environmental Performance			
(β, 95% CI). Standardised Coefficients Results			
Table 6.5: Stepwise Multiple Regression Analysis of Public Sector Governance on			
Environmental Performance159			
Table 6.6: Stepwise Multiple Regression Analysis of Financial Management Factors			
on Environmental Performance			
Table 6.7: Monte Carlo Simulation of the Indirect Effects of Financial Performance on			
Environmental Performance161			

Chapter 1 Introduction

1. Research Background

Public services delivered by the local governments aim to meet a wide range of needs and expectations in communities, such as local roads, firefighting, waste management, health services, and water supply (Chattopadhyay 2013; Maksimovska and Stojkov 2016; Dick-Sagoe 2020). In recent years, subnational governments have focused not only on public service provision but also on environmental protection (Štreimikienė 2015). As mandated by Chapter 28 of Agenda 21 of the United Nations, which relates to sustainable development, sub-national governments of all countries have a pivotal role in protecting the biosphere (Williams, Wilmshurst, and Clift 2011; Wiharani and Holzhacker 2016). However, public services and environmental protection efforts by sub-national governments face many obstacles, especially in developing countries; including due to increasing human populations, lack of local government funding, and rising demands on forest areas (Hutton and Chase 2016; Chong et al. 2016; Leblois, Damette, and Wolfersberger 2017; Arshad et al. 2020). To support sub-national governments to implement effective services, including environmental sustainability, a clear understanding is required of the factors that can affect service and environmental performance for local governments to improve their performance.

Previous studies have found that public sector governance factors, such as transparency and accountability, are essential factors that influence organisational performance, including for local governments (Williams 2015; Kis-Katos and Sjahrir 2017; Han and Hong 2019). Transparency and accountability improve a local

government's outcomes (Ott, Bronić, and Petrušić 2014) and preserves its legitimacy since people have more power to monitor its activities (Benito and Bastida 2009). In developed countries, transparency and accountability are factors that have been studied most commonly with respect to their influence on local government's service and environmental performance (Halkos and Tzeremes 2014; Sahide et al. 2016; Han and Hong 2019). Other commonly studied determinants of effective local government service and environmental performance include internal control (Nuswantara and Maulidi 2017; Ong'unya and Abbey 2019; Kareem, AlHassani, and Ali 2019; Huang and Huang 2020), and financial management factors (Kioko et al. 2011; Kis-Katos and Sjahrir 2017; García-Sánchez, Frías-Aceituno, and Rodríguez-Domínguez 2013).

The public sector governance and financial management factors are expected to have an effect on service and environmental performance in different conditions, especially in the local government context in developing countries with different characteristics in terms of socio-demographic and economic factors, local government size, function, revenue, and expenditure structure (Shah and Shah 2006; Rasool and Ogunbode 2015; UN-HABITAT 2015; Lopez Subires et al. 2019; Pina, Bachiller, and Ripoll 2020; Sinervo 2020).

In developing countries, local governments face problems of inefficient practices of financial management (Ningsih, Kusumawati, and Nohong 2019). Such inefficient practices are usually addressed by having sound internal control systems; which are more likely to be found in developed countries (Muskanan 2014; Tackie, Marfo-Yiadom, and Achina 2016; Asaolu, Adedokun, and Monday 2016; Udeh and Nwadialor 2016; Sorunke 2016; Kewo 2017; Ayagre and Aidoo-Buameh 2017;

Gamayuni 2018; Ujkani and Vokshi 2019). Accordingly, local governments in developing countries need to focus on building effective internal control mechanisms to alleviate budget mismanagement (Donelson, Ege, and McInnis 2017; Nuswantara and Maulidi 2017).

2. Gaps in the Existing Research

The existing literature on service and environmental performance by local governments in developing countries is still limited for a number of reasons. First, prior studies focused mainly on producing models to investigate service performance in the context of developed countries (Boyne 2003; Van de Walle 2006; Torres, Pina, and Yetano 2011); and measured environmental performance in the context of the private sector (Ball and Bebbington 2008; Yin and Schmeidler 2009; Khanna and Kumar 2011; Iwata and Okada 2011; Lodhia, Jacobs, and Park 2012; Darnall and Kim 2012; Fujii and Managi 2016; Arimura et al. 2016). Research on local government's service and environmental performance is relatively scant in developing countries, where the local government faces different contextual and institutional environments than in developed countries (Andrews 2012; Liu et al. 2017). Given this gap, the present study's empirical evidence will provide important insights into the application of service and environmental performance in developing countries. This study expects that the empirical results will be similar compared to previous studies conducted in developed countries (Torres, Pina, and Yetano 2011; Kaufmann and Lafarre 2020) because this study focuses on a developing country that has conducted public sector reforms: governance and financial management reform.

There are several examples of different contextual and institutional environments of local government in developing countries than in developed countries. For example, local communities in developing countries have less access than those in developed countries to public health services, such as clean water and hygiene sanitation (O'Donnell 2007; Chudi 2010). These issues can increase the risk of disease and other health risks (Cameron, Shah, and Olivia 2013; Engel and Susilo 2014; Mulumba et al. 2014; Glavey et al. 2015; Chong et al. 2016; Hutton and Chase 2016). In the same vein, widespread deforestation and other major environmental degradation have become a significant concern in developing countries (DeFries et al. 2007; de Lima and Buszynski 2011; Lubowski et al. 2014; Busch et al. 2015; Leblois, Damette, and Wolfersberger 2017; Arshad et al. 2020). Global predictions identify that by 2050, 30 to 60 million ha of tropical rainforest in developing countries will be lost, on top of what has already been destroyed; with associated massive climate and biodiversity impacts (Leblois, Damette, and Wolfersberger 2017). As well, water management in developing countries is becoming increasingly problematic (Playán, Sagardoy, and Castillo 2018), as challenges from harmful environmental changes and increasing demand from human populations add to the challenges maintaining water quality and availability associated with poor irrigation infrastructure (Tibor et al. 2000). Given these challenges for developing countries, this present study will investigate service and environmental performance at the local government level in Indonesia; a developing country in which local governments would benefit greatly from information about factors affecting service and environmental performance. This information is essential in order to evaluate the local governments' capabilities to fulfil their service and environmental

responsibilities. To date, local authorities in Indonesia have had very limited information about their service and environmental performance.

A second gap in the literature is that there are relatively limited studies that address the influences of public sector governance, such as transparency, accountability, and internal control on public sector service and environmental performance in developing countries (Belal 2015; OECD 2017; Usang and Salim 2018; Sofyani, Riyadh, and Fahlevi 2020; de Silva, Yapa, and Vesty 2020). Research on public sectors in developing countries has typically focused on investigating the effects of socio-cultural, demographic, economic, and political factors on local government's performance (Da Cruz and Margues 2014; Tavares 2017). Therefore, research into the influences of public sector governance represented by financial transparency, accountability, and internal control on local governments' service and environmental performance is necessary to help with identifying and understanding the determinants of service and environmental performance. Financial transparency, accountability, and internal control have been suggested in previous studies as important factors for addressing local government inefficiencies that remain problematic in developing countries (Huka, Mchopa, and Kimambo 2014; Ningsih, Kusumawati, and Nohong 2019; Adams et al. 2019). Developing and maintaining strong financial transparency and accountability have direct positive effects on public organisational performance (Kis-Katos and Sjahrir 2017; Rana et al. 2019; Han and Hong 2019; Sofyani, Riyadh, and Fahlevi 2020), with public service delivery failures and environmental degradation linked to weak financial transparency and accountability (Damania 2002; Gaventa and McGee 2013). Likewise, when internal control is adequate in the public sector, this minimises the risk of fraud and other corruption and increases overall compliance and effective organisational

performance (Chenhall 2003; Muraleetharan 2011; Vijayakumar and Nagaraja 2012; Badara and Saidin 2013; Al-Thuneibat, Al-Rehaily, and Basodan 2015).

A third research gap is that although financial management factors, such as financial flexibility, financial independence, and financial resource management are considered very important for local governments, research that examines in detail the influences of these factors on local governments' service and environmental performance remains scant (Scutariu and Scutariu 2015; Sarr 2015; Kis-Katos and Sjahrir 2017). Currently, many local governments in developing countries suffer from financial dependency, relying on fiscal transfer or grants (Psycharis, Zoi, and Iliopoulou 2016). This issue may reduce their capacity to meet the basic needs of people in their jurisdiction (UN-HABITAT 2015; Marlissa and Urasti Blesia 2018).

Another concern regarding the organisational performance of local governments in developing countries is they often face challenges from public service expenditure reductions (Thompson 2017) and lack of financial capacity (Shah 2006; Smoke and Loffler 2013; UN-HABITAT 2015; Dick-Sagoe 2020). This will have implications for the quality of organisational performance. For the reasons stated above, this study needs to investigate the influence of financial management factors, represented by financial independence, financial flexibility, and resources management, on local government's service and environmental performance in a developing country.

A final research gap is that prior studies have not investigated thoroughly the influences of financial performance on local government's service and environmental performance, although financial performance is an essential factor for the functioning of local governments (Singer 2016; Khairudin et al. 2018; Mutiha 2018). Previous studies have not included financial performance as an integral part of environmental

performance analysis, since they have assumed that financial and environmental performance would be in conflict with each other (Aras and Crowther 2008); which is not necessarily the case. This research gap has led to deficiencies in current knowledge of the determinants of local government's performance; particularly as most previous studies have used financial performance as a dependent variable (Handa 2018; Masanja 2018; Awwaliyah, Agriyanto, and Farida 2019; Yasin 2020). As a result, local authorities have a limited understanding of whether their financial performance is capable of meeting their service and environmental obligations.

This study will investigate the mediation effect of financial performance on the relationship between public sector governance, financial management factors, and service, and environmental performance to help fill the aforementioned research gaps. Implementing financial performance as a mediator in this study clarifies whether public sector governance and financial management factors can explain local governments' service and environmental performance of this study contributes towards a comprehensive understanding of these issues, to enable policymakers to make effective decisions regarding the effective delivery of public services and environmental responsibilities.

3. Research Problems

Community service and environmental performance of local governments have gained attention in the literature (Greiling, Traxler, and Stötzer 2015; Alpenberg, Wnuk-Pel, and Henebäck 2018; Masiya, Davids, and Mangai 2019); with increasing recognition that performance in these two areas has a major influence on the lives, prosperity, health and wellbeing of local people since those performances have a critical effect on the daily lives and prosperity of the people (Štreimikienė 2015). However, in developing countries the quality of local governments' service provisions and provision of environmentally sustainable programs remains questionable (Morudu 2017; Sururi 2020; Dick-Sagoe 2020). Local governments need good public sector governance factors, such as financial transparency, accountability, and effective internal control to minimise risks and inefficiencies from poor financial management practices (Baltaci and Yilmaz 2006; Ott, Bronić, and Petrušić 2014; Zagorchev and Gao 2015; Donelson, Ege, and McInnis 2017; Nuswantara and Maulidi 2017; Han and Hong 2019). Financial transparency and accountability reduce the asymmetry of information, which in turn shifts the orientation of the provision of public services from political interests to a public or customer-driven orientation (Lanin and Hermanto 2019).

This present study follows four steps to examine the effects of public sector governance and financial management factors on service and environmental performance in Indonesia. These four steps are described below in turn. The first step involves investigating the influences of public sector governance represented by financial transparency, accountability, and internal control on service performance, as mediated by financial performance. The second step investigates the impact of financial management factors (financial flexibility, financial independence, and resources management) on service performance, mediated by financial performance. With respect to the focal variables in these first two steps, previous research has reported that both transparency and accountability are significant factors influencing the effectiveness of public management (Van der Waldt 2004; Hope 2005; Grimmelikhuijsen 2012; Bauhr and Carlitz 2020), and that financial transparency and accountability enhance public authorities' commitment to meeting

the needs of local communities and helps to maintain services standards and increase community involvement in public service processes (Chigwende, Mukuze, and Musasa 2013; Maksimovska and Stojkov 2016; Kis-Katos and Sjahrir 2017). Likewise, with respect to the variable 'internal control', previous research indicates that local government internal control mechanisms play an essential role in the quality of public service performance, including by helping to maintain public service standards, prevent abuse of power, and generate efficiency of services (Han and Hong 2019; Hutama and Yudianto 2019). These reported relationships between public sector governance and service performance support the inclusion of these variables in this study's first conceptual model.

With regard to the second step in the research – investigating the impact of financial management factors including financial flexibility, financial independence, and resources management on local government service – previous research has reported that for local governments, financial flexibility and financial independence can increase the capability of local governments to fulfil public needs, without undue constraints from external parties (such as higher-level governments) that may otherwise reduce their capacity to shape their budgets, raise local revenues, and spend their expenditures in the best interests of the local community (Almeida and Campello 2007; Sitek 2014; Kotarba and Kolomycew 2014). With sufficient financial flexibility and financial independence, local governments have more opportunities to allocate adequate budgets to important social sectors such as education and health programs to alleviate disparities in public services availability (Martinez-Vazquez, Lago-Peñas, and Sacchi 2017). These relationships between financial management factors and service performance are investigated in this study's second conceptual model.

The third step in this present study is to examine the effects of public sector governance (financial transparency, accountability, and internal control) on environmental performance, as mediated by financial performance; and the final step investigates the impact of the aforementioned financial management factors on environmental performance, again through the mediating role of financial performance. With regard to the third step's focus on financial transparency and accountability as factors in environmental performance, previous researchers (Bernauer and Koubi 2009; Sahide et al. 2016) have reported that strong financial transparency and accountability in local government leads to sound sustainable resources management, including environmentally sustainable policies and programs. Accountability helps governments to protect their communities from the risks and harms of environmental degradation (de Silva, Yapa, and Vesty 2020), with recent research highlighting the importance of developing countries managing their economic growth and associated infrastructure development carefully to minimise the risks to biodiversity values (Aye and Edoja 2017).

With regard to this study's focus on local government's internal control as a factor in the organisation's environmental performance, previous research has reported the need for local governments to pay careful attention to their internal control mechanisms, given the demonstrated link between internal control and an organisation's environmental credentials (Huang and Huang 2020). An effective internal control system can integrate and harmonise entities' motivation to pursue both financial benefits and environmental sustainability together (Liu 2018). To strengthen the environmental sustainability programs, the local government's financial capacity is also considered since there is a positive relationship between the availability of financial resources and the government's capabilities to invest in

environmental development (Esty and Porter 2005; Qi et al. 2014). Therefore, based on the above-mentioned arguments, the relationship between public sector governance, financial management factors, and environmental performance are included in the third and fourth models. The results of this current study will benefit the central and local authorities to perform comprehensive policies or decisions that lead to the improvement of local government's service and environmental performance.

4. Scope of the Study

This study investigates the influences of a set of factors determining the service and environmental performance of local governments. Specifically, this study examines the effects of public sector governance and financial management factors on the service and environmental performance of local government districts in Indonesia. This research also examines the indirect effects of public sector governance and financial management factors on service and environmental performance, through financial performance. For public sector governance, as noted above, this study concentrates on financial transparency, accountability, and internal control, while financial management factors are measured by financial flexibility, financial independence, and resource management.

In this study, financial performance, service and environmental performance are measured by applying a composite indices framework developed from the indicators selected, based on a local government context in a developing country. The composite indices of these variables are adopted based on principal component analysis, and this study follows the construction of composite indicators as posited by the Organisation for Economic Co-operation and Development (OECD) (2008)

and Mazziotta and Pareto (2013). The construction of composite indices is used to make reading and interpreting of local government performance than when many separate indicators are used (Booysen 2002; Farrugia 2007). Subsequent analyses are conducted on a cross-sectional basis, to investigate the empirical studies regarding how districts' service and environmental performance are affected by public sector governance and financial management factors. This study has chosen districts in Indonesia as the research sites from which to examine the empirical analyses of the theoretical relationship among the variables; as Indonesia is a useful case study in that it is representative of many other developing countries in terms of its administrative, political and cultural characteristics.

5. Research Objectives

To here, this study has two objectives. First, it aims to examine the influences of public sector governance and financial management factors on Indonesian districts' service performance, mediated by financial performance. In this study, public sector governance is represented by financial transparency, accountability, and internal control, while financial management factors are measured by financial flexibility, financial independence, and resource management. Second, this study aims to investigate the impacts of public sector governance and financial management factors on districts' environmental performance, through the mediating role of financial performance.

This study focuses on local government districts in Indonesia as an exemplar of developing countries. District authorities in Indonesia would benefit greatly from additional information about the factors that influence their service and environmental performance; and at present in Indonesia, local government's service

and environmental performance measurements are lacking. In the absence of sufficient information, local governments struggle to develop effective policies to improve their public services and ensure their development programs align with sustainable development principles. As stated in the country's Law 23/2014, in Indonesia the district-level authorities are responsible for delivering all public services (with the exception of foreign policies, national defence and security, the judicial system, fiscal and monetary policies, and religious affairs). Thus, research which provides local government with a greater understanding of the factors that influence their provision of public needs is particularly useful for Indonesia.

6. Research Questions

This study will achieve the research objectives discussed in the previous section by addressing four research questions:

- a. What are the influences of public sector governance and financial management factors on service performance?
- b. Are the relationships between public sector governance, financial management factors, and service performance mediated by financial performance?
- c. What are the influences of public sector governance and financial management factors on environmental performance?
- d. Are the relationships between public sector governance, financial management factors, and environmental performance mediated by financial performance?

7. Organisation of the Thesis

This thesis is organised into seven chapters. Chapter one presents the study's background and identifies research gaps and research problems, followed by the

study's scope, research objectives and research questions (Figure 1.1). Chapter two reviews the relevant literature and theories underlying service and environmental performance, and their linkages with public sector governance, financial management factors, and financial performance. Gaps in the literature are also identified. Chapter two continues with a discussion of institutional theory, to provide a theoretical understanding of the application of institutional isomorphism in public sector governance (financial transparency, accountability, internal control), and financial management practices of local governments. The final part of chapter two presents the study's conceptual framework and the specific hypotheses used to answer the research questions. The first and second theoretical models in this research are formulated based on the first research objective, while the third and fourth theoretical models are formulated based on the second research objective.



Figure 1.1: The Integrated Construct of Chapters

Chapter three presents a contextual background regarding the Indonesian local government. It includes an overview of public sector reforms at the district level in Indonesia, and discusses the decentralisation of political, administrative, and financial functions of local government. Chapter four describes the research methodology used in this study. This chapter is divided into several sections, as follows. It begins by explaining the data collection methods, data sources, and research samples. The next section explains the statistical techniques used in the data analysis, and describes the procedures used to measure the validity and reliability of the instruments. The chapter then presents a brief overview of the definitions and measurements of the focal variables.

Chapters five and six are the results chapters. They present the findings from the data analyses, including the findings from the quantitative methods based on descriptive statistics, multiple regression analysis, and a bootstrapping method. Chapter five focuses on the findings that address the first research objective, and chapter six presents the findings related to the second research objective. The chapters present a discussion of the underlying assumptions of multiple regression analysis and how this study meets them. The chapters then explain the results of the hypotheses testing. The chapters also present the results of the bootstrapping method generated by the Process macro for SPSS 24 programs. The bootstrapping method is applied to analyse the role of the mediation variable used in this study. The robustness tests are also discussed in the chapters. The findings of the data analyses aim to answer the study's research questions.

Chapter seven, the final chapter, discusses the findings of the study and discusses the study's contribution to the literature and existing knowledge; including both

theoretical and practical contributions. The chapter then identifies potential limitations of the study, and recommendations for future research.

Chapter 2 Literature Review, Theoretical Framework, and Hypotheses Development

1. Introduction

This study aims to investigate the determinants of public service and environmental performance in local governments in Indonesia. This study will also investigate how key determinants of local government performance, specifically public sector governance (financial transparency, accountability, internal control) and financial management factors (financial flexibility, financial independence, and resource management) are mediated by financial performance. To this end, this chapter reviews the literature relevant to the main purposes of the study. The linkage of Chapter 2 with the research objectives of this study is shown in Figure 2.1.





2. Service Provision and Protection of Environment

Local governments have responsibilities to provide public service and protect environmental sustainability (Štreimikienė 2015; Wiharani and Holzhacker 2016; Panya et al. 2018). The concept of public service is derived from the private sector, and defined as "deeds, processes, and performances" (Zeithaml, Bitner, and Gremler 2010, p.1). Service is also conceptualised as the interactions between customers and service providers, or between customers and products/services provided to overcome the customers' problems (Edvardsson, Gustafsson, and Roos 2005).

As well, scholars have used different approaches to define and measure service performance in public sectors (Hood 2007; Coste and Tudor 2013). Different constituencies associated with public services, such as consumers, taxpayers, government institutions, and politicians, may also use different criteria to judge the standard of public services (Boyne and Gould-Williams 2003). Whatever the exact criteria used to measure performance, it remains true that local government's service performance will have a significant impact on citizens' daily lives (Industry Commission 1997). Service performance is a complex and multidimensional concept, and 'what is effective service' will vary depending on a wide range of different environments, issues, and contexts (Williams 2004; Dubnick 2005; Coste and Tudor 2013). In general, good service performance is indicated by the provision of sufficient quantity and quality of public services to meet the community's needs and satisfy users (Maurel et al. 2014; Lanin and Hermanto 2019). These definitions have typically separated local government's public service from its environmental

performance, although arguably, service should always include the sound environmental performance, given its importance as described below.

As with the debate about the concept of public service, scholars have not reached a consensus regarding how to conceptualise governments' environmental performance (Aras and Crowther 2008). Several researchers have conceptualised environmental performance of governments by considering two aspects: the expectation of stakeholders; and compliance with environmental regulations to which the organisation is beholden. For example, Sarkis (1998) defined governments' environmental performance as "a commitment to compliance, prevention of pollution, continual improvement, that is to be documented and communicated to stakeholders" (p. 163). With respect to regulatory compliance, environmental performance can be defined as "the compliance of one or more environmental aspects with the regulations, and several others" (Ramos et al. 2009, p. 36). However, more recently Gallego-Alvarez et al. (2014) have argued that a government's environmental performance is not related only to community expectations or compliance with regulations; instead, they posit that it is defined as a government's operational activities to produce environmental public goods and protect biodiversity with its various environmental activities. This is a broader definition which includes the necessity for governments to protect the environment and biodiversity irrespective of stakeholder expectations and regulations; although the definition still contains ambiguity about whether such protection includes protection for the medium and long term, including for future generations.

Public sector organisations currently have more responsibilities to improve ecological sustainability than private sector entities, and arguably more opportunities (Adams,

Muir, and Hoque 2014). Local governments have a pivotal role in protecting the biosphere, as they are the closest public entities and the frontline of sustainable development processes (Thomas 2010; Williams, Wilmshurst, and Clift 2011; Wiharani and Holzhacker 2016). Through their local community reach, local governments can together empower whole societies to adopt sustainable use and protection of ecosystems (Potts 2004; Thomas 2010; Ball and Grubnic 2011).

Effective performance of local governments implies that their provision of public goods and services (including environmental sustainability) occurs efficiently to meet the public's needs (Mihaiu, Opreana, and Cristescu 2010; Da Cruz and Marques 2014). To maintain effective performance, local governments require both sound public sector governance and financial management (Smedby and Quitzau 2016; Salman, Mostert, and Mugwisi 2018; Nguyen et al. 2019). In light of the aforementioned arguments, this study has developed a model to illustrate the factors affecting the performance of local government, as will be discussed in the following subsections.

3. Determinants of Public Service and Environmental Performance of Local Governments

This study applies three determinants of public service and environmental performance of local governments: (1) public sector governance; (2) financial management factors; and (3) the mediating role of financial performance, as shown in Table 2.1. Public sector governance is represented by financial transparency, accountability, and internal control, while financial management factors are measured by financial flexibility, financial independence, and resource management.

No.	Determinants	Measurements
1.	Public Sector	- Financial Transparency
	Governance	- Accountability
		- Internal Control
2.	Financial Management	- Financial Flexibility
		- Financial Independence
		- Resource Management
3.	Financial Performance	

Table 2.1: Determinants of Public Service and Environmental Performance

Public sector governance has become an essential topic of discussion and concern among scholars and international development organisations (Kulshreshtha 2008; Poniatowicz, Dziemianowicz, and Kargol-Wasiluk 2020). It has various lists of principles (Van Doeveren 2011; Eysink and Paape 2016) and scholars have no consensus about which are the principles of good governance in each case (Cerrilloi-Martínez 2017; Alqooti 2020). This study focuses on three public sector governance measurements: financial transparency, accountability, and internal control.

3.1 Public Sector Governance

Figure 2.2 (a) and (b) represents the theoretical framework of this study that outline the relationships between public sector governance factors and service and environmental performance through the mediation of financial performance. Following the arguments of the previous research, sound public sector governance is important in local government performance because it reduces misuse of public resources, which in turn improves the organisation's performance (Adiputra, Utama, and Rossieta 2018). Public sector governance is defined slightly differently by different scholars, organisations and countries (Smedby and Quitzau 2016; Lohrey et al. 2019; Tabassum and Singh 2020); with one recent narrow definition from Adrian Cadbury's report 'The Financial Aspects of Corporate Governance' defining governance as the system that controls and directs organisations (Tabassum and Singh 2020). By this definition, public sector governance provides a series of control mechanisms that aim to ensure resource allocation efficiency and the achievement of specified objectives (Wiharani and Holzhacker 2016). Broader definitions consider that public sector governance is a set of checks and balances that aim to ensure that organisations are accountable to their stakeholders and societies (Solomon 2013). A similar broad definition refers to public sector governance as "the arrangements and practices which enable a public sector entity to set its direction and manage its operations to achieve expected outcomes and discharge its accountability obligations" (ANAO 2014, p.7). In Indonesia's Law 1/2004, public sector governance is indicated by the implementation of financial transparency, accountability, and internal control that aim to enhance the government performance.



Figure 2.2: Theoretical Framework of Public Sector Governance Factors Affecting Local Government Performance

In the case of developing countries, sound public sector governance is often lacking, with inefficient financial management in particular being identified as a major obstacle to the fulfilment of public needs (Abdullahi and Chikaji 2017; Wardhani, Rossieta, and Martani 2017; Sekiguchi 2019; Pumkaew 2019), leading to a lack of provision of basic amenities including health centres, schools, and good roads (Lewis 2017b). Poor public sector governance by local governments likewise has major repercussions for developing countries, particularly with respect to environmental performance. The uncontrolled issuing of licences by local governments for destructive forestry and mining practices (Siombo 2014), alongside the overall lack of effective monitoring and prevention of deforestation and pollution, is leading to major environmental challenges, including unsustainably high deforestation rates in Indonesia and across the developing world (Ahmed et al. 2015).

Local governments need strong public sector governance to manage the environment sustainably in the face of population growth and pressure from powerful interests; in order to ensure clean air, water, biodiversity and all ecosystem services are maintained for current and future generations (Cramer and Cheney 2000; York, Rosa, and Dietz 2003; Cole and Neumayer 2004; Birčiaková, Stávková, and Antošová 2015). Together, these points highlight the urgency of the need for local governments in developing countries to strengthen their public sector governance, to improve their sustainable environmental practice and overall performance (Tacconi 2007; Goryakin et al. 2017; Priono et al. 2019). The public sector governance attributes are reviewed below to investigate the influences of public sector governments, mediated by local government's financial performance. The substantial rationale of

the theoretical framework of public sector governance factors affecting local government performance will be discussed in more detail in the following subsections.

3.1.1 Financial Transparency

Local governments have a critical role in delivering services to meet the needs of communities. However, in developing countries in particular, the public is often either deprived of public services of adequate quality, or experiences challenges related to equal access (Sun, Fu, and Zheng 2017; Beschel et al. 2018; Pradeep and Dickwella 2012). People are required to navigate complicated bureaucratic systems to receive public services (Beschel et al. 2018), and poor service performance is common, with many associated risks. For example, poor health care facilities in developing countries contribute to high levels of maternal mortality and poor child health outcomes (Atuoye et al. 2015); while poor provision of education leads to low rates of primary and secondary school completion (Kuepie, Shapiro, and Tenikue 2015)

Therefore, as part of Indonesia's public sector reform agenda, Law 1/2004 has mandated that all government entities, including local governments, must demonstrate financial transparency. Transparency refers to the government's decision-making processes and information is readily available to and easily accessed by the public (Wijayati, Hermes, and Holzhacker 2016). Transparency includes a demonstrated commitment to openness by an organisation (Dubbink, Graafland, and Van Liedekerke 2008); and has been identified as one of the key elements of governance that can both alleviate misuse of public resources (Fölscher,
Krafchik, and Shapiro 2000; Kolstad and Wiig 2009; Adiputra, Utama, and Rossieta 2018), and reduce information asymmetry between policy-makers and the public (Benito and Bastida 2009; Tejedo-Romero and de Araujo 2018). In Indonesia, the public's right to access public information is stated in Law 14/2008 and Regulation of the Government 61/2010.

Similarly, the literature on government environmental performance recognises that financial transparency supports environmental sustainability (Halkos and Tzeremes 2014). Damania (2002) posited that environmental degradation in developing countries was influenced by the low level of transparency of public institutions. To address environmental degradation therefore, the government should focus on improving democratic public policymaking; indeed, a greater degree of democracy has been reported to be correlated positively with better air quality (Bernauer and Koubi 2009)

The more complete the financial information that is made available by the government to the public, the more transparent the government is; and this has positive impacts on government performance (Bauhr and Carlitz 2020). Public service delivery becomes more effective, efficient and aligned with community' preferences when local governments promote community participation (Devas and Grant 2003; Cheema 2007), empower local voices, respond positively to community needs (Gaventa and McGee 2013) and the environment (Sahide et al. 2016).

3.1.2 Accountability

To improve the performance of public institutions, another public sector governance factor as mandated by Indonesia's Law 1/2004 is accountability. Accountability of

local governments is recognised as a critical factor that influences both service performance (Kloot 1999; Kis-Katos and Sjahrir 2017; Han and Hong 2019) and environmental performance (Ribot, Agrawal, and Larson 2006; Galera et al. 2014; Williams 2015). Service delivery failures are influenced by accountability failures (The World Bank 2003; Gaventa and McGee 2013). A lack of accountability of public service providers to their public leads directly to poor delivery of public services (The World Bank 2003). Similarly, there is also a growing consensus that government authorities must be accountable to the public to address climate change, and that the public deserves protection from climate-related harm and risks (Mees and Driessen 2019). Sustainable development and effective environmental management require high levels of government accountability (Gberevbie et al. 2017).

Therefore, to meet the diverse range of social needs in communities and fulfil their responsibilities in the protection of the environment, local government needs to be responsive and accountable (Maksimovska and Stojkov 2016). Accountability includes facilitating the public to monitor the provision of public services; which means providing them with adequate access to public information (Pérez, Hernández, and Bolívar 2005; Popescu Ljungholm 2015). Accountability, as measured by staffing, performance evaluation, and compensation, has been shown to have a significant influence on governments' organisational performance (Han and Hong 2016). Ensuring that governments have strong accountability is, therefore a prerequisite for effective public service delivery (Kis-Katos and Sjahrir 2017).

The accountability of governments can be assessed in different ways, depending on one's focus. One way of assessing a government's accountability is to consider "for what purposes the [government] services are intended" (Kearns 1994, p. 186). To

aid with this assessment, Flack (2007) identifies four purposes of accountability: accountability for accounting information; accountability for outcomes; administrative accountability, and accountability for a mission. Another way in which accountability can be assessed is to consider "to whom are the services delivered" (Kearns 1994, p. 186). In local government, there are two kinds of "to whom accountability": vertical accountability and horizontal accountability (Signé and Korha 2016; Khotami 2017). Vertical accountability relates to local government fulfilling its obligations to provide accountability reports to the central government. Horizontal accountability is the commitment of local governments to make their accountability reports accessible to the public. Given that accountability is one of the determinants of government performance mandated by Indonesia's Law 1/2004, the government has enacted Presidential Regulation 29/2014 to regulate and improve the accountability of public institutions.

3.1.3 Internal Control

The literature identifies that local government performance can be improved by developing and maintaining effective internal control (Simangunsong 2014). Public sector entities that have well-structured internal control are more likely to improve their performance and meet their stakeholders' expectations (Berrone, Surroca, and Tribó 2007; Ong'unya and Abbey 2019). Given these reports in the literature, this study expects that the variable of 'internal control' will be one of the determinants of the local government's service performance. Many previous studies have demonstrated that effective internal control helps to deter fraud and errors, increases earnings quality, and prevents loss of revenue (Schneider and Church 2008; Brown, Pott, and Wömpener 2008; Prawitt, Smith, and Wood 2009; Jokipii 2010;

Vijayakumar and Nagaraja 2012; Adelin and Fauzihardani 2013; Al-Thuneibat, Al-Rehaily, and Basodan 2015).

Previous studies have found that where internal control is taken adopted by organisations as a proactive preventive approach, this contributes significantly to organisational success (Benedek, Szenténé, and Beres 2014; Pujiono, Sukarno, and Puspitasari 2016), and effective environmental management (Kareem, AlHassani, and Ali 2019; Huang and Huang 2020). Effective internal control has been demonstrated to mitigate fraud and corruption including mismanagement of funds; promote compliance with laws and regulations; and overall, help to maintain sound financial management (Coram, Ferguson, and Moroney 2006). In contrast, inadequate internal control has led to many financial reporting and accounting scandals and financial mismanagement (Donelson, Ege, and McInnis 2017). Ineffective internal control presents risks not only for organisations directly, but also for environmental sustainability, with numerous examples of weak and ineffective local governments monitoring and compliance mechanisms leading to substantial environmental damage (Resosudarmo 2004). Alongside losses to biodiversity, such environmental damage may have major economic and public health impacts, for example, from water, air and soil pollution, and from the disease risks (and economic and social risks) associated with climate change and deforestation (higher exposure to coronaviruses due to deforestation being a case in point).

3.1.4 Other Public Sector Governance Attributes

This section will discuss four other attributes of public sector governance (Armstrong, Jia, and Totikidis 2005; Mbecke 2014). The first attribute is leadership. It refers to the

capability of the chairperson to set the strategic vision and determine the direction of the organisation (Armstrong, Jia, and Totikidis 2005; APEC 2011). Leadership in local government is a burgeoning issue because the leaders face situations that continuously change and create new challenges (Svara 2008). The local government chairperson who can diagnose problems, manage the conflicting voice, ensure the direction of organisation, and conduct practical solutions has a significant impact on public service performance (Pickering et al. 2014; Uche 2014; Rahman et al. 2018). Moreover, the effective leadership enhances citizen engagement and financial resources allocated for environmental sustainability programs (Wang, Van Wart, and Lebredo 2014; Hofstad and Vedeld 2021).

The second attribute is direction. It refers to a strategic vision that shows a broad and long-term perspective and indicates historical, cultural, and social complexities (Graham, Plumptre, and Amos 2003). Direction plays a steering role that determines what tasks to do, who should carry out the authorities, and where someone has to work (Graham, Plumptre, and Amos 2003; Lockwood 2010). Aligning the direction and strategic vision of the organisation with the wildlife conservation can enhance the local government's environmental performance (Du Plessis 2015). The third attribute is authority. It refers to three components: (1) the legitimacy of governing actors to make decisions mandated by laws or stakeholders, (2) the delegation of power to the lower level of authority, and (3) the integrity of decision making (Lockwood et al. 2010). The local governments can fail to achieve good performance because they have lost legitimacy or trust from their stakeholders (Lawton, McKevitt, and Millar 2000; Van de Walle and Bouckaert 2003). Citizen involvement or participation in decision-making is one of the main factors that can maintain the

Lawton, McKevitt, and Millar 2010; Gaspar and Mkasiwa 2015).

The fourth attribute is stewardship. It relates to the awareness of public officials to act as good stewards motivated by the best interest of the communities rather than triggered by their own interests (APEC 2011). Three mechanisms create stewardship behavior: normative, instrumental, and affective mechanism (Viswanathan et al. 2020). Normative mechanism focuses on the fact that the leaders act as stewards because of the duty or norm, while instrumental mechanism influences the leaders to recognize that their roles as stewards can bring long-term benefits to the people. Affective mechanism affects the leaders' character to become stewards because it can build their high sense of empathy towards the people and societies. With attention to these stewardship mechanisms, the public officials can manage their authorities and public resources to improve financial performance, public service delivery quality (ANAO 2014), and environmental sustainability (Viswanathan et al. 2020).

Understanding the explanations in the abovementioned paragraphs, the four public sector governance attributes (leadership, direction, authority, and stewardship) are important as part of public sector governance and there are many studies done. However, this study does not include these four sector governance attributes (leadership, direction, authority, and stewardship) in the analysis. This study concentrates on three public sector governance measurements: (1) financial transparency, (2) accountability, and (3) internal control due to several reasons. The first reason is that transparency, accountability, and internal control are the main variables of good governance for public sector that have been identified by the

previous studies (IFAC 2001; Kulshreshtha 2008; Bhuiyan and Amagoh 2011; ANAO 2014). These three principles have positive impact on sustainable development and public good and service delivery (Kulshreshtha 2008; Bhuiyan and Amagoh 2011; Jha, Zhuang, and Quising 2013; Sahide et al. 2016; Poniatowicz, Dziemianowicz, and Kargol-Wasiluk 2020).

The second reason is that transparency, accountability, and internal control are the main variables that represent public sector governance in Indonesian case (Kristiansen et al. 2009; Guntur 2017). The implementation of public sector governance in Indonesia is mainly based on the implementation of transparency, accountability, and internal control as mandated by Indonesia's Law 28/1999, Law 14/2008, Law 1/2004, Government Regulation 60/2008, and Presidential Regulation 29/2014.

3.2 Financial Management Factors

This study extends the theoretical framework that investigates the influences of financial management factors on service performance mediated by local government's financial performance as further analysis, as shown in Figure 2.3 (a) below. Previous studies have found that local governments' service performance is influenced by a range of public financial management factors (Scutariu and Scutariu 2015; Kis-Katos and Sjahrir 2017; Goryakin et al. 2017). Local governments with good financial management have more opportunities to improve the quality of public service provision (Wardhani, Rossieta, and Martani 2017; Alonso and Andrews 2019). In Indonesia's Law 23/2014, to improve local government's financial management, typical reforms include increasing local government's financial flexibility, financial independence, and financial resource management. Without

financial flexibility and financial independence, local government officials tend to focus on fulfilling the central government's interests rather than those of their citizenry (Martinez-Vazquez and Timofeev 2010).



Figure 2.3: Theoretical Framework of Financial Management Factors Affecting Local Government Performance

This study will also develop another conceptual model to extend the present study's analyses. Therefore, this study examines the impacts of financial management factors on environmental performance by local government mediated by financial performance, as shown in Figure 2.3 (b). Previous research found that financial management factors have influences on the protection of environmental sustainability (Qi et al. 2014; Alcaraz-Quiles, Navarro-Galera, and Ortiz-Rodríguez 2014; Ong'unya and Abbey 2019). The role of financial management in improving environmental sustainability is self-evident (Al Breiki and Nobanee 2019). Governments should adopt fiscal decentralisation to support sustainable development, including both financial flexibility and financial independence of local government (Zhou, Zhou, and Yu 2020; Chen and Liu 2020). Local governments that are financially independent and flexible will have more opportunities to raise

revenues; which in turn may support sustainable development programs (Esty and Porter 2005). The higher level of profitability of the entities, the more opportunity to invest in environmental sustainability (Stanwick and Stanwick 1998; Qi et al. 2014).

This study adopts three financial management factors: (1) financial flexibility; (2) financial independence; and (3) financial resource management, as promulgated by Indonesia's Law 17/2003, Law 33/2004, and Law 23/2014. Through these laws, the Indonesian local governments have financial autonomy to raise their own revenue and manage their financial resources in ways that meet local community needs and protection of the environment. The substantial rationale of theoretical framework of financial management factors affecting local government performance are elaborated below.

3.2.1 Financial Flexibility

Financial flexibility refers to having a sufficient cash balance to finance expenditures (Soenen 2003). Where local governments have sufficient financial flexibility and financial independence, this will increase their financial capacity to meet citizens' needs (Kis-Katos and Sjahrir 2017). In addition, it is expected that local governments with adequate financial capacity will pay more attention to environmental sustainability (García-Sánchez, Frías-Aceituno, and Rodríguez-Domínguez 2013; Alcaraz-Quiles, Navarro-Galera, and Ortiz-Rodríguez 2014) and public service delivery improvement (Kotarba and Kolomycew 2014). Research has found that organisations which are financially flexible tend to have good performance, especially during financial crisis (Campello, Graham, and Harvey 2010; Bancel and Mittoo 2011a; Arslan-Ayaydin, Florackis, and Ozkan 2014; Al-Slehat 2019), as they

have the capacity to avoid financial distress in the face of negative shocks, and increase the entity's opportunities to benefit from short-term investments (Gamba and Triantis 2008).

The availability of spare or 'slack' resources empowers local governments to invest these resources to meet social needs, including to enhance the quality of the environment – so-called 'slack resources theory' (Cheng and Kesner 1997; Waddock and Graves 1997; Chiu and Liaw 2009; Qi et al. 2014). According to this theory, higher profitability will allow local governments and other organisations more opportunities to invest in their social responsibility (Stanwick and Stanwick 1998; Qi et al. 2014). Another theory which posits a significant positive relationship between having sufficient financial capacity and environmental performance is the Kuznets curve theory (Esty and Porter 2005), which explains that governments in municipalities with high-income rates tend to be more concerned with environmental sustainability, such as solar and renewable energy and sustainable technologies, than those in locations with lower income rates.

3.2.2 Financial Independence

Financial independence refers to a local government's autonomy to raise revenue and spend revenue without being influenced inappropriately by external parties (Petkovska 2011). The main aim of the financial independence of local government is to improve the efficiency and effectiveness of financial management, which in turn improves public service performance (Fafurida and Pratiwi 2017). The literature has also shown that financial independence enhances a local government's ability to overcome fiscal imbalances, since with greater independence they will have greater autonomy to raise revenue, which in turn leads to sound cash balances (Martinez-Vazquez and Timofeev 2010). Financial independence has positive effects on the public service performance of local governments because they have adequate access to a broad range of revenues, such as local taxes and retributions to finance a growing range of services to people (Productivity Commission 2008).

When local governments have the capacity to independently manage their financial resources, this also increases their opportunities to contribute further in environmental protection programs (Oulasvirta and Turała 2005) because recent suggest that subnational governments have taken initiatives trends in environmentally friendly development (Corfee-Morlot et al. 2009). For example, many local governments, especially within the Organisation for Economic Cooperation and Development (OECD), have been involved in the greenhouse gas reductions, "some beyond national commitments, or in the absence of national action" (Corfee-Morlot et al. 2009, p.29).

3.2.3 Resource Management

In this study, resource management refers to the local government's realisation of revenue and expenditure that has been planned by the local government. This realisation is measured by (a) the ratio of the actual aggregate expenditures to the approved aggregate budgeted expenditures, and (b) the ratio of the actual aggregate revenues to the approved aggregate budgeted budgeted revenues (Government Regulation 17/2017). The achievement of revenues and expenditure targets has demonstrated significant effects on financial performance (Awwaliyah, Agriyanto, and Farida 2019), and increases the effectiveness of public service provision (Andriati 2019; Lannai

and Amin 2020; Kotarba and Kolomycew 2014). Likewise, the ability of local governments to achieve financial resources in a timely manner increases their opportunities to implement their environmental sustainability programs effectively (Zhang, Zhang, and Steklova 2020).

However, when budget realisation by local government is too slow, this may lead to the downsides of cash sitting idle and poorer-quality public services (Erlina and Muda 2017; Andriati 2019). Idle cash may occur if local governments fail to allocate their financial resources appropriately (Modlin and Stewart 2012). The timely allocation of financial resources is essential for local governments to meet public needs (Satria, Juanda, and Sarma 2018; Tkachenko 2020) and for economic growth (Okoye et al. 2019). Other recent research has reported that local governments' ability to meet revenue targets increases their short-term solvency to pay their current liabilities (Abdullah and Muthia 2019). Therefore, in Indonesia, the central government periodically monitors and periodically evaluates the local government's realisation of revenue and expenditure to have effective financial resources management to avoid idle cash balances (Timoshenko 2018), as mandated by the Minister of Finance Regulation 214/2017.

3.3 Mediating Role of Financial Performance

Figure 2.2 and 2.3 show the mediating role of financial performance on the relationship between public sector governance factors, financial management, and service and environmental performance of local governments. Concerning the underlying basis for investigating the mediating role of financial performance, this study follows the arguments of Aras and Crowther (2008) and Tamawiwy, Sondakh,

and Warongan (2016). Aras and Crowther (2008) suggested that there is no dichotomy between financial performance and organisational performance and 'the two concepts conflate into one concern' (p.439). This means good financial performance leads to good environmental performance (Stanwick and Stanwick 1998; Esty and Porter 2005; García-Sánchez, Frías-Aceituno, and Rodríguez-Domínguez 2013; Qi et al. 2014) and vice versa (Aras and Crowther 2008). Similiarly, Tamawiwy, Sondakh, and Warongan (2016), Terrell-Smith 2018, Mbulawa (2019) argued that solvent financial performance enhances local government's opportunities to finance capital expenditures for public service provision.

In addition, following the arguments in previous research, it is recognised that public sector governance, such as financial transparency (Wanjau, Muturi, and Ngumi 2018), accountability (Darweesh 2015; Zagorchev and Gao 2015), and internal control (Yemer and Chekol 2017) are all associated with financial performance. High organisational transparency, accountability, and internal control lead to sound financial performance, since these factors help to minimise the risk of fraud, corruption and other abuse of power (Benito and Bastida 2009; Kim and Lee 2010; Han and Hong 2019), and enhances taxpayers' and investors' trust, which in turn increases the level of local revenues and government's financial resources (Guillamón, Bastida, and Benito 2011). Likewise, following the previous literature, it is expected that financial management factors represented by financial flexibility and financial independence will impact an organisation's performance (Butt, Hunjra, and Rehman 2010; Veeraraghavan 2018); with sound public financial management demonstrated to improve public service delivery (Welham, Krause, and Hedger 2013).

The literature demonstrates also that local governments with sound financial performance have more capacity to pay expenditures, provide community services, and improve economic growth (Batafor 2011; Dandara-Tabacaru and Danila 2017). The financial performance of local government, in turn, influences public welfare (Yasa and Utama 2015). Therefore, with local governments' service performance depending not only on public sector governance but also on financial performance, the financial performance of the local government is important to include in this study as a mediating variable. The theoretical relationships of this model will be further discussed in the section on hypotheses development, below.

The existing literature has developed a wide range of terms, dimensions, and indicators of financial performance by local government. The diverse dimensions and indicators of financial performance proposed by academia and practitioners do not lead to an agreement, which is the most appropriate ways to evaluate financial performance. Some scholars assume that certain theories are too simple (Howell and Stamm 1979; Adams 1981; Groves, Godsey, and Shulman 1981), while their opponents claim that others' schemes are too broad (Raman 1982; Brown 1993; Kloha, Weissert, and Kleine 2005b). For this study's purposes, a composite index developed from the indicators of short-term solvency, long-term solvency, budgetary solvency, and service-level solvency - is used to measure the financial performance of the local government, as will be discussed in more detail in Chapter 3.

3.4 Control Variables

This study applies three control variables that can potentially influence the performance of local governments. The first control variable is population size.

Population size has impacts on both demand and supply of public services and goods provided by a local government (Hyman 2014). On the one hand, the population size provides resources for a local government because it can increase the revenue base for a local government (Nollenberger, Groves, and Valente 2003). This condition has positive impacts on its financial performance. On the other hand, population size can increase local government's expenditures to fulfil the number of public services and goods required by the people (Wang, Dennis, and Tu 2007; Jones and Walker 2007; Zafra-Gómez, López-Hernández, and Hernández-Bastida 2009a). Likewise, population size might contribute to deforestation (De Sherbinin et al. 2007; Angus and Butler 2011) since the increased population size leads to the increasing needs on forest resources, such as timbers, fuels, and land (Gorenflo et al. 2011).

The second control variable used in this study is the poverty rate. Poverty is defined as a lack of economic resources that has negative effects on social life (Mood and Jonsson 2016). The poverty rate leads to negative impacts on the financial performance of a local government because the poor condition of communities makes it difficult for the local government to generate revenues for public service provision (Carmeli 2008). On the contrary, a local government with prosperous communities has more opportunities to receive more revenues from levies, taxes, and retributions and provide a better quality of public services (Nollenberger, Groves, and Valente 2003). Likewise, poverty is among the principal factors that are closely linked with the degradation of the natural resources in developing countries, such as steeply sloping and erosion-prone hillsides, because poor farmers raise more concerns about short-term needs for their survival rather than long-term sustainability (Masron and Subramaniam 2019). However, it is also acknowledged

that other poor farmers might have awareness on environmental sustainability (Asadi et al. 2008).

The third control variable applied in this research is gender diversity. Gender diversity becomes the burgeoning issue in public spheres indicated by the immense pressure to improve the gender balance between the number of female and male employees in the public sectors, especially in the high-level position (Jones 2009; Baker, Ali, and French 2019). The representation of females is essential because a sufficient number of females in strategic positions can improve the performance of an organisation (Dani, Picolo, and Klann 2019). The environmental conflicts and problems can be solved by the important role of women (Asteria, Brotosusilo, and Apriana 2018) since females are more interpersonally sensitive to engange in ecological issues (Lu and Herremans 2019). Moreover, females have a more participative and communal leadership style to make more dialogues with stakeholders (He and Jiang 2019).

4. Theoretical Underpinning

4.1 The Link Between Institutional Theory and the Attributes of the Study

When considering the use of institutional theory in the conceptual framework of this study, it is relevant to note that in management accounting research, the application of institutional theory has become prevalent (Deegan 2009; Kilfoyle and Richardson 2011; Modell 2012; Hoque and Chia 2012). This study have adopted institutional theory because this study is related to public sector issues where institutions play a big role in designing and framing public sector governance, financial management factors, and local government's performance (Harun and Kamase 2012; Manafe and Akbar 2014; Adiputra, Utama, and Rossieta 2018). Institutional theory applies more

in public sectors because these organizations follow and under guidance of overall institutional framework (Parker and Gould 1999; Torres, Pina, and Yetano 2011). There are many published papers that used institutional theory to explain public sector governance, financial performance, and local government's service performance (Ahyaruddin and Akbar 2018; Hudaya et al. 2015). For these reasons, we have followed the same tradition.

The institutional theory argues that the legitimacy of organisations is achieved by fulfilling the external expectations, performing management control systems (Cavalluzzo and Ittner 2004), and complying with rules and regulations (Akbar and Wijaya 2013). The regulative view of institutions is the one that is most commonly used to explain public sector governance and financial management reform (Reginato, Fadda, and Pavan 2010; Islamiyah, Sukoharsono, and Rosidi 2017; Ahyaruddin and Akbar 2018).

In the context of this study, public sector reform has changed the process of public accountability, transparency, management control, and financial management of local governments. The institutional theory is an effective solution to understand the changes and the adaptation process of local governments influenced by the surrounding environment, such as rules and social expectations (Reginato, Fadda, and Pavan 2010; Islamiyah, Sukoharsono, and Rosidi 2017). The implementation of public sector financial reforms and management accounting was mandated by regulations from the central government (Adiputra, Utama, and Rossieta 2018; Ahyaruddin and Akbar 2018). Local governments are responsible for submitting their performance reporting periodically to the central government, to maintain their legitimacy (Putri and Aswar 2020). Moreover, the central government has the right to

monitor and impose sanctions on local governments if they fail to follow the regulations as mandated by the central government (Alexander, Jackson, and Paterson 2014). The local governments will make adjustments until they reach the organisational homogenisation and achieve the institutionalisation process which in turn influence their performance (Islamiyah, Sukoharsono, and Rosidi 2017).

There are more coercive pressures from central government on the local governments for improving their performance in a decentralised government such as Indonesia (Brignall and Modell 2000). Central government tends to have greater coercive power over local governments than other levels of government (Akbar, Pilcher, and Perrin 2012). Central governments enact laws and regulations that affect local governments; in Indonesia, these regulations aim to improve local governments' performance and support them to achieve reform agendas. For example, the central government has introduced Government Regulation 56/2005 on Regional Financial Information Systems, Law 14/2008, Government Regulation 61/2010 about Public Information on websites or other electronic media. These regulations aim to improve local governments' transparency and encourage the involvement of the citizenry to monitor the process of development programs.

In Indonesia, in addition to the aforementioned regulations, the central government controls local governments' performance through a number of other laws and regulations as well, which is relevant to identify for this research. Two regulations related to accountability are Government Regulation 8/2006 and Presidential Regulation 29/2014 on the Government Accountability System. These two regulations enforce all government entities, including local governments, to build

their accountability profile and that of the other public institutions for which they are responsible, with the aim of improving the performance of public services. The central government is the entity that determines the implementation of the accountability system at the local government level, in terms of the procedures, measurements, and accountability reporting. Another central government regulation, Government Regulation 60/2008, has been enacted to require local governments to implement internal control. According to this regulation, the local government's internal control should include the five interrelated components that have been identified as important in internal control (COSO 2013), namely: control environment; risk assessment; control activities; information and communication; and monitoring activities. This internal control tool aims to provide an adequate measurement that local governments' activities are being conducted effectively and efficiently. For similar reasons, the central government has enacted Indonesia's Law 17/2003, Law 33/2004, and Law 23/2014, which regulate the implementation of the financial management of local governments. Through these laws, the central government regulates the procedures for the use of local government budgets, and determines the revenue resources that can be received by local governments; including their own local revenues, fiscal transfers, grants and loans.

4.2 The Conceptualisation of Institutions

For the present research, it is useful to consider how institutions have been conceptualised in the literature, given that governmental institutions are a focus of this study. Scott (1987) identified that in organisational theory, a wide range of concepts and definitions of institutions and institutionalisation are used by different theorists; he referred to "the many faces of institutional theory". While not claiming to

be exhaustive, Scott (1987) reviewed four sociological formulations of institutionalisation. The first is that institutionalisation is "a process of instilling value" (Scott 1987). Before instilling value and supplying intrinsic worth to an organisation, value has only instrumental utility as a mechanical and disposable tool (Wahid and Sein 2013). However, by inducing value, institutionalisation leads to the stability of the organisation's structure (Scott 1987). The second conceptualisation of institutionalisation reviewed by Scott (1987) is that institutionalisation constitutes a process of creating reality. This means that an organisation is based on a shared social reality, which in turn creates a human construction that "occurs whenever there is a reciprocal typification of habitual actions by types of actors" (Berger and Luckmann 1979, p.54).

The third sociological conceptualisation of institutionalisation reviewed by Scott (1987) is that institutional systems constitute a class of elements. This stream of institutional theory embraces the idea that a distinctive class of elements represents the existence of organisational structure (Scott 1987). A system of shared beliefs creates the formal structures of institutionalisation in organisations (Russo et al. 2012). The fourth and final conceptualisation of institutionalisation reviewed by Scott (1987) is that institutions act as distinct societal spheres. This conception of institution describes the institutionalisation process within various beliefs and human activities that tend to persist, such as religion, works, families, and politics (Scott 1987).

The institutional theory emphasises conformity to the existing values or standards that aim to maintain institutional legitimacy (Meyer and Rowan 1977; DiMaggio and Powell 1983; Eisenhardt 1988; Scott and Meyer 1991; Fogarty 1996). This

conformity has been articulated as compliance and convergence (Ashworth, Boyne, and Delbridge 2009). Here, compliance means that institutional changes comply with isomorphic pressures, while convergence means that an organisation resembles other units that have the same environmental characteristics. Entities respond to pressures from their institutional environment in order to main their legitimacy (Carpenter and Feroz 2001). The organisation considers these types of environments to conduct adaptation or isomorphism.

The concept of isomorphism identifies two types of isomorphism: competitive isomorphism and institutional isomorphism (DiMaggio and Powell 1983). Competitive isomorphism emphasises market competition, where organisations compete freely and openly in an environment (Sakib 2020). Organisations that are less competitive are excluded from the environment, since they fail to compete with other organisations (Johnston 2013). In contrast, institutional isomorphism is free from market competition. It is influenced by political power and institutional legitimacy, for social as well as economic fitness (DiMaggio and Powell 1983). In order to obtain legitimacy, organisations tend to reflect the practices conducted by others that have the same environmental conditions, and show homogeneity in their strategies, processes, and activities (Sakib 2020).

In developing countries, local governments are driven by institutional isomorphism, (Russo et al. 2012; Hudaya 2014). Institutional isomorphism is categorised into three main types: coercive isomorphism; mimetic isomorphism; and normative isomorphism (DiMaggio and Powell 1983). It is useful to describe each type briefly in turn. Coercive isomorphism is the process of adjusting exerted by formal and informal pressure, such as government regulations, other agencies, and the

community (DiMaggio and Powell 1983; Krell, Matook, and Rohde 2009). Based on the coercive isomorphism perspective, legitimate organisations are entities that can fulfil political rather than technical requirements (Ashworth, Boyne, and Delbridge 2009). Therefore, the implementation of coercive pressure tends to be symbolic conformity and fulfilment of regulations (Meyer and Rowan 1977; Cavalluzzo and Ittner 2004; Jamali 2010).

The second of the three main types of institutional isomorphism, mimetic isomorphism, is the process of institutions emulating or modelling the systems or activities of other institutions, when the first organisation faces challenges and uncertainties, in an effort to replicate the other institutions' success (DiMaggio and Powell 1983). Lapsley and Pallot (2000) found that local governments may seek legitimacy by resembling other local authorities, due to the similarities of function, tasks, and financing. However, Lawrence, Sharma, and Nandan (2009) and Hyndman and Connolly (2011) have noted that the use of mimetic isomorphism by local governments to undergo organisational change tends to be merely to achieve perceived legitimacy, rather than to achieve greater effectiveness and efficiency.

The third main type of institutional isomorphism, normative isomorphism, is organisational change driven by professionalism standards; typically characterised by formal education or professional training acquired by personnel or hired individuals from other entities (DiMaggio and Powell 1983). Obvious indicators of institutions' normative pillars include certification and accreditation, as this evidence shows legitimate identification of professional groupings (Scott 2005). With local governments consisting of multiple stakeholders, there is pressure on local government employees and officials to have relevant academic qualifications and

membership of professional associations (DiMaggio and Powell 1991; Mera 2000). Here, compliance with professional standardisation leads to the legitimation of the organisation.

By employing a more analytical approach, Scott (2013) differentiated three analytical pillars that compose institutions. He defined the principal dimensions, assumptions, mechanisms, and indicators of each element or pillar, as shown in Table 2.2. Each pillar is essential; and although those three elements may work in combination, especially in a stable social system, they still work with different mechanisms and processes (Järvenpää 2009).

Table 2.2: The Three Pillars of Inst	itutions
--------------------------------------	----------

	Regulative	Normative	Cultural-Cognitive
Basis of compliance	Expedience	Social obligation	Taken-for-grantedness Shared understanding
Basis of	Regulative rules	Binding	Constitutive schema
order		expectations	
Mechanisms	Coercive	Normative	Mimetic
Logic	Instrumentality	Appropriateness	Orthodoxy
Indicators	Rules, Laws, Sanctions	Certification,	Shared logics of action,
		Accreditation	Isomorphism
Affect	Fear Guilt/	Shame/Honour	Certainty/Confusion
	Innocence		-
Basis of legitimacy	Legally sanctioned	Morally governed	Comprehensible, Recognizable, Culturally supported

Source: (Scott 2013, p. 60)

The first pillar of institutional process is the regulative pillar. The regulative pillar refers to the creation and enactment of regulations and laws that act as coercive means (Palthe 2014; Stål 2017). All institutions have regulative rules, either formal or informal, that aim to regularise, shape, and constraint behaviour (Laffan 2001). Hence, the enactment of regulations requires conformity in order to maintain organisational legitimacy. The second pillar of institutions is the normative pillar. The normative pillar emphasises the roles of values and norms, as guides to social and political behaviour (Laffan 2001). From the normative pillar, institutional mechanisms

are normative, and the organisational changes are generated from a logic of appropriateness (Scott 2013). This means that the basis of legitimacy and the organisational changes are moral or social obligations, rather than formal or legal structures (Palthe 2014). Professional occupations are essential actors who maintain the normative pillar, since they control the day-to-day practices of the members of professions to align with values and moral obligations (Scott 2008).

Scott's (2013) third pillar of institutions is cultural-cognitive pillar. This focuses on organisational changes based on taken-for-granted cultural frames, symbols, signs, and rules that give legitimacy to aligned practices (Alexander 2012; Stål 2017). The cultural rules and frameworks are defined as 'the shared conceptions that constitute the nature of social reality and the frames through which meaning is made" (Scott 2013, p.57). Based on these cultural-cognitive aspects, organisations tend to adopt a change because they believe in it; even though it may not be enforced by regulative policy or workplace norms (Li et al. 2008; Palthe 2014).

5. Hypotheses Development

5.1 Hypotheses Related to the Relationship between Public Sector Governance and Local Government's Performance

As discussed in the earlier sections, this study focuses on financial transparency, accountability, and internal control for public sector governance. Financial transparency and accountability to the community are therefore essential for state governments to maintain, in order to preserve their legitimacy in the eyes of the community and ensure their programs and activities align well with community needs, values and norms (Martinez-Vazquez and Vaillancourt 2011). In Uganda, for example, the government empowered local communities to monitor the provision of

health services, and this was correlated with improved public health services, including a reduction in child mortality rates (Bjorkman and Svensson 2007). Likewise, in Uganda, the government's policy of publishing information about education grant programs (increased transparency) and increased monitoring by parents of grant-related spending on education was associated with increased numbers of enrolments in primary schools (Reinikka and Svensson 2005; Gaventa and Barrett 2010).

Consistent with this argument, financial transparency improves government outcomes and increases the efficiency of public money use (Hirsch and Osborne 2000; Ott, Bronić, and Petrušić 2014). As a key part of governance, an organisation's transparency has been positively related to its financial performance (Opler and Titman 1994; Tsoutsoura 2004; Rogers 2006; Darweesh 2015; Zagorchev and Gao 2015). Financial transparency has a positive impact on financial performance because transparency gives the public more power to monitor budget efficiencies and notice any corrupt or unethical budgetary practices (Alesina et al. 1999; Alt and Lassen 2006; Benito and Bastida 2009).

Recently, scholars have also identified that government transparency may not only improve government service and financial performance, but also has the potential to improve its environmental performance (Halkos and Tzeremes 2014). Transparency of governance is considered a prerequisite for sustainable management of resources (Sahide et al. 2016). In one study, government transparency was positively related to high air quality (Bernauer and Koubi 2009). However, Du (2015) and Gong (2018) warn that the relationship between government transparency and environmental performance is not always positive.

In all the hypotheses developed in this study, 'performance' will be evaluated according to three criteria: financial performance, service performance, and environmental performance. With regard to the aforementioned arguments, this study presents its first hypothesis below.

H1: Financial transparency is positively related to the performance of local governments.

Existing literature provides a guide to a positive and significant relationship between accountability and financial performance. Accountability has been posited to improve financial performance by reducing abuse of power, increasing efficiency and effectiveness, and financial productivity (Asselin 1995; Chye Koh and Woo 1998; Bovens 2007b; Rabrenović 2007; Harrison and Horngren 2008; Kim and Lee 2010; Han and Hong 2019). Haryanto (2010), Kis-Katos and Sjahrir (2017) likewise argue that accountability positively impacts public service delivery since it ensures public service quality is conducted properly. Deininger and Mpuga (2005) found that in Uganda, increased accountability linked to a significant improvement in public service quality, which the authors argued was due to the accountability mechanisms reducing government officials' tendency to use their power corruptly for individual gain. In contrast, a lack of local government accountability will impede the provision of public services (Baloyi and Beyers 2019).

Accountability in the public sector has also been found to have a significant influence on environmental development (de Silva, Yapa, and Vesty 2020; Handoyo 2018). Indeed, local government accountability is a core part of Shahib et al.'s (2020) sustainable paradigm. Shahib et al. (2020) argue that local governments should produce accountability reports that are publicly available, and which demonstrate

their compliance with environmental protection regulations and standards. Publicly available accountability reports can act as tools for communities to monitor the environmental activities of governments and demand change when environmental performance is not acceptable (Burritt and Welch 1997; Shahib et al. 2020).

With regard to the aforementioned arguments, this study presents its second hypothesis:

H2: Accountability is positively related to the performance of local governments.

The literature concerning the relationship between internal control and performance in local government, identifies that internal control has five interrelated components: control environment; risk assessment; control activities; information and communication; and monitoring activities (COSO 2013). Pujiono, Sukarno, and Puspitasari (2016) has reported that internal control has a positive and significant impact on the government's service and financial performance of North Maluku Province. The researchers found that when internal control activities were conducted by the public officials, there was a concurrent increase in the effectiveness and efficiency of their local budget, leading to an overall improvement in public services (Pujiono, Sukarno, and Puspitasari 2016). Similarly, Ong'unya and Abbey (2019) found that internal control had a significant positive impact on the quality of health service delivery of local government in Uganda.

A number of other studies have also investigated the relationship between internal control and institutional performance. For example, Berrone, Surroca, and Tribo (2007) reported a positive correlation between internal control and return on assets (ROA) and market value added (MVA). Strong internal control is likely to empower

personnel in an organisation to understand and ensure the achievement of institutional objectives. Likewise, previous studies have found a positive impact of the control environment (as a component of internal control) and risk assessment management on earnings, market value, and return on equity (Mackay and Moeller 2007; Al-Thuneibat, Al-Rehaily, and Basodan 2015; Florio and Leoni 2017). The control environment is the foundation of internal control and is essential to the effectiveness of local government's operational activities, including minimising the risk of corruption by public officials (Yurniwati and Rizaldi 2015; Nuswantara and Maulidi 2017).

Further studies also indicate links between internal control and an organisation's performance. Internal control has been reported to enhance economic efficiency because effective control activities can enable the retrieval of information from internal or external sources in a timely manner, and can facilitate an organisation's adaptive and corrective responses at all organisational levels, to maintain high performance (Bostan 2010). For the local government, Odoyo et al. (2013) reported a strong positive relationship between local authorities' information systems and the effectiveness and efficiency of revenue collection in Koya. In particular, the computerisation of council activities in Koya enhanced management integrity and enabled the timely collection of revenue.

When considering the literature on the relationship between internal control and an organisation's performance on environmental matters specifically, Huang and Huang (2020) report that of the five indicators of internal control, the most important for improving an organisation's environmental performance are its internal environment, information, and communication indicators. In Sri Lanka, effective internal control

has been reported to empower local governments to improve their environmental sustainability programs (Malalgoda, Amaratunga, and Haigh 2016). Similarly, other authors have emphasised the importance of effective internal control by organisations, to manage environmental issues such as pollution (Kareem, AlHassani, and Ali 2019)

Some of the literature which examines the relationship between internal control and an organisation's overall performance has investigated potential relationships between individual components of internal control (as mentioned earlier, the five components are: control environment; risk assessment; control activities; information and communication; and monitoring activities (COSO 2013), and local government performance. In one study in Indonesia, Tadeko (2017) examined the relationship between control environment and local institutions' service performance at Banggai District, and reported that 'control environment' and 'monitoring activities' both had insignificant effects on service performance for that particular local government. Similarly, for government health organisations in Ghana's upper western region, Ibrahim, Diibuzie, and Abubakari (2017) reported that 'control environment' and 'information and communication' had no significant effect on the organisations' financial performance.

Given the aforementioned arguments, this study presents its third hypothesis:

H3: Internal control is positively related to the performance of local governments.

5.2 Hypotheses Related to the Relationship between Financial Management Factors and Local Government's Performance

The performance of local governments is affected by a range of factors (Choi 2021). One of the most important factors in the local government's performance is financial management (Choi 2021). For financial management factors, this study focuses on financial flexibility, financial independence, and resource management. Sound financial management will enhance a government's ability to deliver its public service effectively (Kioko et al. 2011; Baloyi and Beyers 2019). Given this, researchers have recommended that local governments should be empowered to increase their ability to raise financial resources, and that this requires both financial flexibility and financial independence (Halim 2012; Fafurida and Pratiwi 2017). Facilitating local governments to generate their own revenues is one of the financial strategies that could alleviate the dependency of local authorities on central government, and promotes financial solvency.

When considering the effects that financial flexibility and financial independence may have on a local government's performance, it is useful to clarify what is meant by financial flexibility and financial independence. For local government, financial flexibility is indicated by both the local government's revenue capacity and its financial obligations to other parties (Ritonga 2014a); and also refers to having a sufficient cash balance (Soenen 2003). The second important factor in local government performance, financial independence, refers to the local government's revenue to meet local people's needs (Chapman 1999; Oulasvirta and Turała 2007; Sitek 2014; Kotarba and Kolomycew 2014; Cigu 2014; Scutariu and Scutariu 2015). Institutions have financial independence when they have the capacity to independently manage their financial resources, without being influenced excessively by external parties (Petkovska 2011).

With respect to the effects of financial flexibility and financial independence on local government performance, research on local public administration authorities in Rumania has reported that financial flexibility and financial independence of local government have positive impacts on local development (Scutariu and Scutariu 2015), by allowing local authorities to provide better public services. Organisations that maintain adequate cash reserves and manageable levels of debt are also in a better position to respond to investment opportunities, especially during economic downturns (Meier, Bozec, and Laurin 2013). The appropriate levels of financial flexibility and financial independence allow organisations to do this.

Research into local government in Canada and Colombia found that in both countries, financial flexibility and financial independence lead to sound financial capacity, which in turn is linked to lower infant mortality rates (Rubio 2011; Soto, Farfan, and Lorant 2012). Similarly, local governments with sound fiscal capacity are linked to increased life expectancy (Cantarero and Pascual 2008) and higher school enrolment rates at both primary and secondary school levels (Simatupang 2009). Interestingly, local governments with sound financial capacity also pay more attention to environmental sustainability (García-Sánchez, Frías-Aceituno, and Rodríguez-Domínguez 2013; Alcaraz-Quiles, Navarro-Galera, and Ortiz-Rodríguez 2014).

With respect to the relationship between resource management and local government performance, the literature argues that effective management of financial resources is crucial for effective performance. This includes maintaining a credible budget, with a little deviation between the approved budget and budgetary target achievement, with respect to both revenue and expenditure (Fölscher 2006).

Budget target achievement by government builds public trust; facilitates the achievement of macroeconomic goals; improves the efficiency and effectiveness of public service provision; and attracts investment in development programs (Sarr 2015). When governments manage their financial resources well, and achieve budget targets and plans, this increases investor trust and associated investment, leading in turn to strong local economies and job growth (Risti, Nicolaescu, and Tagaduan 2013; Sarr 2015).

Given the above arguments, this study presents its next three hypotheses:

- H4: Financial flexibility is positively related to the performance of local governments.
- H5: In local governments, financial independence is positively related to performance.
- H6: In local governments, resource management is positively related to the performance of local governments.

5.3 The Mediating Role of Financial Performance on Service and Environmental Performance

The last two hypotheses in this study to be generated in this study relate to the mediating role of financial performance on a local government's service performance, and on its environmental performance. When considering the relationship between financial performance and service, the financial performance of the local government is often indicative of the government's service performance, given that financial performance highlights a government's capability to finance capital expenditure for the provision of public services (Tamawiwy, Sondakh, and Warongan 2016). Sound financial performance can also attract investment; generate sufficient revenue to pay expenditures; provide community services; and sustain

public welfare – all of which are aspects of effective service provision (Batafor 2011; Dandara-Tabacaru and Danila 2017; Tampubolon 2019). Conversely, governments with poor financial performance may struggle to provide effective social, health, and economic services (Mishkin 1990; Hakkio and Keeton 2009; Bauducco, Bulíř, and Čihák 2011; Carlson, King, and Lewis 2011; Creel, Hubert, and Labondance 2015). Therefore, maintaining the financial performance of local governments is essential to ensure the quality of public service delivery.

With respect to the relationship between financial performance and environmental performance for local government, a number of studies have reported positive links between the two (Waddock and Graves 1997; Orlitzky, Schmidt, and Rynes 2003; Salama 2005; Konar and Cohen 2001; Al-Tuwaijri, Christensen, and Hughes li 2004). In government agencies in Czechoslovakia, for example, researchers reported that sound financial performance was linked with greater investment in technologies to reduce pollution (Earnhart and Lizal 2006). Increasing pressure on governments at all levels to protect the environment and biodiversity via sustainable practices has encouraged governments to adopt the concept of Triple Bottom Line (Elkington 1998); whereby entities accommodate the values of not only economic performance but also social and environmental performance in their business principles. Considering Triple Bottom Line, better financial performance has the capacity to facilitate better environmental performance (Waddock and Graves 1997).

Although organisations with greater funds should be able to divert relatively more funds to environmental management, in practice, this often does not occur. A number of empirical studies of private firms in developed countries have found a negative relationship between pollution prevention and financial performance

(Barbera and McConnell 1990; Brännlund, Färe, and Grosskopf 1995; Albrecht 1998; Sarkis and Cordeiro 2001; Wagner et al. 2002); potentially because profit is prioritised by these organisations at the expense of strong environmental management. Similarly, Ahlström and Ficeková (2017) observed that the more funds an organisation has, the less likely it is for the organisation to make investments that support the environment. Earlier economic theory has assumed that environmentally sustainable practices require net economic cost (Dyllick and Hockerts 2002; Spangenberg 2004). However, this assumption considers only the immediate term; and indeed, recent modelling of the economic effects of environmental policies indicates that even in the short term, the economic costs from poor environmental management will often be far greater than the economic costs of managing environmental risks effectively (Lewney 2021). The economic cost of zoonotic pandemics associated with poor management of deforestation and ecosystem health is a striking example (Everard et al. 2020). Economic theory is increasingly recognising that "economy or environment" is a false dichotomy, and that sustainable economies require sound environmental performance (Gopel 2016; Lewney 2021).

While the financial capacity of local government represents a substantial source of funds to support environmental sustainability, to date, this point has received limited attention from researchers and policymakers (Singer 2016). Previous studies have tended not to include financial performance as an integral part of environmental performance analysis, because of the (false) assumption that aims of sound financial performance and sound environmental performance will be in conflict with each other (Aras and Crowther 2008). However, as Aras and Crowther (2008) state, financial performance is an important factor to consider when analysing environmental

performance, as sound financial and environmental performance are not mutually exclusive.

Given the above arguments, this study presents its last two hypotheses:

- H7: In local governments, financial performance mediates the relationships between public sector governance, financial management factors, and service performance.
- H8: In local governments, financial performance mediates the relationships between public sector governance, financial management factors, and environmental performance.

The research hypotheses of this study are summarized in Table 2.3, as shown below.

Table 2.3: Summary of Research Hypotheses

Items	Hypotheses		
H1	Financial transparency is positively related to the performance.		
H2	Accountability is positively related to the performance.		
H3	Internal control is positively related to the performance.		
H4	Financial flexibility is positively related to the performance.		
H5	Financial independence is positively related to performance.		
H6	Resource management is positively related to the performance.		
(For H1-H6, 'performance' will be evaluated according to three criteria: financial performance, service performance, and environmental performance.)			
H7	Financial performance mediates the relationships between public sector		
	governance, financial management factors, and service performance.		
H8	Financial performance mediates the relationships between public		
	sector governance, financial management factors, and environmental		
	performance.		

6. Conclusion

This chapter has presented an overview of service and environmental performance in organisations, and described the factors affecting local government service and environmental performance. As outlined above, the literature on service and environmental performance in the public sector is limited, especially for developing countries. The present study aims to address this research gap. This chapter has then reviewed the application of institutional isomorphism in financial transparency, accountability, internal control, and financial management practices of local governments. The chapter also presented relevant conceptual models to explain the relationship between public sector governance (financial transparency, accountability, and internal control), financial management factors (financial flexibility; financial independence; and resource management) and service performance and environmental performance, through a mediating variable (financial performance). The next chapter presents the Indonesian local governments.
Chapter 3 Indonesian Local Governments

1. Introduction

As discussed in the previous chapters, this research has the main purpose to investigate the influences of public sector governance and financial management factors on service and environmental performance through the mediation of financial performance of local government districts in Indonesia. Therefore, it is useful to provide some background regarding Indonesian local governments to support an understanding of the context of this study. Local governments, as the closest tier of government to the public, are considered to have the capacity for better responsiveness than other levels of government to local needs. An appropriate amount of local government's responsiveness to the provision of public services, allowing local governments the financial flexibility and financial independence to meet local community needs.

This chapter comprises five sections, including this introduction. Section 2 describes public sector reforms at the district level in Indonesia; Section 3 discusses the service performance of districts; Section 4 presents the environmental performance of districts; and Section 5 presents the chapter's conclusion. The integration of Chapter 3 with the study's research objectives is shown in Figure 3.1.

Figure 3.1: The Integration of Chapter 3 with the Research Objectives



2. Public Sector Reforms at the District Level in Indonesia

Indonesia is chosen in this study due to several reasons. The first reason is that functioning of public sector is important for Indonesia because it provides a lot of services and it is a huge sector. If they do not function properly due to less of public sector governance, it will affect life of people (Nopyandri 2014; Subhan 2019). So many people are affected by public services of local governments because Indonesia is a big country. It has a large number of local governments that consist of 34 provinces, 416 districts, 98 cities, 7,024 sub-districts and it has 16.056 islands and 250 ethnic groups (BPS 2018; Fernandes 2019). Indonesia also has 939,000 square kilometres of forest (Menlhk 2018).

The second reason that Indonesia is chosen in this study is that there are relatively limited studies that address the influences of public sector governance and financial management on local government service and environmental performance in developing countries (Belal 2015; OECD 2017; Usang and Salim 2018; Sofyani, Riyadh, and Fahlevi 2020; de Silva, Yapa, and Vesty 2020). Mostly these sorts of literature were done in developed countries but in public sector of local government in Indonesia have not been addressed adequately. For these reasons, this study has selected Indonesia as an example of developing countries.

Indonesia has two tiers of government: central government, and local government. The central government administration is in Jakarta, the capital city of Indonesia. There are three levels of local governments: province-level, district-level, and municipality-level (Figure 3.2). Between 1966-1998, provincial governments, districts, and municipalities were largely seen as extensions of central government

as all local government policies and programs were determined by central government (Akbar 2011). However, from 1999 and onwards, (the 'reformation era'), central government decentralised its authority to the local governments. As a result, the local governments, especially the districts and municipalities, now have much greater autonomy to conduct their public service delivery, including the delivery of public works, public health, education, and environmental services.





According to the Indonesian Law 23/2014, provincial government in Indonesia has two functions (Iqbal 2017). First, it has the regional autonomy to provide services and manage resources without interference by the central government. Second, the provincial government acts as a representative of the central government, in that it has the responsibility to coordinate and monitor the development programs of districts and municipalities. This responsibility serves to integrate the development programs of districts/municipalities with similar programs at the national level. As a representative of central government, the provincial government also acts as a mediator or facilitator during disputes among districts and municipalities. For example, the centra government mediated during disagreements over regional borders (Arifin 2016), which have increased following the proliferation of districts or municipalities (Riyadi 2009; Sulistyono, Nuryadin, and Hadi 2014; Lewis 2017a).

This study focuses on the district level of government. Districts play an essential role to disperse political power, extend individual participation, promote efficient public services, and generate public welfare (Bekker and Jeffery 1989; Havenga 2002; Witing 2006). In the past in Indonesia (for example, during the Suharto era), a centralised government system has hampered the critical role of sub-national governments in the effective implementation of development programs (Aspinall and Fealy 2010; Kuncoro 2018). More recently the Indonesian government across all levels has been tasked with public sector reform, including decentralisation, marketisation, good governance, and direct election (Tambulasi 2007; Mimba, Helden, and Tillema 2007; Brinkerhoff and Brinkerhoff 2015). The following sections of the chapter will discuss local government decentralisation, public sector governance, financial management, and the performance of local government districts to examine the impact of these public sector reforms on districts in Indonesia (Figure 3.3).



Figure 3.3: The Impacts of Public Sector Reforms on Districts in Indonesia

(Source: adopted from Mir and Sutiyono 2013; Brinkerhoff and Brinkerhoff 2015)

2.1 Local Government Decentralisation

After the reform era in Indonesia in 1998, the central government applied a 'big bang' approach regarding decentralisation policy (Hofman and Kaiser 2004). Since then, the local governments have had more authority to deliver public goods and services and to manage their local resources. The reform era also empowered community participation in local development plans, to improve the alignment between development projects and local needs (Purba 2011; Wirawan, Mardiyono, and Nurpratiwi 2015; Laily 2015; Manaf et al. 2016; Winarni, Ahmad, and Suharno 2019).

As shown in Figure 4.3 above, decentralisation, as part of Indonesia's public sector reform agenda, improves resource allocation, public participation, accountability, and the responsiveness of local agencies (Jütting et al. 2005; Alawattage, Hopper, and

Wickramasinghe 2007). However, it is important to note that decentralisation does not always improve local government's performance (BAPPENAS and UNDP 2008; Lewis 2017a) where it occurs in the absence of sufficient capacity and resources. This can lead to poor quality of public service delivery by local governments (Kuncoro 2012).

2.2 Organisational Structure of a Local Government

As outlined above, Indonesian central government has decentralised most of its functions to the districts to enhance their performance (Kristiansen and Trijono 2005; Darmawan 2008; Sujarwoto 2012; Ostwald, Tajima, and Samphantharak 2016). Figure 3.4 below shows the organisational structure of district-level government in Indonesia. The *bupati* or head of a district is assisted by several subordinate individuals and agencies, including the local secretariat, the inspectorate, and relevant agencies, offices, and subdistricts. The local secretariat is responsible for assisting the head of the district to formulate local government policies and coordinate the administrative tasks. The inspectorate is an internal auditor, whose role is to assist the head of the district in supervising the various local government entities to conduct open and effective governance. The agencies conduct strategic services related to planning, finance, human capital, research, and development. The offices are responsible for carrying out local government affairs, including agriculture, education, health, forestry, transportation, and public works.



Figure 3.4: Organisational Structure of a District

Source: adopted from Indonesian Government Regulation 18/2016

The role of subdistricts is to support the coordination of government administration, public services, and community empowerment at the village level. There are two categories of subdistricts: Type A and B (Indonesian Government Regulation 18/2016). Type A refers to subdistricts that have heavy workloads, and Type B are subdistricts with lower workloads. 'Workload' is measured based on the size of the population, the total area of the subdistrict, and the number of villages within it. An officer with a lower rank (IIIb) will lead Type B subdistricts, while Type A subdistricts are led by an officer with rank IIIa. Type A subdistricts also have more sub-divisions than Type B, including more villages, which, as the lowest level of local government, are responsible for conducting administrative duties, development programs, and financial management at the village level.

According to the Indonesian Government Regulation 18/2016, the heads of the local agencies are directly responsible to the head of the district. These individuals must ensure that all employees of their relevant local agencies are performing their duties professionally. The recruitment of all employees of these local government agencies

must be open and transparent, as mandated by the Law 5/2014. In addition, the personal management is conducted based on the merit system (Wawanudin and Sudarno 2018; Wibowo 2020).

2.3 Public Sector Governance of Districts

Indonesia's reform agenda for government aims to encourage subnational-level governments to conduct good governance (Mardiasmo, Barnes, and Sakurai 2008). Improved governance of local governments has the capacity to improve local government's financial management, and reduce the risk of inequity, corruption or other misuses of public resources (Adiputra, Utama, and Rossieta 2018). As specified by the Indonesia's Law 1/2004, government entities should adhere to three main agendas of good governance: transparency, accountability, and internal control.

An effective and cheap method of disseminating information to the public regarding development programs and financial management is the internet (Adiputra, Utama, and Rossieta 2018), and in Indonesia, this is recognised and mandated by government regulations (Law 14/2008 and Government Regulation 61/2010). These two regulations ensure the rights of the people to have access to public information. This publication of government financial information facilitates communities to monitor the usage by government of its financial resources, and the implementation of development programs.

In addition to supporting district-level financial transparency and accountability, as outlined above, the Indonesian central government has set out ways by which to support district-level internal control. The Finance and Development Supervisory

Agency of Indonesia (2016) has developed a set of internal control guidelines to strengthen the internal control system of local governments in Indonesia. These guidelines stipulate several essential requirements for improving government internal control systems which include developing an internal control system plan, an appropriate and monitoring system for coordinating, monitoring, and recording the implementation of the internal control system.

2.4 Financial Management of Districts

Indonesian local governments have four main sources of revenue, as identified in the Law 23/2014, which includes local government's own local revenue, including from local taxes, local retributions, local companies, licencing, and income from interest on and sale of assets. They also receive revenue from the central government as part of fiscal transfers. Fiscal transfers aim to reduce both vertical imbalance (the fiscal imbalance between central and local governments) and horizontal imbalance (the mismatch of revenue powers and expenditure responsibilities among local governments) (Ahmad and Craig 1997; Slukhai 2003; Bird and Tarasov 2004; Martinez-Vazquez and Searle 2007; Patamasiriwat 2012; Eyraud and Lusinyan 2013). In Indonesia, fiscal transfers include three types of intergovernmental transfers: the general allocation fund; the special allocation fund; and shared revenue funds. The general allocation fund is a block grant transferred by central government to local governments to reduce the fiscal gap (Wulandari, Wahyudi, and Rani 2018). Subnational governments in Indonesia have tended to depend heavily on the central government to finance their local expenditures (Harvanto 2015; Butarbutar and Sofi 2017; Kristianto 2017). For example, one 2017 study found that 66.4% of districts depended on fiscal transfers from central government, with central

government transferring increasingly large proportions of the national budget to local government annually (±32.28%, 34.49%, and 37.27% of the national budget in fiscal years 2014, 2015, and 2016 respectively) (Mulyani 2017a). When considering how local governments in Indonesia spent their budgets, a very large proportion of the budget (an average of 38.54% in 2016) is currently spent on routine employee expenditures, including salaries, meetings, and official travel, while capital expenditure in the same year was only 23.9% of the total budget (DJPK 2019)

The subnational and local governments can borrow funds from external sources, such as from the central government, other local governments, domestic financial institutions (banking and non-banking), and domestic capital markets. Likewise, grants can be obtained from any domestic grant body: subnational governments have full authority to seek and obtain grants and loans from any domestic source. However, they are not allowed to seek or borrow funds directly from foreign sources without prior permission from the national Ministry of Finance. If a local government fails to follow central government regulations on this matter, the Ministry of Finance may delay or cancel the fiscal transfers of the local government.

Given the current inadequacy of local governments' budgets for development programs and other core community needs, central government in Indonesia has begun encouraging local governments to engage actively in revenue generation initiatives, to make local jurisdictions more self-sufficient financially (Krishnamohan 2016). This revenue may be obtained through intensification and extensification programs of regional taxes and levies (Wardhono, Indrawati, and Qori`ah 2013; Octovido, Sudjana, and Azizah 2014; Safitra 2019). Intensification strategies involve intensifying inspections and the collection of tax arrears, including increasing

efficiencies through technologies, improving the professionalism of tax officials, reviewing tax regulations, coordinating with other institutions, and enhancing taxpayer compliance (Marhani 2018; Safitra 2019). Extensification includes obtaining additional tax revenue by registering additional taxpayers (Nasution 2014; Levina 2015).

The high degree of financial flexibility and financial independence relates to local governments having full authority to generate funds via those three revenue sources and allocate this revenue for their spending needs without interference by the central government (Brodjonegoro 2018). As a comparison, the general allocation funds, which are intended to pay the salaries of civil servants, has high expenditure autonomy but average income autonomy (Nikijuluw 2012). The shared revenue funds offer subnational governments an average degree of both financial flexibility and financial independence, on both income and expenditure dimensions; given that some control is retained by the central government. For example, usage of the shared tobacco excise funds and reforestation funds are controlled by central government (World Bank 2010; Mulyani 2017b, c). However, usage of the shared funds derived from property tax and personal income tax are managed directly by each region (Directorate General of Fiscal Balance 2017). The shared revenue funds, foreign grants, and foreign loans have weak income and expenditure autonomy, due to centralised control and of these revenue types (Brodjonegoro 2018); for example, local government cannot receive the shared revenue funds and foreign resources without approval from central government.

3. The Service Performance of Districts

The fiscal decentralisation is expected to increase the level of financial independence of districts to support their provision of public services, reduce unnecessary bureaucracy, align public service delivery with local community preferences and needs, and increase efficiencies. These expectations are supported by the findings of Hamid (2013), who found that in Indonesia, decentralisation has had a positive impact on the quality or roads either built or upgraded by subnational governments as measured by an increase (following decentralisation) in the proportion of villages at district level which had asphalt roads. Likewise, local-level financial autonomy in Indonesia has reportedly increased the opportunities for people in villages to access health services such as hospitals and pharmacies (Skoufias, Narayan, and Kaiser 2011).

However, despite fiscal decentralisation to the districts, Indonesia is still a long way from delivering adequate access to health and associated environmental service. For example, although clean water and sanitation are basic public needs, 2018 data from Indonesia's Central Agency of Statistics (2018) reported that 13.4% of the Indonesian population had no access to clean water, and a massive 40% of the population had no access to proper sanitation, and still relied on the use of springs, rivers, ponds, and lakes as their water sources. Data from 2014 revealed that only 11% of the Indonesia population had access to piped water in their houses (Statistics Indonesia 2014). Researchers have raised concerns that inadequate water management means that even the piped water may be contaminated with faecal coliform (Bakker 2007); with diarrhea remains a significant public health threat in Indonesia, and a leading cause of death for children under five years old (Agtini et

al. 2005; Patunru 2015). This lack of health services is due, in large part, to insufficient spending by the government on health and associated environmental services; recent research reported that over a third of local governments (35%) allocated less than 10% of their total budget to the health sector (Mulyani 2017a). Other examples of the continuing inadequacy of health services at the community level include insufficient numbers of doctors at the district level (Meliala, Hort, and Trisnantoro 2013; Laksono, Ridlo, and Ernawaty 2020).

Indonesia's policy of decentralisation has also not, to date, overcome the tremendous challenges faced by districts regarding the provision of quality education. Recent research has identified that many district schools comprise dilapidated buildings that are unfit for purpose and often dangerous; some schools are in such poor condition that teaching occurs outside the classrooms (Sukasni and Efendy 2017). One study reported that a mere 24% of primary schools and 28% of junior schools possessed classrooms in good condition (Suhardi 2017). Many students (23% of all students) currently do not continue any formal education after Year 6 (Suhardi 2017). This low educational performance places Indonesia in the lowest rank of the assessed countries, behind Malaysia and Thailand (Schleicher 2018). Clearly, in the context of decentralisation, districts are choosing not to allocate sufficient funds for the education of the children in their communities. The Indonesian Constitution (UUD 1945) directs local governments to provide at least 20% of their local budget on education.

4. Environmental Performance of Districts

In Indonesia, decentralisation has increased the authority of districts to manage their natural resources (Dermawan, Komarudin, and McGrath 2006). In one common example, districts can generate local revenue by issuing forestry licenses and logging permits to companies to collect forest products (Palmer 2001; Tripathi 2015; Nasution 2017). Revenue collected from forestry licences can alleviate the financial dependency of districts on the central government (Saad 2001; McCarthy 2001). However, in the context of sustainable management of the environment, a policy of decentralisation in the absence of strong independent oversight will have increased the risk of conflict-of-interest, where local authorities now have the power to choose between safeguarding the environment and biodiversity for the future or choosing to approve environmental damage and loss to obtain short-term financial gain. In the absence of effective oversight, this is a pattern observed globally. In Indonesia, there is extensive evidence that this is occurring; as of 2003, two million hectares of native forest in Indonesia were under forestry licences (Palmer and Engel 2007) with over a million hectares of deforestation of native habitat and associated biodiversity loss every (Kusmana 2011; Nawir and Rumboko 2007). A key risk of devolving power to local agencies without effective oversight is the high likelihood of mismanagement, and this is a major issue for Indonesia; the Indonesian Corruption Eradication Commission has identified a wide range of problems with the mining permits issued to businesses and companies by local governments (Siombo 2014; Indreswari 2016).

Decentralisation in Indonesia has also increased the authority of local governments and the participation of farmers in the management of local water resources

(Harsanto and Simin 2006). This includes irrigation, with farmers – as major users of water – being empowered to undertake irrigation management to increase their awareness about the value of water resources and the benefits of effective management (Fulazzaky 2017; Rejekiningrum and Kartiwa 2018). In Indonesia, 4.78 million hectares (58.4%) of the national rice area is irrigated rather than rainfed (Suwandi et al. 2017). Therefore, the maintenance of irrigation is essential to maintain productivity and incomes for farmers (Hussain et al. 2006; Hadihardaja and Grigg 2011; Shrestha, Aquino, and Pandey 2014).

As stated in Regulation of The Government 20/2006, the management of Indonesia's primary irrigation canals (dams and reservoirs) and secondary canals (rivers and canals that deliver water to farmers and to the tertiary irrigation networks) is the responsibility of local government, while the management of the tertiary irrigation networks (the canals that bring water directly to farmland such as paddy fields and ponds) is the responsibility of the farmers' associations and water users' associations. This regulation also mandates that local governments must provide appropriate technical and institutional assistance to the farmers' associations and water tertiary irrigation networks. The aim of educating the farmers and water users' associations and providing farming guidance is to ensure the sustainable management of the tertiary irrigation systems (Pasandaran 2010; Sato, Yamaji, and Kuroda 2011; Ricks and Arif 2012).

Some recent research has found that the decentralisation of irrigation management in Indonesia is correlated with improved irrigation management at the district level (Daud 2006; Rondhi et al. 2020). For example, both water supply and rice production

increased after the implementation of community-managed irrigation systems, with water users' associations now able to rehabilitate and maintain their irrigational canals with the funds and resources they have been given (Rustinsyah and Prasetyo 2019; Alaerts 2020). Likewise, the farmers can now determine, collect, and manage their irrigation fees autonomously (Bruns 2004).

Despite the improvements, around 52% of irrigation in Indonesia (3.3 million hectares) remains in a damaged condition (Bahrun and Erawan 2015; Kesuma, Maryunianta, and Muda 2018) leading in turn to poorer agricultural productivity and welfare of farmers (Digdowiseiso, Sugiyanto, and Zainul 2018). This issue is related to a lack of appropriate guidance, technical capabilities, and knowledge received by water users' associations (Kesuma, Maryunianta, and Muda 2018). For example, Harsanto and Simin (2006) found that most people who are members of water users' associations do not have appropriate educational backgrounds or expertise to be of use in irrigation construction and maintenance. As a result, the farmers are forced to ask the contractor to conduct the irrigation maintenance; and the lack of expertise prevents irrigation management from working optimally. Several researchers (e.g., Harsanto and Simin 2006; Syaukat and Siwi 2009) have also reported frequent cases of unfair distribution of water and conflicts of interest among water users' association members; further exacerbating the ineffectiveness of irrigation management at the local level.

An important related point is that even when irrigation is successful, short-term increases in farm water supply and rice production do not necessarily indicate sustainable long-term water management. Indeed, they often indicate the opposite if long-term sustainability is sacrificed for short-term gain. In Indonesia, the risk of local

authorities focusing on water acquisition for immediate use is exacerbated by the increasing scarcity of water resources overall, given the demand for water from non-agricultural sectors such as industry and housing is rising rapidly (Pasandaran 2010; Syaukat, Arifah, and Minha 2014). Water scarcity and associated mismanagement not only pose risks for food security but also have major environmental and associated social and health impacts, including, increasingly, in the immediate term (Digdowiseiso, Sugiyanto, and Zainul 2018). Given these increasing risks, it is essential for Indonesia to put policies and practices in place to ensure the effectiveness of water management and broader environmental performance by local governments.

5. Conclusion

In Indonesia, the central government have decentralised its political, administrative, and financial functions. Decentralisation has meant that the heads of districts are now directly elected by the people; and districts now have more authority to manage their government functions. Districts finance their expenditure via several sources of incomes, including their own local revenue, the general allocation fund, special allocation funds, the shared revenue fund, and loans and grants. Decentralisation has triggered a proliferation of districts; the number of districts has increased by 55%. However, as outlined in this chapter, decentralisation has not led to many districts delivering sound public services. The delivery of quality education, health, and sustainable forestry and water management remains poor in most districts. With respect to the management of health, 39.28% of people living in districts still have no access to clean water, and 45.84% have no access to proper sanitation (Purba, Safitri, and Andianti 2017). When considering the adequacy of environmental

management, since decentralisation a high proportion (38.5%) of mining business permits issued by districts have not met the requirements of clean and clear mining business permits (Indreswari 2016)

As well, two-thirds of districts (66.8%) remain highly dependent on fiscal transfer from central government (DEPIKD 2018). Several authors have argued that high financial dependence on central government will have a negative impact on district performance (Boetti, Piacenza, and Turati 2012; Eyraud and Lusinyan 2013; Kyriacou, Muinelo-Gallo, and Roca-Sagalés 2017; Dewata et al. 2018). In Indonesia, this dependency of districts on funding from the central government is exacerbated by the current weakness of internal control.

To address these issues and increase local government performance, as mandated by Government Regulation 60/2008, Law 14/2008, Law 23/2014, and Presidential Regulation 29/2014, the Indonesian central government will need to better manage subnational governments to strengthen their public sector governance and financial management. The next chapter will discuss the research methodology.

Chapter 4 Research Methodology

1. Introduction

The aim of this study is to examine the effects of public sector governance and financial management factors on service and environmental performance of local government districts in Indonesia. In this study, public sector governance is represented by financial transparency, accountability, and internal control, while the financial management factors to be examined are financial flexibility, financial independence, and resource management. The mediating effects of financial performance on service and environmental performance are also examined. Public sector governance and financial management factors of local governments are expected to enhance the financial performance of local governments, which in turn will influence the service and environmental performance of local governments.

This chapter presents the research methods used in this study. The secondary data, correlation analysis, multiple regression analysis, and bootstrapping methods are used to achieve the research objectives and to answer the four research questions. Chapter 3 consists of four sections, including this Introduction. Section 2 presents the research methods; Section 3 describes the definitions and measurements of the variables; and Section 4 presents the chapter's conclusion. Figure 4.1 depicts the integration of Chapter 4 with the research objectives.



Figure 4.1: The Integration of Chapter 4 with the Research Objectives

2. Research Methods

This study employs quantitative methods and secondary data-based research to examine the following research questions: (1) the effects of public sector governance and financial management factors on service and environmental performance; and (2) the mediating role of financial performance on the relationship between public sector governance, financial management factors, and service performance and environmental performance. Heaton (2004) defines secondary data analysis as 'a research strategy which makes use of pre-existing quantitative data or pre-existing qualitative research data for the purposes of investigating new questions or verifying previous studies' (p.16). From this definition, secondary data analysis refers to analysing and reusing data previously collected by other researchers or agencies for purposes other than the study at hand (Hox and Boeije 2005; Smith et al. 2011; Andrews et al. 2012; Todaro and Smith 2006; Chenhall and Langfield-Smith 2007). These secondary data sets are used to answer the research questions for this study.

Secondary data analysis is a well-established methodology that is becoming more commonly applied for academic purposes, with advantages including its efficiency with respect to time and financial resources, and its typical capacity to include large quantities of data (Cowton 1998; Greenstein 2006; Smith et al. 2011; Johnston 2017). Use of pre-existing large data sets has been reported to expedite research findings (Doolan et al. 2009) including for the use of policy makers and government policy implementation (Magee et al. 2006).

Secondary data sets may be based on either published data, or on original data that are available from various unpublished by reliable sources, whether digital or in print,

including books, reports, articles, archives, and the internet (Church 2002). The digital age has made available an enormous amount of published data online, such as census information, official statistics, annual reports, and socioeconomic indicators, which are easily accessible to researchers. In accounting research specifically, such sources of secondary data have greatly increased the opportunity to conduct quantitative research (Heaton 2004).

Cowtown (1998) classified secondary data sources into five categories. The first category is data sets published by governmental and regulatory bodies. The second category is data from other organisations, such as annual reports, data on operating activities, and statements from chairs. These data sets tend to be easily accessed from organisations' websites. The third category of data source is the press; including newspaper articles, magazines and other sources. The fourth data source comprises data from other academic researchers; for example, meta-analysis of the data and results from other research. The fifth category of the data source is private sources. These data sets are gathered from other individuals or agencies based on negotiation.

In the Indonesian context, according to the Law 14/2008 regarding the openness of public information, all government agencies have obligations to provide and publish accurate, true, and non-misleading public information to the public. Further, all government agencies are expected to utilise electronic media to disseminate this public information efficiently, and to enable people to access the data easily. This commitment is in line with Article 28F of the 1945 Constitution of the Republic of Indonesia, which guarantees everyone the right to deliver and communicate their ideas; to obtain information for their own personal development and social

environment; and to seek, obtain, possess, store, process and deliver information using all types of available channels. One caveat, as per the Law 14/2008, is that government entities may restrict access to data that could endanger the security of the country, or data related to personal rights and confidential job positions.

The broad commitment of the Indonesian governmental entities to publish public information enhances the opportunity for researchers to conduct research using secondary data. Given the accessibility and other advantages of secondary data to support research, as outlined above, this study uses secondary data sources to investigate the effects of public sector governance (financial transparency, accountability, internal control), and financial management factors (financial flexibility, financial independence, and resource management) on service and environmental performance in local government. Existing data published by Indonesian governmental institutions are also utilised to analyse the mediating effect of financial performance on the relationships between public sector governance.

2.1 Data Sources

This study applies secondary data relating to the year 2016, which was collected from publicly available datasets through various Indonesian government agencies. During the data collection period (2017 to 2018), the latest data available in Indonesian local governments was 2016. Public sector authorities collect data and internally keep it but they do not release every year but they release every three or four years. In that case, when the data collection was done in 2018, 2016 data was

the latest data and available at that time. This is quite normal in public sector cases. Thus, 2016 data is recent data up to now.

All the collected secondary data sets are freely available without cost, as mandated by Law 14/2008 dealing with public information openness. This research analyses data collected at one specific point in time (financial year 2016) because this research constitutes a cross-sectional study. The data sets related to government agencies' financial performance, financial flexibility, financial independence, and resource management indicators were collected from the audited financial statements of local governments. The audited financial statements consisted of balance sheets, cash flows statements, and budget realisation statements. These data sets were obtained from the Supreme Audit Board.

Likewise, the secondary data regarding service performance indicators, which included information about sanitation, healthy drinking water, vaccination rates for children under five years old, childbirth medical assistance, and school infrastructure quality, were publicly available data sets collected from the Indonesian Ministry of Finance and the Ministry of Education and Culture. The secondary data sets for the environmental performance indicators of local governments, which included information about forest rehabilitation, forest quality, irrigation, and green open space, were collected from the Ministry of Home Affairs.

For information about internal control, this study used secondary data regarding internal control indicators of local governments obtained from the Finance and Development Supervisory Agency of Indonesia. These data sets regarding internal control indicators comprised five measures: control environment; risk assessment;

control activities; information and communication; and monitoring activities. The data regarding financial transparency indicators were collected from financial information uploaded to the official websites of local governments; and the administrative accountability data was collected from the Indonesian Ministry of Administrative and Bureaucratic Reform. Data regarding population size, poverty rates, and gender diversity, which were used as control variables, were collected from the Indonesian Central Agency of Statistics, the Ministry of Finance, and the State Personnel Agency respectively.

2.2 Research Samples

Indonesia, as a multi-tiered unitary state, comprises thirty-four provinces, 416 districts, and ninety-eight municipalities (Pandjaitan 2016). Districts and municipalities constitute the lowest level of regional entities of the state (Figure 4.2). The heads of districts and municipalities are directly elected by the local people, every five years. There are several differences between districts and municipalities in Indonesia. One difference is that districts have wider regional areas than municipalities (based on the regulation of the Ministry of Home Affairs 137/2017).

Figure 4.2: Structure of the Indonesian Local Governments



According to Indonesian Law 23/2014, districts have access to three main sources of revenue to finance their public service expenditure: own-source revenue, intergovernmental transfers, and other local revenue. The own-source revenues consist of local taxes, local levies, profits of locally-owned companies, and profit-sharing cooperation with third parties; such as interest income and asset sales. The intergovernmental transfers are the main revenue resources of districts, and comprise two types of transfer funds: transfer funds received by districts from the central government (balance funds and village funds); and transfer funds acquired by the districts from the other local governments (revenue sharing funds). Other local revenues include all income other that own-source revenues and intergovernmental transfers; this involves primarily grants and emergency funds. The financial resources of districts has been discussed in more detail in Chapter 3.

This study will focus on the district level due to two reasons. First, districts have a broader range of public service facilities than municipalities, including public schools, local health centres, and subdistrict offices (PDSPK 2016; Budijanto 2018). The second reason is Indonesia's agricultural and forestry sources are mainly produced in district area (Andoko 2019). The sampling method of this study is purposive sampling, and the data of all variables are the data of 2016 with the following sampling criteria. The research uses only the data of districts (*kabupaten*) that acquired an unqualified or qualified opinion of a financial statement issued by the Supreme Audit Board. The disclaimers opinions and adverse opinions are not used because such financial statements would reduce the reliability of data (Ritonga 2014b). The financial statements of the adverse and disclaimers opinions were not fairly presented based on the accounting principles of Indonesia in all materials and financial aspects (IAI 2001). Table 4.1 below shows that the number of districts that

had an unqualified or qualified opinion were 391 local governments. However, there are 153 local governments that have data of service performance, environmental performance, financial performance, internal control, accountability, and financial management factors required in this study. This study removes 28 observations because they are considered as outliers measured by a z-score greater than 3 or less than -3(Hair et al. 2011; Ghozali 2016; Tabachnick and Fidell 2014). As a result, the data used for this research are 125 observations.

Table 4.1: The Number of Samples Used in the Study

No	Existing Data	Number of
		Data
1.	Number of existing districts (<i>kabupaten</i>) with an unqualified or qualified opinion	391
2.	The availability of data	153
3.	Outlier Data	28
4.	Data utilised	125

2.3 Data Analysis

This study used SPSS Statistics version 24 to analyse the research data, including to generate descriptive statistics and undertake correlation tests and multiple regression analysis, as described in the following paragraphs. Version 3 of PROCESS macro (Hayes 2018) was used for the bootstrapping methods to analyse the mediation effects of financial performance on the relationships between public sector governance, financial management factors, and service performance, and environmental performance.

2.3.1 Testing Validity and Reliability

Validity refers to how well the instruments used in a study define the intended concept, while reliability refers to the degree of internal consistency of the measures

(Hair, Black, and Babin 2010). The validity and reliability of tests are interconnected, in that the validity of a measurement depends on its reliability (Nunnally and Bernstein 1994). The validity of a test is concerned with homogeneity or unidimensionality, while reliability test describes the internal consistency of instruments (Sekaran and Bougie 2010; Williams, Onsman, and Brown 2010; Bonett and Wright 2015; Shirali, Shekari, and Angali 2018; Ghozali 2016; Swank and Mullen 2017). In this study, the construct validity and unidimensionality are tested using the Measure of Sampling Adequacy and Bartlet's test of Factor Analysis (Williams, Onsman, and Brown 2010; Shirali, Shekari, and Angali 2018) or Pearson product moment correlation analyses (Piland et al. 2003; Ghozali 2016; Swank and Mullen 2017), while the reliability test is conducted by using Cronbach's alpha. Cronbach's alpha is the most common measure of internal consistency (Bland and Altman 1997; Hinton et al. 2011; Bonett and Wright 2015).

2.3.2 Descriptive Statistics and Correlation Tests

This study uses SPSS to present the descriptive statistics regarding service performance, environmental performance, financial performance, public sector governance, and financial management factors. The descriptive statistics analyses include calculating means, medians, standard deviations, skewness and kurtosis which together aim to provide baseline information about the samples in the research. In addition, correlation tests will be used to analyse the direction and strength of significant relationships among variables used in this research. A positive correlation means that as values of one variable increase, values of another variable increase as well (Allen, Bennett, and Heritage 2019).

2.3.3 Multiple Linear Regression and Bootstrapping Analysis

The ordinary least square (OLS) regression analysis and Bootstrapping method are applied to test the models of the study as shown in Table 4.2 below. Multiple regression analysis is used to examine the effects of public sector governance and financial management factors on service, environmental, and financial performance of local governments. Multiple regression analysis is a method that investigates the functional relationships and the magnitude of effects between dependent and independent variables. The dependent variable (also known as the response, explained, predicted, regressed, criterion, endogenous, controlled, outcome, and consequent variable) is represented by Y, while independent variables (also known as stimulus, explanatory, predictor, regressor, antecedent, exogenous, control, or covariate variables) is represented by X (Yan and Su 2009; Gujarati and Porter 2009; Chatterjee and Hadi 2015; Chatterjee and Simonoff 2013; Hayes 2018).

Table 4.2: Models and Equation	ons
--------------------------------	-----

Model	Equations		
Model 1	Public Sector Governance and Service Performance, Mediated by Financial Performance		
	$SP = \beta_0 + \beta_1 IC + \beta_2 FT + \beta_3 AC + \beta_4 Pop + \beta_5 Pov + \beta_6 Gdr + \epsilon_{SP}$		
	$FP = \beta_0 + \beta_1 IC + \beta_2 FT + \beta_3 AC + \beta_4 Pop + \beta_5 Pov + \beta_6 Gdr + \varepsilon_{FP}$		
	$SP = \beta_0 + \beta_1 IC + \beta_2 FT + \beta_3 AC + \beta_4 Pop + \beta_5 Pov + \beta_6 Gdr + \beta_7 FP + \epsilon_{SP}$	Research	
Model 2	Financial Management Factors and Service Performance, Mediated by Financial Performance	Question 1 and 2	
	$SP = \beta_0 + \beta_1 FF + \beta_2 FI + \beta_3 RM + \beta_4 Pop + \beta_5 Pov + \beta_6 Gdr + \epsilon_{SP}$		
	$FP = \beta_0 + \beta_1 FF + \beta_2 FI + \beta_3 RM + \beta_4 Pop + \beta_5 Pov + \beta_6 Gdr + \varepsilon_{FP}$		
	$SP = \beta_0 + \beta_1 FF + \beta_2 FI + \beta_3 RM + \beta_4 Pop + \beta_5 Pov + \beta_6 Gdr + \beta_7 FP + \epsilon_{SP}$		
Model 3	Public Sector Governance and Environmental Performance, Mediated by Financial Performance		
	$EP = \beta_0 + \beta_1 IC + \beta_2 FT + \beta_3 AC + \beta_4 Pop + \beta_5 Pov + \beta_6 Gdr + \varepsilon_{EP}$		
	$FP = \beta_0 + \beta_1 IC + \beta_2 FT + \beta_3 AC + \beta_4 Pop + \beta_5 Pov + \beta_6 Gdr + \varepsilon_{FP}$		
	$EP=\beta_0 + \beta_1IC + \beta_2FT + \beta_3AC + \beta_4Pop + \beta_5Pov + \beta_6Gdr + \beta_7FP + \varepsilon_{EP}$	Research	
Model 4	Financial Management Factors and Environmental Performance, Mediated by Financial Performance	Question 3 and 4	
	$EP = \beta_0 + \beta_1 FF + \beta_2 FI + \beta_3 RM + \beta_4 Pop + \beta_5 Pov + \beta_6 Gdr + \varepsilon_{EP}$		
	$FP = \beta_0 + \beta_1 FF + \beta_2 FI + \beta_3 RM + \beta_4 Pop + \beta_5 Pov + \beta_6 Gdr + \varepsilon_{FP}$		
	$EP = \beta_0 + \beta_1 FF + \beta_2 FI + \beta_3 RM + \beta_4 Pop + \beta_5 Pov + \beta_6 Gdr + \beta_7 FP + \varepsilon_{EP}$		
SP = service performance; EP = environmental performance; FP = financial performance; IC =			
internal control; FT = financial transparency; AC = accountability; FF = financial flexibility; FI =			
financial independence; RM = resource management; Pop = population size; Pov = poverty rate; Gdr			
= gender	diversity; ε = random error		

Generally, the regression model uses the ordinary least square (OLS) method. There are three types of regression; simple linear regression; non-linear regression and multiple regression (Yan and Su 2009). Simple linear regression models the relationship between two variables, X and Y with the equation $Y_i = (b_0 + b_1X_i) + \varepsilon_i$ where Y_i is the dependent variable, X_i is the independent variable, b_0 is the intercept or regression constant, b_1 is the regression coefficient or regression weight, and ε_i is the random error or residual. The third type of regression, multiple linear regression, is the type used in this study. Multiple linear regression is a linear regression model that comprises one predicted variable and more than one predictor variable, with the following statistical equation: $Y = (b_0 + b_1X_1 + b_2X_2 + ... + b_kX_k) + \varepsilon$ where Y is the dependent variable, X_1 , X_2 , X_k are independent variables, b_0 is the intercept or regression constant, b_1 is the regression coefficient or regression weight, and ε_i is the random error or residual.

As noted above, this research also examines the mediating effects of local government's financial performance on the relationships between public sector governance, financial management factors, and public service and environmental performance. The mediation analysis investigates the causal relationship between one or more independent variables and a dependent variable, through a mediator variable (Pardo and Román 2013; Hayes 2018). For these analyses, this study uses Bootstrapping methods via Hayes' PROCESS macro in SPSS version 24.

This current research proposes the indirect, total, and direct effects of public sector governance and financial management factors on the service and environmental performance variables, based on the results of the bootstrapping analysis generated by PROCESS macro. Here, 'indirect effect' refers to the condition when the variables of public sector governance and financial management factors predict the service or environmental performance variable through financial performance as a mediating variable. This means that an indirect effect is the combined effects of the relationships between an independent variable and a mediating variable, and between the mediating variable and the dependent variable (Edwards and Lambert 2007; Meule 2019). The 'total effect' is the effect of public sector governance and financial management factors as the independent variables on service and environmental performance as the dependent variables when financial performance

as the mediating variable is not present in the model. If the mediator variable is included, this relationship is called the direct effect (MacKinnon 2011; Lancee et al. 2015; Agler and De Boeck 2017; Meule 2019).

2.4 Assumptions of the OLS regression and Goodness of Fit of the Model

Prior to using the multiple regression models as shown in Table 4.2 above, several assumptions need to be tested: the assumptions of multicollinearity, autocorrelation, linearity, homoscedasticity, and normality of residuals. These assumptions of regression analyses aim to meet the criteria of the best linear unbia4sed estimators (Cohen et al. 2013; Gujarati and Porter 2009; Field 2009; Hair, Black, and Babin 2010; Tabachnick and Fidell 2014; Hayes 2018).

2.4.1 Multicollinearity Test

Multicollinearity constitutes the substantial collinearity among the independent variables. Several guidelines should be considered whether a regression model has multicollinearity or not (Gujarati and Porter 2009; Field 2009; Hair, Black, and Babin 2010; Cohen et al. 2013; Ghozali 2016). For example, the existence of a correlation of the regressor of 90 or higher indicates multicollinearity (Ghozali 2016). Other scholars argue that as a rule of thumb, a bivariate correlation of .80 or higher might lead to a problem (Gujarati and Porter 2009; Whitley and Kite 2013). Moreover, multicollinearity is indicated by a tolerance value > .10, or the Variance Inflation Factor (VIF) < 10 (Allen, Bennett, and Heritage 2019). The results of the multicollinearity test of the models used in this study will be presented in Chapter 5 and 6.

2.4.2 Autocorrelation Test

Autocorrelation is "the internal correlation between members of a series of observations ordered in time or space" (OECD, 2008b, p.37). This study uses the Durbin-Watson coefficient to test for the presence of autocorrelation in the regression model. The acceptable Durbin-Watson coefficient is between 1.5 and 2.5 (Zakerian and Subramaniam 2009; Atici 2011; Garson 2012). Field (2009) states that if the Durbin Watson value is less than one or greater than three, this indicates an unacceptable level of autocorrelation. The results of the autocorrelation tests of the models applied in this current research will be presented in Chapter 5 and 6.

2.4.3 Tests of Linearity and Homoscedasticity

Several methods can be used to test linearity, such as the graphical method by examination of the residuals, Ramsey's regression error specification (RESET) test, and Lagrange Multiplier (LM) test (Ghozali 2016). Garson (2012) states that the linearity test can be assessed by the scatterplot of the standardised residual graph. As well, the scatterplot results can be used to check homoscedasticity (Gujarati and Porter 2009; Field 2013). If the plots are scattered and there is no pattern of scatterplot graph, the assumptions of linearity and homoscedasticity are fulfilled. This study uses the graphical method by examination of the residuals to check the linearity and homoscedasticity. The results of linearity and homoscedasticity tests of this current research will be presented in Chapter 5 and 6

2.4.4 Test of Normality of Residuals

A normal probability distribution is a symmetrical bell-shaped curve with mean $\mu = 0$ and a standard of deviation or variance $\sigma^2 = 1$ (Stine and Foster 2011). The normality test aims to assess that the residual around the regression line is expected to have a normal distribution (Brown and Caylor 2006). Normally distributed data could be assessed by several methods (Das and Imon 2016; Ghozali 2016). Histogram and probability plots are methods that are very common as visual examination to investigate the normality. The second method to check the normality is based on the summary statistics, namely skewness and kurtosis. The skewness determines the lack of symmetry, while the kurtosis measures a deviation from normality and assesses the prevalence of outliers (Stine and Foster 2011).

Another statistical method that measures the normality of residuals is the Kolmogorov-Smirnov (KS) test (Drezner, Turel, and Zerom 2010). The presence of normality is indicated by the significance of the Kolmogorov-Smirnov (KS) test. As a rule of thumb, the model is not normally distributed if Sig. < 0.05. This study uses the Kolmogorov-Smirnov (KS) test to investigate the normality of residuals. The results of normality of residuals test of the models applied in this current research will be discussed in Chapter 5 and 6.

2.4.5 Assessing the Goodness of Fit of the Model

Statistical procedures are used to measure the goodness of fit in the ordinary least squares (OLS) multiple linear regression, such as R-squared, Adjusted R-squared, and F-test (Hagquist and Stenbeck 1998; Hair, Black, and Babin 2010; Ghozali 2016; Hayes 2018). The coefficient of determination or R^2 is a widely used

goodness-of-fit measure (Cameron and Windmeijer 1997) which its value ranges from zero to one. The value of R² of 1 indicates that the regression model is good. The R² value closes to zero indicates that the proposed model is poor. Likewise, Ftest is conducted to evaluate the goodness of fit of the model. The F-test aims to show whether a group of variables are jointly significant or not. If the p-value for the F-test is less than the significance level, the null hypothesis is rejected and accepts Ha. It means that the regression model is good. If the calculated F-value is larger than F-statistics, the regression model is good.

3. Robustness Checks

This study uses several types of robustness checks. The first type of robustness check is stepwise regression analysis. The stepwise regression analysis is used to check the robustness of the multiple linear regression models and investigate which combinations of variables best predict local government performance. The stepwise multiple regression analysis is the combination of forward selection approach and backward elimination method that aims to investigate the order of importance of variables and to determine the best-fit combination of independent variables indicated by the higher R square (Ghani and Ahmad 2010; Studenmund 2014; Grignon 2019). The predictors that closely determine the dependent variable will be included in the model, while the predictors that have insignificant effects on the dependent variable will be excluded from the model.

There are several procedures to follow with the stepwise regression analysis (Burkholder and Lieber 1996; Pasha 2002; Field 2009). The stepwise regression proceeds the first step or model of the simple regression by entering the variable that
has the highest relationship with the dependent variable. The significance of the variable is measured against the F-distribution. The next step is conducting the partial correlation between the dependent variable with the remaining predictors to find the most highly correlated predictor. Then, the second highly correlated predictor is regressed with the previously independent variable of the first model. If these predictors significantly improve the model to predict the dependent variable, the same process is repeated for the next regression equations. While the variables that have a non-significant contribution to the model, they are excluded from the regression equation. The process of stepwise regression is iterated until no more additional predictors that have no statistical significance to the model (Lang 2016).

A second type of robustness check is Monte Carlo Simulation developed by MacKinnon, Lockwood, and Williams (2004). Monte Carlo Simulation is performed to examine the robustness of the mediation analysis. Concerning the Monte Carlo approach, it is more recommended than the Sobel Test for indirect effect analysis (Rungtusanatham, Miller, and Boyer 2014; Schoemann, Boulton, and Short 2017; Caron 2019). The Sobel Test may result in the inappropriate conclusion of the mediation effects due to the normal distribution assumption of a specific indirect effect (Cheung and Lau 2008). Additionally, Rungtusanatham, Miller, and Boyer (2014) argued that the Sobel test might result in misleading conclusions of mediation effects since the product of two normally distributed variables is a non-normal distribution. Therefore, the Monte Carlo method is comparable to other approaches for constructing the confidence intervals of the mediation effect that is not affected by the normality assumption of the sampling distribution (Preacher and Selig 2012; Zhang 2014).

4. Definitions and Measurements of Variables

4.1 Developing Composite Indices

A composite index is an aggregate measurement that summarises a set of indicators or multi-dimensional realities without collapsing the significance of the informational basis (JRC 2008; Dharmawardena, Thattil, and Samita 2017). The construction of composite indices is essential since the composite index tends to be easier to read and interpret than many separate indicators (Booysen 2002; Farrugia 2007). The composite index has been used to assess the performance of public sectors, such as health care services, social services, education, and local government entities (Freudenberg 2003). There are several methods of developing composite indices, such as Human Development Index (UNDP 2014), Climate Change Performance Index (Burck, Bals, and Ackermann 2009), Happy Planet Index (Abdallah et al. 2012), Environmental Performance Index (Hsu, Lloyd, and Emerson 2013), Multidimensional Poverty Index (OPHI 2018), and Financial Performance Index (Ritonga 2014a; Alam 2019).

This study adopts several steps to construct composite indices as recommended by the Organisation for Economic Co-operation and Development (OECD) (2008) and Mazziotta and Pareto (2013). The first stage is determining the theoretical concept that provides the foundation for selecting relevant indicators to construct service performance, environmental performance, and financial performance indices. The second stage is identifying relevant and essential indicators for constructing the composite index of these variables. This step is very important since it is a guide that could assess whether an indicator is part of the composite index or not. The third stage is selecting indicators which are relevant to the variables being measured. The fourth stage is conducting transformation for indicators that have highly skewed data. It also aims to reduce the difference between the maximum and the minimum value. The sixth stage is choosing the Min-Max technique for the normalisation of data. The equation Min-Max method is the actual value minus minimum value divided by the maximum value minus minimum value. Then, weighting of variables derived from the principal component analysis or equal weights. The final step of composite index construction used in this study is selecting linear aggregation as an approach that aggregate the values of all indicators based on the weight of each indicator.

In this research, service performance, environmental performance, financial performance, financial flexibility, financial independence, resource management, financial transparency, and internal control are related to composite indices, while accountability is related to measures of variable, as shown in Table 4.3 below. The measurement of the variables used in this study will be discussed in more detail in the following subsections.

No.	Methods of Measurement	Variables
1.	Composite Index	- Service Performance
		- Environmental Performance
		- Financial Performance
		- Financial Flexibility
		- Financial Independence
		- Resource Management
		- Financial Transparency
		- Internal Control
2.	Measure of Variable	- Accountability

Tahla .	1 2.	Mothode	of Moa	euromont
י מטוכי	4.0.	INCLINUS		อนเธเมธเม

4.2 Service Performance

Local governments, as the closest entity of the government, play essential roles in efficient public services delivery. Academics and practitioners have various approaches to evaluate the achievement of public service performance at local authorities because countries have different constitutions that regulate the service functions provided by the local governments. For example, Kelly and Swindell (2002) evaluated the service performance of several cities in the United States based on four service categories, inter alia police services, fire and emergency medical services, road maintenance, and parks and recreation services. Zafra-Gómez, López-Hernández, and Hernández-Bastida (2009) and Torres, Pina, and Yetano (2011) used basic services indicators to evaluate the service performance of local governments in Spain, such as roads and highways, public parks, street lighting, waste management, police services, fire services, animal control, housing, and libraries. In the United Kingdom and Canada, service performance of local governments assessed by several services provision, such as the primary and secondary schools, fire services, police services, road maintenance, waste management, subsidised housing, cultural services, waters supply, licenses, parks, library services, health care services, and social services (Pollanen 2005).

In the Indonesian context, based on the Law No.23 of 2014, there is a separation between the public services provided by the central government and sub-national authorities. The central government focuses on six matters, videlicet political foreign affairs, defence, security, judiciary, national monetary and fiscal, and religion. While local governments have three dimensions of responsibilities of the public services as outlined below:

- I.Compulsory and Basic:Education, health, spatial planning, publicPublic Servicehousing, community safety, social
- II. Compulsory and Non-Basic Public Service
 Workforce, women empowerment, food, land, environment, public administration and civil registration, village empowerment, family planning, transportation, communication and information, cooperative, investment, sport, statistics, culture, library, and archives
- III. Optional Governmental : Maritimes and fishery, tourism, agriculture, Services forestry, energy and mineral resources, trade, industry, and transmigration.

This study applies a composite index of service performance developed from five indicators of basic public services as determined by the Minister of Finance Regulation 266/2015. The first indicator is the percentage of households with access to sound sanitation. This study makes a comparison between the number of households that have access to sound sanitation as a numerator and the total number of households in a district as the denominator. In 2010, it was found that 46 percent of the population in Indonesia accessed to unhealthy sanitation (Patunru 2015). Also, in Indonesia, 29 to 31 million people especially in rural area were still practicing open defecation (Odagiri et al. 2017). Therefore, the local governments are encouraged to overcome this problem as mandated by Government Regulation 66/2014 and Minister of Health Regulation 3/2014.

The second indicator of service performance used in this study is the percentage of households with access to healthy drinking water. The unavailability of healthy drinking water not only leads to health problems but also the well-being of people. According to Rochmi (2016), Indonesia is among the countries in South-East Asia that have a worse achievement in providing healthy drinking water for the people. Therefore, local governments in Indonesia are empowered to build a local water supply-owned company that aims to fulfill the need for dwellers on healthy drinking water. In this study, the indicator of healthy drinking water access is measured by the number of households with healthy drinking water access that plays as the numerator and the number of households in a local government as the denominator.

The third indicator of service performance is the immunisation rate for children under five years old. In Indonesia, local community health centres (LCHC) and local hospitals provide free immunisation services for children under five years old. Local community health centres have essential roles in the health development program since they are the first level of public health service facilities that provide health services to the residents. The LCHCs are in sub-districts of local government and supported by general practitioners, dentists, nurses, midwiferies, pharmacists, nutritionists, and other medical personnel. In this study, the immunisation rate for children is measured by the ratio of the number of children under five years old who have got immunisation divided by the total number of five years old children.

The fourth indicator of service performance is the birth rate attended by medical assistance. The Indonesian government highly concern about the maternal and neonatal rates since every day two mothers and eight newborns die (Utami 2018). Traditional birth attendants or unskilled health workers who help the birth process are parts of the factors enhancing the high rate of maternal and neonatal in Indonesia. Accordingly, the government has a commitment to enhancing the number of midwives who could work in rural or villages. In this study, the ratio of the birth rate attended by medical assistance is assessed by the number of childbirths assisted by

doctors or midwives divided by the total of childbirths in the geographical area of local government which they are responsible for in certain period.

The fifth indicator of service performance is the number of classrooms at the public primary and junior high schools in good condition. According to Law 20/2003, every citizen aged seven to fifteen is obliged to attend the primary and junior high schools. Accordingly, the local government has responsibilities to ensure the implementation of this policy and the residents have no pay to attend this basic formal education. All data of service performance indicators used in this study are collected from the Ministry of Finance and the Ministry of Education.

4.3 Environmental Performance

Protecting the biosphere is becoming more important not only for private companies but also for public institutions since it could influence the longevity of entities and the prosperity of future generations. Sustainable development constitutes a process of synergetic integration and co-evolution of economic, social, physical, and environmental factors that aim to maintain the prosperity of the people and to protect the surrounding biosphere (Camagni 2017). Scholars embraced quite diverse indicators to measure the environmental performance of local government (Recatalá and Sacristán 2014; Tran 2016; Alpenberg, Wnuk-Pel, and Henebäck 2018). The selection of environmental indicators tends to be a subjective decision (Fricker 1998; Bell and Morse 2012; Dizdaroglu 2015). It could depend on the scale of studies, availability of data and resources, the feasibility of measurement, and time constraints (Agol, Latawiec, and Strassburg 2014). In the same vein, data availability, target, and a conceptual framework should be considered in examining the environmental performance of local government (Verma and Raghubanshi 2018).

For instance, applying the principal component analysis (PCA) method, Recatalá and Sacristán (2014) recommended using the eighteen indicators of environmental resources such as the quality of air, water, soil, and biodiversity, to assess the environmental quality of the Valencian region and other parts of Europe. In line with this, Tran (2016) pointed out that applying a comprehensive approach that integrates science, policy-making, and stakeholders' involvement could lead to a better result of environmental assessment. Other studies developed environmental performance indices that did not only focus on the indicators highlighted in the previous paragraphs, but also on the waste management, gas emission, forestry, greenhouse gases, public transport (Jones 2006; Shen et al. 2011; Wendling et al. 2018; Alpenberg, Wnuk-Pel, and Henebäck 2018), compliance with environmental regulations, budget associated with the environmental aspects (Jasch 2000), natural disaster (King 2016; Panya et al. 2018).

This study develops a composite index of environmental performance generated from four indicators of environmental performance as set out in the Government Regulation 20/2006, the Government Regulation 76/2008 (revised through the enactment of the Government Regulation 26/2020), and the Law 26/2007. The first indicator of environmental performance is forest rehabilitation. The Government Regulation 26/2020 defines a forest as a large area of land determined by the central government that contains biological natural resources dominated by trees that cannot be separated from one another. Based on the data collected from the Ministry of Home Affairs, 125 districts used for this research have the forest. It is measured

by the total area of forest in the geographical area of local government that had been rehabilitated divided by the total area of forest damage. Forest rehabilitation programs have crucial roles in restoring trees cover on formerly forested land (de Jong 2010) and returning the forest to productive and sustainable condition through planting, seeding, agroforestry, and assisted natural regeneration (Chokkalingam 2006; de Jong 2010). In Indonesia, according to the Government Regulation 76/2008 as revised by the Government Regulation 26/2020, the head of local government and the concession holders have responsibilities to conduct the forest rehabilitation programs.

The second indicator of environmental performance applied in this study is forest areas in good condition. With the third-largest amount of rainforest in the world, Indonesia's forests contribute to the social and economic development of districts, especially after the implementation of decentralisation in 2001 (Hidayati, Vidyattama, and Gordon 2016). The districts receive local revenues from the forestry licences and forestry royalty (Tripathi 2015; Nasution 2017). However, it was found that there were 1.08 million ha of forest degradation in Indonesia every year due to illegal logging, encroached forest land, forest fire, and unplanned agricultural expansion (Kusmana 2011; Nawir and Rumboko 2007). Therefore, the districts and the local people are enforced by Law 41/1999 to build sustainable forest management that ensures the good condition of the forest. This study makes a comparison between the total area of forest in good condition as a numerator and the total area of forest as the denominator.

The third indicator of environmental performance used in this study is irrigation. Irrigation plays an essential role for agricultural crops, especially rice as the staple

food for most Indonesians (Booth 1977) since 71% of rice farmland in district areas is entirely dependent on irrigation (Vermillion et al. 2000). However, 30.4% of irrigation infrastructure managed by the districts was in poor condition (SDA 2018). Accordingly, the districts require the involvement of the farmers in the irrigation management system to overcome this problem (Sukandar, Masimin, and Fatimah 2018). According to the Government Regulation 20/2006, the head of the district has the authority to form an irrigation commission that has responsibilities to manage and coordinate the irrigation system (SDA 2019). The members of the irrigation commission are the district officials and farmer representatives (Prasetijo 2012). The indicator related to irrigation management is measured by the total area of irrigation in good condition that plays as the numerator and the total area of irrigation as its denominator.

The fourth indicator of environmental performance used in this study is the public green open space. According to Law 26/2007 in relation to spatial planning, public green open space is an open green space owned and managed by the local government for the benefit of the community. It consists of parks and trees along the road, rivers, and beaches. Each local government in Indonesia is encouraged to provide at least twenty percent of the public green open space of their geographical area. The indicator related to public green open space is measured by the total area of public open green space that plays as the numerator, and the total area of land and building that have certificates as its denominator.

4.4 Financial Performance

Scholars have lack of agreement on what and how to measure the financial performance of local governments (Bahl 1984; Chaney, Mead, and Schermann 2002; Honadle, Costa, and Cigler 2004; Kloha, Weissert, and Kleine 2005b; Kamnikar, Kamnikar, and Deal 2006; Wang, Dennis, and Tu 2007; Cabaleiro, Buch, and Vaamonde 2013). They applied various terms and nuances when highlighting and assessing the financial performance of local governments. It can be used as a dependent or an independent variable (Wang, Dennis, and Tu 2007). On the one hand, it becomes a dependent variable when the study aims to measure the variables that affect the financial performance of local governments (Honadle 2003). On the other hand, it serves as an independent variable of a framework when the research investigates the impact of financial performance on dependent variables such as organisational achievements.

Most of the financial performance studies of local government are conducted in industrialised countries, such as in the United States (Kloha, Weissert, and Kleine 2005a), Europe (Lotz and Bahl 1981; Bailey 1991; Zafra-Gómez, López-Hernández, and Hernández-Bastida 2009b), Japan (Takahashi 1999), and Australia (Dollery, Crase, and Byrnes 2006). However, Ritonga, Clark, and Wickremasinghe (2012) and Ritonga (2014) formulated the model of the financial performance of Indonesian local governments, as one of developing countries. They measured the financial performance of local governments based on six dimensions and eighteen reliable indicators.

This study uses a composite index of financial performance adopted from the measurement of financial performance developed by Ritonga, Clark, and Wickremasinghe (2012) and Ritonga (2014) because this study applies the Indonesian situation as well. This study uses four dimensions and fourteen indicators of financial performance, as shown in Table 4.4 below. The first dimension of financial performance is short-term solvency. Short-term solvency is the ability of the local government to pay its obligation within 12 months (Wang, Dennis, and Tu 2007). The short-term solvency of districts is measured by comparing the current assets as a numerator and current liabilities as the denominator (Ritonga, Clark, and Wickremasinghe 2012). The districts have good short-term solvency if the ratio of the indicators is high since it indicates that the districts have a better position to pay their current liabilities.

Dimensions		Measures
Short-term solvency	ST1	(cash and cash equivalent + short term investment)/current liabilities
	ST2	(cash and cash equivalent + short-term investment + account receivables)/current liabilities
	ST3	current assets/current liabilities
Long-term solvency	LT1	long-term liabilities/total assets
	LT2	long-term liabilities/investment equities
	LT3	investment equities /total assets
	LT4	long term liabilities /population
Budgetary solvency	BS1	(total revenues – special allocation fund revenue) / (total
		expenditures – capital expenditure)
	BS2	(total revenues – special allocation fund revenue) / operational expenditure
	BS3	(total revenues – special allocation fund revenue) / employee expenditure
	BS4	total revenue / total expenditure
Service-level	SL1	total equities / population
solvency		
	SL2	total assets / population
	SL3	total expenditures / population

Table 4.4: Dimensions of the Financial Performance of Local Governments

Adopted from Ritonga (2014)

The second dimension of financial performance is long-term solvency. Local governments have long-term solvency reflected by the lower value of its indicators that compare between long-term liabilities as a numerator and total assets or investment equities as the denominator (Ritonga, Clark, and Wickremasinghe 2012). The investment equities constitute the sum of long-term investments, fixed assets, and other assets. Local governments with sufficient assets to pay their long-term liabilities indicate that they have long-term solvency. In contrast, local governments have a low level of long-term solvency if their assets or investment equities are not enough to pay their long-term liabilities.

The third dimension of financial performance is budgetary solvency. It measures the budget ability of local government by comparing the revenue of the local government and their expenditures. The higher the value of indicators, the more solvent the budget of local government. A special allocation fund or earmarked is deducted from the total revenue since local governments do not receive it regularly (Ritonga 2014a). In Indonesia context, the central government sets and controls the special allocation fund based on three criteria, namely general¹, special², and technical criterion³ as regulated by Law 33/2004 and Government Regulation 55/2005.

The fourth dimension of financial performance used in this study is service-level solvency. It measures the ability of the local government to provide community services and facilities for residents (Wang, Dennis, and Tu 2007). The service level solvency of districts is measured by comparing the total equities, total asset, and

¹ General criterion is formulated based on financial capacity of local government reflected on their own revenues deducted by local employee expenditures.

² Special criterion is set based on the laws that regulate special autonomy and special characteristics for certain local governments such as underdeveloped regions, located in border zone with other countries, disaster prone areas, potential regions of tourism.

³ Technical criterion is based on the infrastructure condition of local government.

total expenditures as a numerator and population as the denominator (Ritonga 2014a). The districts have capabilities to provide more public services if the ratio between the numerator and the denominator is high.

4.5 Financial Flexibility

Financial flexibility is related to an ability of an entity to raise financial resources and to seize investment opportunities more effectively and timely when responding any unexpected events and dynamic changes in the work conditions such as environment, technology, market, and structure (Smith 2005; Bancel and Mittoo 2011b; Zeng, Zhang, and Wei 2013; Nouri and Jafari 2016). Soenen (2003) argued that financial flexibility or financial slack of entities is characterised by sufficient cash balance. Firms that hold high reserves have more capabilities to fund their new profitable projects since they do not rely on external funds. Similarly, firms with low leverage ratios have better investment opportunities to leverage their performance (Grüner and Raastad 2018). If firms have a lower level of debt, they tend to be easier to borrow money when they need extra cash.

Some studies measure financial flexibility based on several dimensions, such as debt financing flexibility (Marchica and Mura 2010), holding high cash balance (Soenen 2003; Almeida and Campello 2007), and a combination of cash balance and low debt level (Meier, Bozec, and Laurin 2013). Bancel and Mitto (2011) developed a financial flexibility index of a firm based on firms' leverage, liquidity, and operating ratio. In public sectors, Ritonga (2014) developed the measures of financial flexibility of local government in Indonesia adopted from the Canadian Institute of Chartered Accountants (CICA) (1997). He measured financial flexibility

based on the revenue capacity and the number of obligations to other parties. The revenue capacity as a numerator should be deducted by mandatory expenses and restricted revenues, such as employee expenditures and special allocation funds, or earmarked before divided by the amount of debt of local government (liabilities, principal loan, and interest expenditures) during the financial period.

The indicators of financial flexibility used in this study are adopted from Ritonga (2014), namely indicator A = (total revenues – special allocation fund revenue – employee expenditures)/ total liabilities and indicator B = (total revenues –special allocation fund revenue) / total liabilities. A logarithmic transformation is conducted to improve the normality of the data. The composite index of financial flexibility is aggregated and constructed from these two sets of indicators (transformed) with equal weight. Based on these measurements, the higher the value of indicators, the more flexible local governments are to pay their debts or liabilities.

4.6 Financial Independence

Institutions have financial independence if they can manage their financial resources independently without influenced by external parties (Petkovska 2011). The level of financial independence also can be judged by the incorporation of the independence of entities to raise revenues and how to spend those revenues (Chapman 1999; Oulasvirta and Turała 2007; Sitek 2014; Kotarba and Kolomycew 2014). The local governments have good financial independence if they have more own revenues since they do not depend on central government or other sources transfer to fulfil their expenditures. The indicators of financial independence used in this study are adopted from the Ritonga (2014), namely indicator A = total own revenues / total

revenues and Indicator B = total own revenues / total expenditures. This indicates that the higher the value of financial independence indicators, the more independence of local governments. A square root transformation is conducted to improve the normality of the data. Then, the composite index of financial independence is aggregated and constructed from these two sets of indicators (transformed) with equal weight.

4.7 Resource Management

A government budget constitutes an estimation of the present 12-month period of revenues and expenditures to support the economic management (Kiringai 2002; Rubin 2007; Liou 2011). Besides, the budget has goals to improve the operational efficiency of the government; to bestow legitimacy on the government; and to underpin a public contract between the government and the stakeholders such as the taxpayers, business interests, political groups, and donors (Simson and Welham 2014; Helpap 2015). Therefore, it is crucial for the government to have accurate financial resource management so that the budgets are distributed properly for the welfare of the people.

In the Indonesian context, according to Government Regulation 12/2019 dealing with the financial management of local government, the local governments have the autonomy to conduct financial resources management. This regulation aims to accelerate and enhance the effectiveness of public service provision, especially in relation to critical public services, such as education, health, and transportation. Also, the autonomy will enhance the capabilities of public entities to generate revenues and allocate capital and services spending in the most efficient manner

possible. Therefore, the local governments are enforced to reach the targets of the expenditures and revenues, as stated in the local government budget.

Achieving the targets of expenditures is critical since the low budget absorption might lead to the failure of the local government (Gunarto, Sentri, and Said 2018; Okoye et al. 2019). For example, if the local government fails to absorb the regional transfer funds allocated for road construction, it made local products lose their competitiveness compared to other regions. Furthermore, the inability of local government to achieve the target of budget realisation could lead to idle cash and has an impact on the loss of benefits of spending since the funds are not utilised effectively and timely (Prasetyo 2015). In the same vein, the low achievement of the revenues target has an impact on short-term solvency and the ability of the local government to fulfil the needs of people on public services.

This study uses the secondary data of the audited financial statement of local governments to measure the financial resources management of local governments. There are two indicators used in this study to measure the financial resources management of the local government: (1) the ratio of the actual aggregate expenditures/the approved aggregate budgeted expenditures; (2) the ratio of the actual aggregate revenues/the approved aggregate budgeted budgeted revenues (Government Regulation 17/2017). The aggregate revenues and expenditures are measured over a 12-month period.

Then, the results of these two indicators are transformed using the percentile approach since it gives the meaning or interpretation of the scores and allows the comparison of the scores (Stockburger 2001), which in turn, the reader are relatively

easy to understand the meaning (Robinson 2013; Bowman 2002). This study uses SPPS 24 to generate the percentile method. The results of the percentile method are as follows. The ratios of aggregate revenues (X) are categorised into 1 (low) = X<93.67, 2 (moderate) = 93.67 < X < 96.75, 3 (high) = 96.75 < X < 99.07, and 4 (very high) = $X \ge 99.07$. While the ratios of aggregate expenditures (Y) are categorised into 1 (low) = 1 (low) = X < 87.74, 2 (moderate) = 87.74 < X < 90.6, 3 (high) = 90.6 < X < 93.55, and 4 (very high) = $X \ge 93.55$. The aggregate value of financial resource management of local government is measured by Σ ratio revenues and expenditures/8.

4.8 Financial Transparency

Researchers have developed different measures of financial transparency of local governments, and most of them used online data published on the official websites of the government (Laswad, Fisher, and Oyelere 2005; Alt and Lassen 2006; Styles and Tennyson 2007; Guillamón, Bastida, and Benito 2011; Caamano-Alegre et al. 2013; Del Sol 2013; Bernick et al. 2014; Garcia-Tabuyo, Sáez-Martín, and Caba-Pérez 2016; Lowatcharin and Menifield 2015; Araujo and Tejedo-Romero 2018; 2018). However, to the author's knowledge, literature focusing on financial transparency measurements of subnational governments in developing countries is rare. Previous research focused on the financial transparency of local governments in developed countries such as the United States, New Zealand, and Spain.

For instance, Styles and Tennyson (2007) and Bernick et al. (2014) applied the Comprehensive Annual Financial Report (CAFR) published on the official website as the indicator of financial transparency of municipalities in the United States. They posited that the availability of the CAFR on the official website indicated the financial transparency of the entities. Laswad, Fisher, and Oyelere (2005) measured the financial transparency of local government in New Zealand based on four indicators of financial information, namely financial highlight, annual reports, annual plan, and combination of the annual reports, plans, and financial highlights. They calculated the value of financial transparency based on the criteria whether the local authority provides those financial reports or not on the official website.

Another approach of financial transparency measurement was developed by Lowatcharin and Menifield (2015). They utilised the transparency checklist that consists of 10 items. The 10-point transparency items are budget, open meeting laws, elected officials, administrative officials, building permits and zoning, audits, contracts, lobbying, public records, and taxes. If each indicator is available on the official website of subnational government, it will be scored 1 point. If the data is not available, it will be scored 0. The points were then aggregated.

In Spain, Guillamón, Bastida, and Benito (2011) measured the financial transparency of municipalities by using twenty parameters. They scored the index of financial transparency based on the availability of the financial information on the official website of the municipalities. Gesuele, Metallo, and Longobardi (2018) used thirteen parameters to investigate the transparency of municipalities, related to organisation, staff, wages, organisational structure, internal auditing, performance evaluation, financial statement, economic planning, support planning, and assets. Interestingly, Gesuele, Metallo, and Longobardi (2018) used not only the official website of the municipalities but also the social media, such as Facebook and Twitter to indicate the availability of those thirteen parameters.

In the Indonesian context, considering the mandate of Law No.14/2008 and the paramount roles of the internet, the central government instructed all local governments to publish their financial information on the official websites. However, the central government did not provide the measurements of financial transparency. Therefore, this study adopts the financial transparency measures developed by Guillamón, Bastida and Benito (2011) since they provided comprehensive measures of financial transparency, as shown in Table 4.5 below. This study uses two dimensions, namely the availability and the accessibility of financial information published on the websites (Adriana and Ritonga 2018). The availability means that the financial documents of fiscal year 2016 are available on the local government's websites, while the accessibility shows that the financial documents could be downloaded by the people.

Table 4.5: Indicators of Financial	Transparency
------------------------------------	--------------

No	Indicators
	Publication of:
1.	Annual financial report
2.	The annual financial report approval date
3.	The Local Government budget
4.	The budget approval date
5.	Budget changes approved
6.	The budget of the decentralised entities and municipal corporations
7.	Regularly publication of information on budget execution (minimum monthly)
8.	Audit reports of Local government
9.	Audit reports Local owned companies and other municipal agencies
10.	Surplus (deficit) per capita
11.	Fiscal autonomy (tax revenue / total revenues)
12.	Tax revenue per capita
13.	Expenditure per capita (expenditures/population)
14.	Investment per capita (investment expenditure / population)
15.	Average payment period (outstanding liabilities x 365 / total expenditures)
16.	Average collection period (outstanding credits x 365 / total revenues)
17.	The municipal debt
18.	Data provided on the yearly evolution of debt
19.	Debt per capita (current financial liabilities/population)
20.	Relative debt (debt / total budget)
(Gui	Ilamón, Bastida, and Benito 2011)

In this study, the scoring process of each measure of financial transparency is presented as follows. If each item is available and accessible on the local government's websites, it will be given a score of 1. If the data is not available and inaccessible, it will be given a value of 0. The aggregate value of financial transparency of each subnational government is determined based on the equal weight of 0.5 for the indicator of availability and accessibility. Based on this equal weight, the total score of financial transparency of each dimension. Then, the total score is divided by the expected score (20 points) to obtain the transparency index of each local government.

4.9 Internal Control

Internal control is an indispensable tool since it improves the efficiency and effectiveness of a business, ensures standard compliance with policies and procedures, prevents business failure, and counters threats (IFAC 2012). Furthermore, applying the effectiveness of internal control is essential for maintaining the creditworthiness of financial reporting and hindering the tendency of deception (Mohrman 1996; Rittenberg and Schwieger 2005; Schneider and Church 2008; Al-Thuneibat, Al-Rehaily, and Basodan 2015).

The implementation of internal control system in the subnational levels requires more attention from all parties since local autonomy is considered to be the main pillar of public sector reforms in the last two decades in developing countries (Baltaci and Yilmaz 2006). For instance, in Indonesia, the central government has continuously empowered local government organisations by enhancing political, administrative,

and financial autonomies. The financial autonomy is indicated by the independence of sub-national authorities to raise revenues and to spend expenditures. These aim to improve the performance of regional governments, to alleviate the prosperities gap among the local governments, and to enhance the quality of public service provision.

Huefner (2011), Ng'eni and Chalam (2016) and Sutopo et al. (2017) argued that the autonomy policy tends to be successful if this policy is accompanied by self-governance system, such as the internal control system since it could minimise the possibility of risks, threats, and the violation of the authority of local officials. On the contrary, a slackening of internal control could imperil not only on public service quality but also on the sustainability of the environment. For instance, the uncontrollable licenses in mining and forestry business issued by local governments for the interest of private companies exacerbate the ecological population (Rahman et al. 2013; Siombo 2014). Accordingly, to assure the implementation of internal control in local governments in Indonesia, the central government enacted Government Regulation 60/2008 and Government Regulation 4/2016 in relation to Government Internal Control System that instructs all government entities, including local governments, to build their internal control system.

This study applies a composite index of internal control developed from five indicators, as recommended by the Finance and Development Supervisory Agency of Indonesia and the Committee of Sponsoring Organisation of the Treadway Commission (COSO). The COSO identified the five indicators of internal control (Jokipii 2010), as shown in Table 4.6 below. The first indicator is control environment. Control environment constitutes the foundation of internal control

system that incorporates a set of standards, ethical values, legal framework, organisational structure, and competency of human capital (Smith et al. 2005; O'Leary, Iselin, and Sharma 2006; Sudsomboon and Ussahawanitchakit 2009; Leung, Cooper, and Perera 2011; Karagiorgos, Drogalas, and Giovanis 2011; Eldridge et al. 2014). It could reduce the level of fraudulent practices in the organisation (Wood and Brathwaite 2013). Also, control environment empowers all employees to ensure that organisational resources are optimised effectively and efficiently for the purposes of goals achievement (Aldridge and Colbert 1994).

No	Component Subcomponents				
1.	control	enforcement of integrity and ethical value, commitment to			
	environment	competence, conducive leadership, organisational structure,			
		the delegation of authority and responsibility, preparation and			
		implementation of healthy policies on HR development, the			
		effectiveness of the role of internal control apparatus, and			
		working relationships with other government agencies.			
2	risk assessment	risk identification and risk analysis			
3.	control activities	performance review, HR development, information system control, asset control, performance indicators determination and review, function separations, transaction authority, accurate and timely reporting, access limitations on resources and reporting, accountability of reporting and resources, documentation of internal control, and other important transactions.			
4.	information and communication	information relevance, and communication effectiveness			
5.	monitoring activities	monitoring sustainability, and separated evaluation			

Table 4.6: The Measurements of Internal Control

(The Finance and Development Supervisory Agency, 2016)

The second indicator of internal control is risk assessment. It is dynamic and iterative assessment processes that identify and control the possibility of internal and external threats such as changes of regulations, natural disasters, and security issues. These threats should be mitigated in appropriate circumstances since any exceptional or undesired events could have a negative impact on the financial, strategic, and operational losses (Acharyya 2010; Dionne 2013; Halonen 2014). The third indicator of internal control is control activities that refer to manual or automated activities that ensure the policy and procedural guidelines are performed by all employees (Rezaee, Elam, and Sharbatoghlie 2001; Aikins 2011; COSO 2013). These control activities comprise verification, approval activities, and performance reviews. Furthermore, segregation of duties and functions, rewards, and punishments are also control activities that aim to ensure the objectives of entities are performed within the framework of business process (Kohn 1999). Mahadeen et al. (2016) categorised the control activities into four groups, namely directive, detective, preventive, and corrective function. Directive function refers to regulations, guideline, procedure, and policies that guide the entity to achieve their objectives. Detective function is related to the development of alert management that detects the unexpected events. Preventive activities are the process that could deter the predicting problems might occur, while corrective activities refer to the processes that improve the current performance affected by the unexpected conditions and develop the procedures to avoid the recurrence of the problems in the future.

The fourth indicator of internal control is information and communication tool. It is developed by an organisation that aims to capture and disseminate the relevant information in a timely way (Aldridge and Colbert 1994; Mahadeen et al. 2016). Sharma and Patterson (1999) argued that the transmission of information could be performed through formal and informal contact since the most important thing is that personnel and managerial levels could capture and share the information to carry out their responsibilities effectively (Jokipii 2010). Utilising modern technology such as computer-based information, the internet, and mobile communication in control

system of organisation could integrate hardware, software, and human capacities to generate the decision making with a high degree of accuracy and in a timely manner (Gamayuni 2019). The fifth indicator of internal control is monitoring activities. This indicator is related to the process of evaluation that measures whether the internal control system is present and operating as intended. It also involves in taking appropriate changes that aim to alleviate the deficiencies. The implementation of monitoring activities can be either as periodic evaluation or real-time evaluation (Benedek, Szenténé, and Beres 2014). The monitoring activities also involve in taking changes and carry out innovation in order to maintain the effectiveness of system's performance (Yemer and Chekol 2017).

4.10 Accountability

Accountability constitutes a power relationship between an accountor (agent/steward/subordinate) and an accountee (principal/superior) (Mayston 1993; Mulgan 2000; Coy, Fischer, and Gordon 2001; Kluvers 2003; Broadbent and Laughlin 2003). It delineates the obligation of the accountor to supply the information on the action, process, output, or outcome to the accountee or higher authorities (Laughlin 1990; Patton 1992; Kearns 1996). The accountee has the power to evaluate and to render judgment, feedback, and sanction (Quinn and Schlenker 2002; Steccolini 2004).

Accountability has different forms and characteristics, such as professional accountability, legal accountability, social accountability, and administrative accountability (Wang 2002; Bovens 2007a; Poulsen 2009; Christensen and Lægreid 2015). This study uses the secondary data of the local government accountability

reports collected from the Ministry of Administrative and Bureaucratic Reform. Based on Presidential Regulation 29/2014 and Regulation of the Minister of Administrative and Bureaucratic Reform 25/2012 and 12/2015, accountability reports encapsulate the evaluation of public service standards and ethics codes carried out by the local government (Afriyanti, Sabanu, and Noor 2015). As well, the local government accountability reports show the achievement of targets required by the public service standards. The government institutions, including local governments, have responsibilities to periodically submit the local government accountability reports to the Ministry of Administrative and Bureaucratic Reform. The Ministry of Administrative and Bureaucratic Reform evaluates the reporting documents and assesses the quality of the administrative accountability of local authorities based on the five components, as described in Table 4.7 below.

Table 4.7: Administrative Accountability of Local Government in Indonesia

1.	Program Planning	Strategic Planning: the availability and the quality of strategic planning documents meet required standards; strategic planning documents are used as a guideline for program and budget planning; Yearly Program Planning: the availability and the quality of yearly programs planning meet required standards; program targets are already applied and monitored
2.	Program	the availability and the quality of key performance indicators
	Measurement	meets required standards; key performance indicators are
	Achievement	already applied and review periodically
3.	Accountability	the availability of accountability reporting meet the required
	Reporting	standard; the comprehensiveness of accountability reporting;
		and the usefulness of information on the accountability
		reporting for future performance improvement
4.	Internal	the availability and the implementation of internal evaluation
	Evaluation	guideline; and the utilisation of evaluation
5.	Target	reports on the achievement of targets; and the improvement of
	Achievement	achievement compare to the previous year.
(Th	Alinietry of Admini	strative and Burgaucratic Poform 2015)

(The Ministry of Administrative and Bureaucratic Reform, 2015)

5. Conclusion

This chapter describes the research methods, including data collection, data sources, and research samples. Definitions and measurements of public sector governance (financial transparency, accountability, and internal control), financial management factors (financial flexibility, financial independence, and resource management), financial performance, service performance, and environmental performance are explained in detail as well. This study applies multiple regression analysis to examine the effects of public sector governance and financial management factors on the service and environmental performance of local governments. Bootstrapping method is used to investigate the mediating effects of financial performance on service and environmental performance. The next chapter presents the findings regarding the effects of public sector governance and financial management factors on service and environmental performance in district governments in Indonesia and investigates the role of financial performance on service and environmental performance in district governmental performance.

Chapter 5 Service Performance of Local Governments: Data Analysis and Hypotheses Testing

1. Introduction

The main objective of this chapter is to report data analysis and examine the hypotheses discussed in Chapter 2 (see Figure 5.1). Based on the results of hypotheses testing, this chapter will answer the following research questions:

- 1. What are the influences of public sector governance and financial management factors on service performance?
- 2. Are the relationships between public sector governance, financial management factors, and service performance mediated by financial performance?

Detailed linkage of those research questions with the research objectives are shown in Figure 5.2.



Figure 5.1: Structure of Chapter 5





2. Data Preparation and Analysis

As discussed in Chapter 3, this study focuses on the district level. The number of samples used in this study is 125 local governments after removing the outliers. This study conducts data preparation and data analysis, such as validity and reliability tests, to examine whether the data are suitable for further analysis. This study investigates Kaiser-Meyer-Olkin (KMO) Measure of Sampling Adequacy (MSA) and Bartlett's test of sphericity to measure the validity test (Williams, Onsman, and Brown 2010; Shirali, Shekari, and Angali 2018). The validity test can be measured by Pearson product-moment correlation analyses as well (Piland et al. 2003; Ghozali 2016; Swank and Mullen 2017). The Kaiser-Meyer-Olkin Measure of MSA is a single-value used to investigate the sampling adequacy of the inter-correlations of a set of variables for factor analysis (Field 2013).

This study uses Cronbach's alpha to perform the reliability test. The Cronbach's alpha is related to the internal consistency that shows how closely related a set of scale or test items (UCLA 2006). It is the most widely used measure of the internal consistency of the data (Bonett and Wright 2015; Taber 2018). The detailed results validity and reliability test of the variables used in this study are described in the following paragraphs.

The validity and reliability results of service performance show that the total variance explained by the factor is 50.08%. The Bartlett's test of sphericity is significant (p =.000), and the overall test of KMO-MSA is .706. The indicators of service performance have good internal consistency to measure the same construct because the Cronbach alpha of the variable of service performance is 0.711. The

validity test of financial performance is measured by applying factor analysis that resulted in deleting LT3. LT3 is long-term solvency, one of the indicators of financial performance (see Table 3.7 of Chapter 3). The measure of LT3 is deleted since the KMO-MSA value of LT3 is less than .50. The total variance explained by the factor is 93.06%. The Bartlett's test of sphericity is significant (p =.000), and the overall test of KMO-MSA is .710. The standardised Cronbach coefficient alpha of financial performance is 0.703. The reliability test of this variable is measured by the standardised Cronbach coefficient alpha of financial performance have multi-point scales (Santos 1999; Ritonga 2014b). Based on the coefficient, the variable of financial performance shows the satisfactory internal reliability of the scale. Likewise, the construct validity of the variables of internal control, financial transparency, financial flexibility, financial independence, and resource management are found significant (p =.000), and these variables have good internal consistency to measure the same construct indicated by the Cronbach alpha (respectively, .691, .711, .916, 915, and .702).

3. Descriptive Statistics and Correlations

Many statistical methods have been applied to test the normality of data (Kim 2013). However, there is no agreement on which values indicate normality or non-normality (Orcan 2020). One method to check any deviation from normality is using skewness and kurtosis (Orcan 2020). In the previous accounting research (e.g., Ritonga 2014, Afris 2017, Mahbub 2019), a rule thumb of the acceptable values of skewness and kurtosis fall within a range of ± 2 (Hair, Black, and Babin 2010; George and Mallery 2010; Gravetter and Wallnau 2017). Table 5.1 shows that the skewness and kurtosis of service performance are 0.673 and 0.541, respectively. Other variables of the study are within the acceptable limit as well (between -2 and +2)⁴. Therefore, these results are suitable for further analysis.

		Descriptive statistics						Correlations								
	Variables	Mean	Med.	SD	Min.	Max.	Sk.	Ku.	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
	1. Service Performance	.467	.453	.090	.291	.758	.673	.541	1							
Sector	2. Financial Performance	.329	.323	.081	.159	.566	.485	.150	.479**	1						
	3. Financial Transparency	.192	.204	.112	.000	.425	003	-1.025	.395**	.462**	1					
olic S verr	4. Accountability	.546	.546	.073	.316	.702	387	.529	.362**	.446**	.401**	1				
n of L	5. Internal Control	.508	.525	.186	.135	.854	079	923	.409**	.465**	.509**	.399**	1			
l ent	6. Financial Flexibility	.495	.488	.231	.000	1.000	.040	731	.373**	.447**	.360**	.370 ^{**}	.371**	1		
gen	7. Financial Independence	.463	.474	.182	.001	.952	.006	045	.343**	.474**	.304**	.357**	.388**	.310**	1	
lana	8. Resource Management	2.56	2.500	.928	1.000	4.000	024	991	.364**	.472**	.292**	.295**	.285**	.442**	.196 [*]	1

Table 5.1: Descriptive Statistics and Correlations

N= 125; Med. = Median; SD = Standard Deviation; Min. = Minimum; Max. = Maximum; Sk. = Skewness; Ku. = Kurtosis * p < 0.05; ** p < 0.01

Table 5.1 above shows Pearson's correlation matrix of variables. As the table suggests, there are no high correlations among the variables. The correlation analysis is conducted to measure the degree of a linear relationship between variables. The coefficient correlation denoted by r can range between +1 and -1, while values of r close to 0 represent no linear correlation between the variables (Stine and Foster 2011). The correlation coefficient of +1 means that the relationship between two variables is a perfect positive correlation, while the coefficient of -1 indicates the perfect negative correlations represent that one variable is precisely predictable from another variable (George and Mallery 2010). As a rule of thumb, a bivariate correlation of .80 or higher may lead to a problem (Gujarati and Porter 2009; Whitley and Kite 2013).

⁴ Mean and standard deviation of the study variables are also computed. They are .467 and .09 for service performance, .329 and .081 for financial performance, .192 and .112 for financial transparency, .546 and .073 for accountability, .508 and .186 for internal control, .495 and .231 for financial flexibility, .463 and .182 for financial independence, .256 and .928 for resource management respectively.

In this study, as discussed in the previous subsections, public sector governance is represented by financial transparency, accountability, and internal control, while financial management factors are examined by financial flexibility, financial independence, and resource management. It is found that there is a positive and significant correlation between financial transparency, accountability, internal control, and service performance (respectively, r = .395, p < 0.01; r = .362, p < 0.01; r = .409, p < 0.01). These results show that the higher financial transparency, accountability, and internal control, the better the service performance of local government. Similarly, the results indicate that there is a positive and significant correlation between financial transparency, and significant correlation performance (respectively, r = .462, p < 0.01; r = .446, p < 0.01; r = .465, p < 0.01).

In a similar vein, it is found that that there is a positive and significant correlation between financial flexibility, financial independence, resource management, and service performance (respectively, r = .373, p < 0.01; r = .343, p < 0.01; r = .364, p < 0.01). Table 5.1 also shows that there is a positive and significant correlation between financial flexibility, financial independence, resource management, and financial performance (respectively, r = .447, p < 0.01; r = .474, p < 0.01; r = .472, p < 0.01). These findings show that if financial flexibility, financial independence, and resource management exist, service and financial performance are likely to be achieved. As well, Table 5.1 above presents that financial performance is found to be significantly correlated with service performance (r = .479, p < 0.01). This shows that if financial performance is strengthened, then service performance will be enhanced. This result indicates a probability of financial performance as a mediating variable.

4. The Impacts of Public Sector Governance

Multiple regression analysis is conducted to examine the level of prediction of public sector governance (financial transparency, accountability, and internal control) on service and financial performance while controlling for population size (measured by taking the natural logarithm), poverty rate, and gender diversity. Population size, poverty rate, and gender diversity are included in the analysis as control variables since prior literature suggested that population size, poverty rate, gender diversity would likely influence the performance of entities (Wang, Dennis, and Tu 2007; Casal and Gómez 2011; Lewis 2017b; Baker, Ali, and French 2019; Dani, Picolo, and Klann 2019).

This study uses the Centered Leverage and Cook's Distance to analyse a multivariate outlier. The multivariate outlier is a combination of unusual values on two or more variables (Parke 2013). The acceptable values of the Centered Leverage and Cook's Distance are less than one and close to zero (Hair, Black, and Babin 2010; Field 2013). The values for Centered Leverage and Cook's Distance of Equation 5.1 and 5.2 are less than one and close to zero. These results indicate that the outliers do not have any influences on Equations 5.1 and 5.2. The models (Equation 5.1 and 5.2) have no evidence of multicollinearity since the VIF values of both equations are considerably less than the generally accepted maximum level of 10. There is no autocorrelation as well since the Durbin Watson values of Equation 5.1 (d = 1.766) and Equation 5.2 (d = 1.879) are considered an acceptable level, in a range between 1 and 3 (Field 2013). The Kolmogorov-Smirnov test indicates that the assumption of normality of residuals is fulfilled since the *p*-value is .200, which is higher than 0.05. As a rule of thumb, the residuals are normally distributed if the *p*-

value of Kolmogorov-Smirnov is higher than 0.05 (Drezner, Turel, and Zerom 2010; Ghozali 2016). Likewise, the assumption of linearity and homoscedasticity of Equation 5.1 and 5.2 are fulfilled because the scatterplot of the standardised residual graph of both models are scattered (see Appendix). The assumption of linearity and homoscedasticity are met if plots are scattered and patternless (Baiocchi 2001; Hair, Black, and Babin 2010; Garson 2012). Based on these analyses, both models are deemed to meet the assumptions of ordinary least squares (OLS) regression analysis.

Table 5.2: Multiple Regression Analysis of Public Sector Governance on Financial and Service Performance

	Servic	e Perforn	nance	Financial Performance			
	B SE		β	В	SE	β	
Public Sector Governan	ce:						
- Financial Transparency	.158	.078	.196*	.155	.056	.214**	
- Accountability	.226	.111	.183*	.261	.080.	.235**	
- Internal Control	.109	.047	.223*	.086	.034	.196*	
Control Variables:							
- Population	009	.009	081	032	.007	315**	
- Poverty Rate	066	.137	040	213	.100	143*	
- Gender Diversity	.139	.139	.082	.272	.101	.178**	
R ²	.261			.518			
Sig.		.000		.000			

Note: * p < 0.05; ** p < 0.01

SP= .311 +.196FT + .183AC + .223IC - .081Pop - .040Pov + .082Gdr (Equation 5.1)

FP = .407 + .214FT + .235AC + .196IC - .315Pop - .143Pov + .178Gdr (Equation 5.2) SP = service performance; FP = financial performance; FT = financial transparency; AC = accountability; IC = internal control; Pop = population size; Pov = poverty rate; Gdr = gender diversity

Table 5.2 above presents that financial transparency, accountability, internal control, and control variables (population, poverty rate, and gender diversity) explain 26.1% of the variance in service performance ($R^2 = .261$, F (6,118) = 6.932, *p* < .001). Likewise, the results of the regression indicate the six predictors explained 51.8% of the variance in financial performance ($R^2 = .518$, F (6,118) = 21.119, *p* < .001). This means that there is still another 73.9% and 48.2% variance in service performance

and financial performance that has not been explained by the models respectively. Based on the results of regression, it is found that financial transparency, accountability, and internal control significantly predict service performance (respectively, $\beta = .196$, p < 0.05; $\beta = .183$, p < 0.05; $\beta = .223$, p < 0.05) and financial performance (respectively, $\beta = .214$, p < 0.01; $\beta = .235$, p < 0.01; $\beta = .196$, p < 0.05). The positive sign means that an increase of one unit in financial transparency, accountability, and internal control results in an increase of .196, .183, and .223 in service performance respectively, assuming that other variables are held constant. Likewise, an increase of one unit in financial transparency, accountability, and internal control results in financial transparency, accountability, and internal control service performance, accountability, and internal control results in financial transparency, accountability, and internal control leads to an increase of .214, .235, and .196 in financial performance respectively, assuming that other variables are held fixed.

Based on the hypotheses testing, the study found empirical support for hypotheses H1, H2, and H3. The hypotheses are accepted. It can be concluded that that financial transparency, accountability, and internal control have a major role in improving financial and service performance of local governments. In this study, the local government's financial and service performance are conducted in a separated analysis since service and financial performance are essential and significant as indicated by R^2 and *p*-value, as discussed in the previous paragraphs.

5. The Impacts of Financial Management Factors

In Hypothesis 4, 5, and 6, financial management factors are represented by financial flexibility, financial independence, and resource management. The multiple regression analysis is performed to test these hypotheses. Control variables (population, poverty rate, and gender diversity) are included in regression models
(Equation 5.3 and 5.4). The values for Centered Leverage and Cook's Distance of Equation 5.3 and 5.4 are less than one and close to zero. These results indicate that the outliers do not have any influences on Equations 5.3 and 5.4. Both regression models (Equation 5.3 and 5.4) have no evidence of multicollinearity as the VIF values of the models range from 1.041 to 1.354. These values are considerably less than the generally accepted maximum level of 10. There is no autocorrelation as well since the Durbin Watson values of Equation 5.3 (d = 1.626) and Equation 5.4 (d = 2.064) are considered an acceptable level. Kolmogorov-Smirnov test indicates that the assumption of normality of residuals has been fulfilled since the *p*-value is .200, which is higher than 0.05. The assumption of linearity and homoscedasticity is fulfilled since the scatterplot of the standardised residual graph of both models are patternless (see Appendix). Thus, the regression models (Equation 5.3 and 5.4) meet the assumptions of OLS regression analysis.

Independent Variables	Ser	ice Perfor	mance	Financial Performance				
lindependent variables	В	SE	β	В	SE	β		
Financial Management	Factors	:						
- Financial Flexibility	.076	.036	.194*	.068	.109	.193**		
- Financial Independence	.112	.042	.225**	.140	.029	.314**		
- Resource Management	.020	.009	.210*	.021	.006	.245**		
Control Variables:								
- Population	10	.010	092	032	.007	316**		
- Poverty Rate	092	.136	055	222	.093	149*		
- Gender Diversity	.057	.138	.034	.196	.095	.129*		
R ²		.254		.566				
Sig.		.000		.000				

Table 5.3: Multiple Regression Analysis of Financial Management Factors on Service and Financial Performance

Note: * p < 0.05; ** p < 0.01

SP= .440 + .194FF + .225FI + .210RM - .092Pop - .055Pov + .034Gdr (Equation 5.3) FP = .513 + .193FF + .314FI + .245RM - .316Pop - .149Pov + .129Gdr (Equation 5.4) SP = service performance; FP = financial performance; FF = financial flexibility; FI = financial independence; RM = resource management; Pop = population size; Pov = poverty rate; Gdr = gender diversity

Table 5.3 shows that the results of the regression indicate financial flexibility, financial independence, resource management, and control variables (population, poverty rate, and gender diversity) explain 25,4% and 56.6% of the variance in service performance (R^2 = .254, F (6,118) = 6.680, p < .001) and financial performance (R^2 = .56.6, F (6,118) = 25.642, p < .001) respectively. There is still another 74.6% and 43.4% variance in service performance and financial performance that has not been explained by the models respectively. The findings show that financial flexibility, financial independence, and resource management significantly predict service performance (respectively, β = .194, p < 0.05; β = .225, p< 0.01; β = .210, p < 0.05) and financial performance (respectively, β = .193, p < 0.01; β = .314, p < 0.01; β = .245, p < 0.01). The positive sign means that an increase of one unit in financial flexibility, financial independence, and resource management leads to an increase of .194, .225, and .210 in service performance respectively, assuming that other variables are held constant. Likewise, an increase of one unit in financial flexibility, financial independence, and resource management leads to an increase of .193, .314, and .245 in financial performance respectively, assuming that other variables are held fixed

Thus, the regression analysis findings indicate that hypotheses H4, H5, and H6, which predict a positive and significant relationship between financial management factors and financial and service performance, are accepted. It can be concluded that the higher level of financial flexibility, financial independence, resource management, the better the financial and service performance of local governments.

6. The Mediating Role of Financial Performance on Local Government's Service Performance

This study employs the Bootstrapping method generated by PROCESS Macro version 3 to test Hypothesis 7, which posits the mediating effect of financial performance on service performance. Bootstrapping is a versatile method that uses computer-intensive resampling to derive a more accurate statistical inference analysis (Lockwood and MacKinnon 1998; Hayes 2018). There are three advantages of bootstrapping method (Wood 2005). First, bootstrapping is more transparent and more simple than other conventional methods. Then, bootstrapping is more easily checked and less restrictive. Finally, the bootstrapping method can be applied where other methods may be difficult to be implemented to analyse small and non-normal distribution data.

This study presents the indirect (mediation), total, and direct effects of predictors on the service performance based on the results of the bootstrapping analysis. Mediation refers to the condition when independent variables predict a dependent variable through a mediating variable called an indirect effect (Edwards and Lambert 2007; Meule 2019). An indirect effect is the combined effects of two relationships: 1) between an independent and a mediating variable, and 2) between the mediating variable and the dependent variable. In this study, the mediating effect is measured by the 95% bootstrapped confidence intervals with 5000 of the number of bootstrap samples. If the bootstrapped confidence intervals do not contain zero, it shows that there are the effects of a mediating variable on the relationship between predictors and the dependent variable. Thus, it rejects the null hypothesis. The total effect is the effect of an independent variable on a dependent variable when the mediating variable is not present in the model. If the mediator variable is included, this relationship is called the direct effect (MacKinnon 2011; Lancee et al. 2015; Agler and De Boeck 2017; Meule 2019).

In this study, the outliers do not have any influence on Equations 5.5 and 5.6 because the values for Centered Leverage and Cook's Distance of both equations are less than one and close to zero. Both regression models (Equation 5.5 and 5.6) have no evidence of multicollinearity as the VIF values of the models range between 1.079 and 2.304, which are less than the generally accepted maximum level of 10. There is no autocorrelation as well since the Durbin Watson values of Equation 5.5 (d = 1.694) and Equation 5.6 (d = 1.58) are considered an acceptable level. Kolmogorov-Smirnov test indicates that the assumption of normality of residuals has been fulfilled since the *p*-value is .200, higher than 0.05. Thus, Equations 5.5 and 5.6 meet the assumptions of ordinary least squares regression.

Table 5.4: Mediating Role of Financial Performance on Service Performance (B, 95% CI). Standardised Coefficients Results.

,					
Dependent Variable	Independent Variables	a path	Indirect effect (ab)	c path (total effect)	c' (direct effect)
Service Performance	Public Sector Governance	•			
	- Financial Transparency	.2145 (.0437, .2667)	.0613 (.0059, .1339)	.1962 (.0043, .3121)	.1349 (0464, .2639)
	- Accountability	.2355 (.1022, .4197)	.0673 (.0054 .1476)	.1828 (.0067, .4449)	.1155 (0809, .3661)
	- Internal Control	.1963 (.0179, .1538)	.0561 (.0011, .1253)	.2228 (.0148, .2024)	.1666 (0129, .1753)
	Total Model	R ² = .300, F (7, 117) =	7.165, p < .001		
	Financial Management Fa	ctors:			
	- Financial Flexibility	.1935 (.2971, .7281)	.0539 (.0017, .1369)	.1938 (.0041, .1476)	.1400 (0178, .1274)
	- Financial Independence	.3139 (.0830, .1963)	.0874 (.0161, .1847)	.2255 (.0290, .1946)	.1381 (0206, .1576)
	- Resource Management	.2448 (.0094, .0334)	.0681 (.0116, .1408)	.2098 (.0029, .0380)	.1417 (0043, .0319)
	Total Model	R ² = .287, F (7, 117) =	6.734, p < .001		

All paths are included control variables (population size, poverty rate, and gender diversity)

SP=.181+.135FT+.115AC+.167IC+.286FP+.009Pop+.001Pov+.031Gdr (Equation 5.5) SP=.281+.140FF+.138FI+.142RM+.278FP-.004Pop-.014Pov-.002Gdr (Equation 5.6) In this study, the mediating role of financial performance on service performance is presented in two parts. The first part investigates the mediation effects of financial performance on the relationships between public sector governance (financial transparency, accountability, and internal control) and service performance (Equation 5.5). The Equation 5.5 shows that the significant total effects are observed for financial transparency (β = .1962, 95%CI = .0043 to .3121), accountability (β = .1828, 95%CI = .0067 to .4449), and internal control (β = .2228, 95%CI = .0148 to .2024) on service performance after including population size measured by taking the natural logarithm, poverty rate, and gender diversity. The total effect of financial transparency, accountability, and internal control on service performance is significant since zero is not found in the lower level and upper level of 95% confidence intervals. Based on these findings, an increase in financial transparency, accountability, and internal control service performance after including control variables.

Table 5.4 shows direct effects of financial transparency, accountability, and internal control on service performance after including the mediator and control variables (respectively, β = .1349, CI = -.0464 to .2639; β = .1155, CI = -.0809 to .3661; β = .1666, CI = -.0129 to = .1753). These findings show that if a local government has a higher level of financial transparency, accountability, and internal control than other local governments, but they achieve equal financial performance level, then the local government is estimated to be .135, .116, and .167 units higher in-service performance than the other local governments respectively. However, the direct effect of financial transparency (β = .1349), accountability (β = .116), and internal control (β = .167) on service performance is insignificant as the bootstrapped confidence intervals contain zero.

Likewise, Table 5.4 shows indirect effects of financial performance on service performance (Equation 5.5). Zero is not found in the lower level and upper level of

95% confidence intervals of financial transparency (β = .0613, 95%CI = .0059 to .1339), accountability (β = .0673, 95%CI = .0054 to .1476), and internal control (β = .0561, 95%CI = .0011 to .1253). The positive sign of the coefficient of financial transparency, accountability, and internal control shows that service performance increases by .061, 067, and 056 units indirectly through financial performance for every unit increase in financial transparency, accountability, accountabil

The second part of mediation analysis is examining the mediation effects of financial performance on the relationships between financial management factors (financial flexibility, financial independence, and resource management) and service performance after controlling for population size, poverty rate, and gender diversity (Equation 5.6). Table 5.4 shows that there are significant effects of financial flexibility (β = .1935, 95%CI = .2971 to .7281), financial independence (β = .3139, 95%CI = .0830 to .1963), and resource management (β = .2448, 95%CI = .0094 to .0334) on financial performance after including control variables since 95% bootstrap confidence intervals are entirely above zero. Also, Table 5.4 presents that the total effects of the financial flexibility (β = .1938, 95%CI = .0041 to .1476), independence (β = .2255, 95%CI = .0290 to .1946), and resource management (β = .2098, 95%CI = .0029 to .0380) on service performance are statistically significant after including control variables since 95% bootstrap confidence intervals are confidence are statistically significant after including control variables since 95% bootstrap confidence intervals are performance are statistically significant after including control variables since 95% bootstrap confidence intervals are entirely above zero.

In addition, Table 5.4 presents the direct effect of financial flexibility, financial independence, and resource management on service performance. The direct effects of financial flexibility (β = .1400, 95%CI = -.0178 to .1274), financial independence (β = .1381, 95%CI = -.0206 to .1576), and resource management (β =

.1417, CI = -.0043 to .0319) on service performance, are positive after including the mediator and control variables. These results mean that if the local government 'A' that has a higher level of financial flexibility than local government 'B', but they have equal financial performance level, so the local government 'A' is estimated to be .140 units higher in-service performance than the local government 'B'. However, after including financial performance and the control variables in the models, the direct effect of financial flexibility, financial independence, and resource management on service performance is insignificant since the bootstrapped confidence intervals are not statistically different from zero.

In Equation 5.6, this study examines the indirect effects of financial management factors (financial flexibility, financial independence, and resource management) on service performance through financial performance. Table 5.4 above presents that zero does not occur between the lower level and upper level of 95% confidence intervals for financial flexibility (β = .0539, 95%CI = .0017 to .1369), financial independence (β = .0874, 95%CI = .0161 to .1847), and resource management (β = .0681, 95%CI = .0116 to .1408). These results show that the service performance increases by .0539, .0874, and .0681 units indirectly through financial performance for every unit increase in financial flexibility, financial independence, and resource management, respectively. Thus, the study findings confirm that financial performance has a mediating role on service performance. Hypothesis 7 is accepted since financial performance has a significant effect on service performance.

7. Robustness Checks

This current study employs two types of robustness checks to determine which combinations of variables would best predict service and financial performance. First, this study conducts stepwise regression analysis. It is the combination of the forward selection approach and backward elimination method that aims to investigate the order of importance of variables and significantly improve the model to predict the dependent variable (Grignon 2019). The stepwise regression process is repeated until no more additional predictors have no statistical significance to the model (Lang 2016). It means that the variables that have a non-significant contribution to the model are excluded from the regression equation.

Table 5.5 below shows the stepwise regression results on service and financial performance (Equation 5.1 and 5.2). In the first step or model, internal control is the single most predictive variable for service performance ($R^2 = .167$, F (1,123) = 24.657, *p* < .001). This result means that the variable of internal control accounts for 16.7% of the variation of service performance. Internal control significantly predicts service performance ($\beta = .409$, *p* < 0.01). In the second model, two variables in combination are predictors of variance in service performance: internal control and accountability ($R^2 = .214$, F (2,122) = 16.632, *p* < .001). Internal control and accountability significantly predict service performance (respectively, $\beta = .314$, *p* < 0.01; $\beta = .237$, *p* < 0.01). The final step of the stepwise multiple regression analysis constitutes the best fit model that indicates three variables in combination are predictors of variance in service performance: internal control, financial transparency, and accountability ($R^2 = .242$, F (3,121) = 12.910, *p* < .001). Internal

control, accountability, and financial transparency significantly predict service performance (respectively, β = .230, *p* < 0.05; β = .190, *p* < 0.05; β = .202, *p* < 0.05).

Service Performance				Financial Performance																								
Independent Variables		Mode	11		Mode	2		Model	3	Independent Variables	1	Model	1		Model	2		Model	3	CIIUII	Mode	, 4		Model	5		Mode	6
	В	SE	β	В	SE	β	В	SE	β		В	SE	β	В	SE	β	В	SE	β	В	SE	β	В	SE	β	В	SE	β
Public Sector Governar	nce:																											
- Internal Control	.199	.040	.409**	.153	.043	.314**	.112	.046	.230*	Internal Control	.203	.035	.465**	.171	.032	.392**	.122	.033	.279**	.078	.035	.180**	.092	.035	.210**	.086	.034	.196*
- Accountability				.293	.108	.237**	.234	.110	.190*	Population Size				039	.008	386**	038	.007	373**	038	.007	373**	034	.007	337**	032	.007	315**
- Financial Transparency							.163	.077	.202*	Accountability							.321	.083	.290**	.259	.083	.234**	.249	.081	.225**	.261	.080.	.235**
										Financial Transparency										.172	.058	238**	.175	.056	.242**	.155	.056	.214**
										Gender Diversity													.271	.102	.177**	.272	.101	.178**
										Poverty Rate																213	.100	143*
R ²	.167			.214			.242			R ²	.216			.359			.430			.469			.499			.518		
Sig.	.000			.000			.000			Sig.	.000			.000			.000			.000			.000			.000		

Table 5.5: Stepwise Multiple Regression Analysis of Public Sector Governance on Service and Financial Performance

Note: * p < 0.05; ** p < 0.01

Including control variables (population size, poverty rate, and gender diversity)

In a similar vein, stepwise regression analysis shows the robustness of the combinations of variables that would best predict financial performance (Equation 5.2), as shown in Table 5.5 above. There are six models of equations based on stepwise regression analysis. In the first step or model, internal control is the single most predictive variable for financial performance ($R^2 = .216$, F (1,123) = 33.879, *p* < .001). This result means that the variable of internal control accounts for 21.6% of the variation of financial performance. Internal control significantly predicts financial performance ($\beta = .465$, *p* < 0.01). In the second model, two variables in combination are predictors of variance in financial performance: internal control and population size ($R^2 = .359$, F (2,122) = 34.233, *p* < .001). Internal control and population size significantly predict financial performance (respectively, $\beta = .392$, *p* < 0.01; $\beta = .386$, *p* < 0.01). The third model of the stepwise multiple regression analysis presents that three variables in combination are predictors of variance are predictors of variance of variance are predictors of variance in financial performance (respectively, $\beta = .392$, *p* < 0.01; $\beta = .386$, *p* < 0.01). The third model of the stepwise multiple regression analysis presents that three variables in combination are predictors of variance in financial performance in financial performance: internal control, population size, and accountability ($R^2 = .430$, F (3,121) = 30.429, *p* < .001). The fourth and fifth models of the stepwise regression indicate the predictors

explain 46.9% and 49.9% of variance in financial performance respectively (respectively, $R^2 = .469$, F (4,120) = 26.539, p < .001; $R^2 = .499$, F (5,119) = 23.708, p < .001). The final model of the stepwise multiple regression analysis constitutes the best fit model that indicates the six variables in combination are predictors of variance in financial performance: internal control, population size, accountability, financial transparency, gender diversity, and poverty rate ($R^2 = .518$, F (6,118) = 21.119, p < .001). The final model of the stepwise multiple regression analysis shows the best-fit combination of independent variables as its model has the greatest variance in financial performance. The higher R square indicates the best-fit model of the regression equation since it measures the percentage of variation of the dependent variable (Studenmund 2014). Internal control, population size, accountability, financial transparency, gender diversity, and poverty rate significantly predict financial performance (respectively, $\beta = .196$, p < 0.05; $\beta = .315$, p < 0.01; $\beta = .214$, p < 0.01; $\beta = .178$, p < 0.01; $\beta = .143$, p < 0.05).

Based on the robustness checks discussed in previous paragraphs, there are no different results from the primary findings that predict a significant relationship between public sector governance (financial transparency, accountability, and internal control) and service and financial performance. These findings indicate that Hypotheses 1, 2, and 3 are accepted after including population size, poverty rate, and gender diversity as control variables. The improvement of the public sector governance of local governments leads to the enhancement of the service and financial performance.

Likewise, Table 5.6 below shows the robustness check of the Equation 5.3. The results of robustness check show that financial flexibility is the single most predictive

variable for service performance ($R^2 = .139$, F (1,123) = 19.818, p < .001). This result means that the variable of financial flexibility accounts for 13.9% of the variation of service performance. Financial flexibility significantly predicts service performance (β = .373, p < 0.01). In the second model, two variables in combination are predictors of variance in service performance: financial flexibility and financial independence (R^2 = .196, F (2,122) = 14.872, p < .001). Financial flexibility and financial independence significantly predict service performance (respectively, β = .294, p < 0.01; β = .252, p< 0.01). The final step of the stepwise multiple regression analysis constitutes the best fit model that indicates three variables in combination are predictors of variance in service performance: financial flexibility, financial independence, and resource management (R^2 = .239, F (3,121) = 12.644, p < .001). Financial flexibility, financial independence, and resource management significantly predict service performance (respectively, β = .197, p < 0.05; β = .236, p < 0.01; β = .231, p < 0.01).

Table 5.6: Stepwise Multiple Regression Analysis of Financial Management Factors on Service and Financial Performance

	Service Performance Financial Performance)																			
Independent Variables		Mode	1		Mode	12		Model	3	Independent Variables		Model 1			Model	2		Mode	3		Mode	4		Mode	5		Mode	16
	В	SE	β	В	SE	β	В	SE	β		В	SE	β	В	SE	β	В	SE	β	В	SE	β	В	SE	β	В	SE	β
Financial Management	Facto	ors:																										
- Financial Flexibility	.146	.033	.373**	.115	.033	.294**	.077	.036	.197*	Financial Independence	.211	. 035 .	474**	.195	.031	.439**	.170	.029	.381**	.148	.029	.332**	.144	.029	.323**	.140	.029	.314**
- Financial Independence				.125	.042	.252**	.117	.041	.236*	Population Size				043	.007	424*'	037	.007	360**	037	.007	365**	035	.007	340**	032	.007	316**
- Resource Management							.022	.009	.231*	Resource Management							.028	.006	.320**	.021	.006	.236**	.021	.006	.238**	.021	.006	.245**
										Financial Flexibility										.074	.025	.210**	.066	.025	.189**	.068	.025	.193**
										Poverty Rate													216	.095	145*	222	.093	149*
										Gender Diversity																.196	.095	.129*
R ²	.139			.196			.239			R ²	.224			.403			.497			.530			.550			.566		
Sig.	.000			.000			.000			Sig.	.000			.000			.000			.000.			.000			.000		

Note: * p < 0.05; ** p < 0.01

Including control variables (population size, poverty rate, and gender diversity)

This study also presents the robustness checks of Equation 5.4, as shown in Table 5.6. Table 5.6 above shows that there are six models of regression equations. The first model indicates that financial independence is the single most predictive variable for financial performance ($R^2 = .224$, F (1,123) = 35.573, *p* < .001). This

result means that the variable of financial independence accounts for 35.6% of the variation of financial performance. Financial independence significantly predicts financial performance (β = .474, *p* < 0.01). In the second model, two variables in combination are predictors of variance in financial performance: financial independence and population size ($R^2 = .403$, F (2,122) = 41.117, p < .001). Financial independence and population size significantly predict financial performance (respectively, β = .439, p < 0.01; β = -.424, p < 0.01). The third model of the stepwise multiple regression analysis shows that three variables in combination are predictors of variance in financial performance: financial independence, population size, and resource management ($R^2 = .497$, F (3,121) = 39.869, p < .001). The fourth and fifth models of the stepwise regression indicate the predictors explain 53% and 55% of variance in financial performance (respectively, $R^2 = .530$, F (4,120) = 33.88, p < .001; $R^2 = .550$, F (5,119) = 29.097, p < .001). The final model of the stepwise multiple regression analysis constitutes the best fit model that indicates the six variables in combination are predictors of variance in financial performance: financial independence, population size, resource management, financial flexibility, poverty rate, and gender diversity ($R^2 = .566$, F (6,118) = 25.642, p < .001). The final model of the stepwise multiple regression analysis shows the best-fit combination of independent variables as its model has the greatest variance in financial performance. Thus, the models of stepwise wise regression show the same findings as the primary models, which support Hypotheses 4, 5, and 6. These findings show that if financial management factors (financial flexibility, financial independence, and resource management) are strengthened, local governments' service and financial performance will be enhanced.

The second type of robustness check is Monte Carlo Simulation developed by MacKinnon, Lockwood, and Williams (2004). This second type of robustness check aims to examine the certainty of Equation 5.5. and 5.6, which investigate the mediation role of financial performance on service performance. Regarding the Monte Carlo test, this study applies the web-based calculator of Monte Carlo Simulation (Preacher 2012; Lachowicz, Preacher, and Kelley 2018). The R version 3.4.4 is generated as a free and open-source statistical computing software to conduct Monte Carlo Simulation (Lachowicz, Preacher, and Kelley 2018). The Monte Carlo approach is comparable with other methods for constructing the confidence intervals of the mediation effects (Preacher and Selig 2012; Zhang 2014).

Table 5.7 below shows that there is a significant mediating effect of financial performance on the relationship between public sector governance (financial transparency, accountability, and internal control) and service performance since zero is not found in the lower and upper levels (respectively, 95%CI= .0062 to .1097; 95%CI= .0133 to .1772; 95%CI= .0025 to .0637). Likewise, Table 5.7 below indicates the mediating role of financial performance on the relationships between financial management factors (financial flexibility, financial independence, and resource management) and service performance. Based on the Monte Carlo simulation, the lower and upper levels of confidence intervals of financial flexibility, financis, financial flexibility, financial flexibility, fina

Dependent Variable	Indonondont Variables	95% Confidence Interva								
	independent variables	LL	UP							
Service Performance	Public Sector Governance									
	- Financial Transparency	.0062	.1097							
	- Accountability	.0133	.1772							
	- Internal Control	.0025	.0637							
	Financial Management Factors:									
	- Financial Flexibility	.0018	.0492							
	- Financial Independence	.0065	.0886							
	- Resource Management	.0001	.0144							

Table 5.7: Monte Carlo Simulation of the Indirect Effects of Financial Performance on Service Performance

8. Conclusion

This chapter presents the findings of the proposed hypotheses using the Statistical Package for Social Sciences version 24 (SPSS 24). The correlation tests, multiple regression analysis, and Bootstrapping method are used to test the study hypotheses. All the assumptions underlying the multiple regression analysis are met. Based on the correlation and the multiple regressions analysis results, it can be concluded that Hypothesis 1, 2, and 3, which predict a positive and significant impact of public sector governance (financial transparency, accountability, and internal control) on service and financial performance, are supported. It means that if public sector governance is strengthened, then financial and service performance will be enhanced. Additionally, this study found that an increase in financial management factors (financial flexibility, financial independence, and resource management) leads to an increase in financial and service performance. Thus, these findings are consistent with Hypotheses 4, 5, and 6.

The findings of this study also support Hypotheses 7, suggesting that the effect of public sector governance and financial management factors on service performance is mediated by financial performance. This study generates stepwise regression and the Monte Carlo Simulation to strengthen the robustness of the analyses. The results show the same as the primary findings. The next chapter analyses the mediating role of financial performance on the relationship between of public sector governance, financial management factors, and environmental performance.

Chapter 6 Local Government's Environmental Performance: Hypotheses Testing

1. Introduction

In this chapter, data analysis and hypothesis testing are conducted to address two research questions. First, what are the influences of public sector governance and financial management factors on environmental performance? Second, are the relationships between public sector governance, financial management factors, and environmental performance mediated by financial performance? Validity and reliability tests are presented prior conducting data analysis to test the hypotheses related to the environmental performance. These tests aim to ensure how well and consistent the indicators of environmental performance used in this study. The linkage of Chapter 6 with the current research objectives are shown in Figure 6.1 below.



Figure 6.1: The Linkage of Chapter 6 with the Research Objectives

2. Descriptive Statistics and Correlations

The descriptive statistics of environmental performance variable shows that the skewness and kurtosis of environmental performance are -.352 and -.640, respectively. These results are within the acceptable limit (between -2 and +2). Mean and standard deviation of the environmental performance are .575 and .173 respectively. The relatively high standard deviation value shows that environmental performance has adequate variability in the data. Table 6.1 below shows Pearson's correlation matrix between public sector governance (financial transparency, accountability, internal control), financial management factors (financial flexibility, financial independence, resource management), and environmental performance. It is found that there is a positive and significant correlation between internal control, financial transparency, accountability, and environmental performance (respectively, r = .409, p < 0.01; r = .411, p < 0.01; r = .371, p < 0.01). These results show that the higher internal control, financial transparency, and accountability, the better the environmental performance of local government.

In respect to financial management factors, it is found that there is a positive and significant correlation between financial flexibility, financial independence, resource management, and environmental performance (respectively, r = .378, p < 0.01; r = .368, p < 0.01; r = .347, p < 0.01). These results indicate if financial flexibility, financial independence, and resource management exist, environmental performance is likely to be achieved. It is also found that financial performance is significantly and positively correlated with environmental performance (r = .606, p < 0.01). This result shows that if financial performance is strengthened, environmental

performance will be enhanced. Thus, it appears the probability of financial performance as a mediating variable.

				[Descrip	otive st	atistics	6		Correlations							
		Variables	Mean	Med.	SD	Min	Max	Sk.	Ku.	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
		1. Environmental Performance	.575	.595	.173	.135	.881	352	640	1							
ce fo		2. Financial Performance	.329	.323	.081	.159	.566	.485	.150	.606**	1						
Sec	٢	3. Financial Transparency	.192	.204	.112	.000	.425	003	-1.025	.411**	.462**	1					
blic over		4. Accountability	.546	.546	.073	.316	.702	387	.529	.371**	.446**	.401**	1				
ηĞ	L	5. Internal Control	.508	.525	.186	.135	.854	079	923	.409**	.465**	.509**	.399**	1			
al nent	٢	6. Financial Flexibility	.495	.488	.231	.000	1.000	.040	731	.378**	.447**	.360**	.370**	.371**	1		
anci agen		7. Financial Independency	.463	.474	.182	.001	.952	.006	045	.368**	.474**	.304**	.357**	.388**	.310**	1	
Fin Jana	L	8. Resource Management	2.56	2.500	.928	1.000	4.000	024	991	.347**	.472**	.292**	.295**	.285**	.442**	.196 [*]	1
~																	

Table 6.1: Pearson's Correlation Matrix

N= 125; Med. = Median; SD = Standard Deviation; Min. = Minimum; Max. = Maximum; Sk. = Skewness; Ku. = Kurtosis * p < 0.05; ** p < 0.01

3. Multiple Regression Analysis

The multiple regression analysis is applied to test the prediction level of financial transparency, accountability, internal control, financial flexibility, financial independence, resource management on environmental performance. Population size (measured by taking natural logarithm), poverty rate, and gender diversity are included as control variables. This study uses Bootstrapping approach generated by PROCESS macro to test the mediating effects of financial performance on environmental performance (Hypotheses 8). The assumptions of the OLS regression analysis are conducted to draw a valid interpretation and avoid the distortion of regression analysis.

3.1 Effects of Public Sector Governance on Environmental Performance

Hypothesis 1, 2, and 3 predict that public sector governance represented by financial transparency, accountability, and internal control has positive effects on

environmental performance. Multiple regression analysis is conducted after controlling for population size, poverty rate, and gender diversity to test the effects of financial transparency, accountability, and internal control on environmental performance. Population size, poverty rate, and gender diversity are included in the analysis since prior literature suggested that those variables would likely influence the environmental performance of organisations (Asadi et al. 2008; Gorenflo et al. 2011; Asteria, Brotosusilo, and Apriana 2018; He and Jiang 2019).

The values for Centered Leverage and Cook's Distance of Equation 6.1 are less than one and close to zero. These results indicate that the outliers do not have any influences on Equation 6.1. The regression model (Equation 6.1) has no evidence of multicollinearity since the VIF value of the regression model is considerably less than the generally accepted maximum level of 10 (Allen, Bennett, and Heritage 2019). There is no autocorrelation since the Durbin Watson value of Equation 6.1 (d =1.887) is between 1 and 3, which is considered an acceptable level. The Kolmogorov-Smirnov test indicates that the assumption of normality of residuals is fulfilled since the *p*-value is .200, which is higher than 0.05 (Drezner, Turel, and Zerom 2010; Ghozali 2016). The assumption of linearity and homoscedasticity of Equations 5.1 and 5.2 are also fulfilled because the scatterplot of the standardised residual graph of both models is scattered (Hair, Black, and Babin 2010; Garson 2012). Based on these analyses, the regression model of Equation 6.1 meets the assumptions of ordinary least squares (OLS) regression analysis.

Table 6.2: The Effects of Public Sector	r Governance on Environmental	Performance
---	-------------------------------	-------------

Independent Variables	Environm	nental Per	formance
	В	SE	β
Public Sector Governan	ce:		
- Financial Transparency	.386	.147	.250*
- Accountability	.421	.209	.178*
- Internal Control	.207	.090	.222*
Control Variables:			
- Population	024	.018	109
- Poverty Rate	.458	.260	.144
- Gender Diversity	.092	.263	.028
R ²	.281		
Sig.	.000		

Note: * p < 0.05; ** p < 0.01

EP= .371 + .250FT + .178AC + .222IC - .109Pop + .144Pov + .028Gdr (Equation 6.1) EP = environmental performance; FT = financial transparency; AC = accountability; IC = internal control; Pop = population size; Pov = poverty rate; Gdr = gender diversity

In combination, financial transparency, accountability, internal control, and control variables accounted for a significant 28.1% of the variability in environmental performance ($R^2 = .281$, F (6,118) = 7.673, p < .001), as shown in Table 6.2. Financial transparency, accountability, and internal control significantly predict environmental performance (respectively, $\beta = .250$, p < 0.05; $\beta = .178$, p < 0.05; $\beta = .222$, p < 0.05). The positive sign means that an increase in the coefficient of financial transparency, accountability, and internal control by one-unit leads to an increase of .250, .178, and .222 in the environmental performance respectively, while other variables held constant. Thus, it can be concluded that the hypotheses H1, H2, and H3 are accepted.

3.2 Effects of Financial Management on Environmental Performance

Table 6.3 below shows that the regression model of Equation 6.2 has no evidence of multicollinearity because the VIF values of the model range from 1.041 to 1.354, which are considerably less than the generally accepted maximum level of 10 (Allen, Bennett, and Heritage 2019). Also, there is no autocorrelation since the Durbin Watson value of the regression model (d = 2.082) is considered acceptable, in a

range between 1 and 3 (Field 2013). Kolmogorov-Smirnov test indicates that the assumption of normality of residuals has been fulfilled since the *p*-value is .200, which is higher than 0.05. The assumptions of linearity and homoscedasticity are fulfilled as the scatterplot of the standardised residual graph of this model are scattered (see Appendix). Thus, the regression model of the Equation 6.2 meets the assumptions of OLS regression analysis.

Financial flexibility, financial independence, resource management, and control variables in combination accounted for a significant 27.1% of the variability in environmental performance ($R^2 = .271$, F (6,118) = 7.307, *p* < .001). It is found that financial flexibility, financial independence, and resource management significantly predict environmental performance (respectively, $\beta = .227$, *p* < 0.05; $\beta = .269$, *p* < 0.01; $\beta = .179$, *p* < 0.05). The positive sign means that an increase in the coefficient of financial flexibility, financial independence, and resource management by one unit leads to an increase of .227, .269, and .179 in environmental performance, respectively, while other variables held constant. Based on the aforementioned hypotheses testing, the study's findings show the empirical support for hypotheses H4, H5, and H6.

Table 6.3: Multiple Regression Analysis of Financial Management Factors on Environmental Performance

	Environn	nental Per	formance
	В	SE	β
Financial Management F	actors:		
- Financial Flexibility	.170	.069	.227*
- Financial Independence	.256	.079	.269**
- Resource Management	.034	.017	.179*
Control Variables:			
- Population	027	.018	126
- Poverty Rate	.399	.258	.126
- Gender Diversity	087	.262	027
R ²	.271		
Sig.	.000		

Note: * p < 0.05; ** p < 0.01

EP= .640 + .227FF + .269FI + .179RM - .126Pop + .126Pov - .027Gdr (Equation 6.2) EP = environmental performance; FF = financial flexibility; FI = financial independence; RM = resource management; Pop = population size; Pov = poverty rate; Gdr = gender diversity

4. The Mediating Role of Financial Performance on Local Government's Environmental Performance

Table 6.4 below presents the mediating effects of financial performance on environmental performance. This study found that the outliers do not have any influence on Equations 6.3 and 6.4 because the values for Centered Leverage and Cook's Distance of both equations are less than one and close to zero. The Equation 6.3 and 6.4 have no evidence of multicollinearity as the VIF values of the models range between 1.041 and 2.074, which are less than the generally accepted maximum level of 10. There is no autocorrelation as well since the Durbin Watson values of Equation 6.3 (d = 1.8), and Equation 6.3 (d = 2.082) are considered an acceptable level. Kolmogorov-Smirnov test indicates that the assumption of normality of residuals has been fulfilled since the *p*-value is .200, which is higher than 0.05. In combination, the independent variables of Equation 6.3 and 6.4 after including control variables account for a significant 46.2% and 27.1% of the variability in environmental performance (respectively, $R^2 = .462$, F (7,117) = 14.353, p < .001; $R^2 = .271$, F (7,117) = 7.307, p < .001).

Table 6.4: Mediating Role	of Financial Performance	on Environmental	Performance
(β, 95% CI). Standardised	Coefficients Results		

Dependent Variable	Independent Variables	a path	Indirect effect (ab)	c path (total effect)	c' (direct effect
Environmental	Public Sector Governance	:			
Performance	- Financial Transparency	.2145 (.0437, .2667)	.1315 (.0498, .2182)	.2499 (.0953, .6775)	.1183 (0779, .4439)
	- Accountability	.2355 (.1022, .4197)	.1444 (.0570, .2382)	.1779 (.0068, .8357)	.0334 (2966, .4550)
	- Internal Control	.1963 (.0179, .1536)	.1204 (.0277, .2176)	.2216 (.0297, .3846)	.1012 (0635, .2528)
	Total model	R ² = .462, F (7, 117) =	: 14.353, p < .001		
	Financial Management Fa	ctors:			
	- Financial Flexibility	.1935 (.0189, .1170)	.1238 (.0127, .2285)	.2268 (.0343, .3063)	.1030 (0452, .1999)
	- Financial Independence	.3139 (.0830, .1963)	.2008 (.1029, .3283)	.2688 (.0987, .4125)	.0680 (0856, .2149)
	- Resource Management	.2448 (.0094, .0334)	.1566 (.0813, .2419)	.1793 (.0003, .0667)	.0227 (0263, .0347)
	Total Model	R ² = .271, F (7, 117) =	= 7.307, p < .001		

All paths are included control variables (population size, poverty rate, and gender diversity)

EP= -.163+.118FT+.033AC+.101IC+.613FP+.084Pop+.232Pov-.081Gdr (Equation 6.3) EP= -.061+.103FF+.068FI+.023RM+.640FP+.076Pop+.221Pov-.109Gdr (Equation 6.4) EP = environmental performance; FT = financial transparency; AC = accountability; IC = internal control; FP = financial performance; Pop = population size; Pov = poverty rate; Gdr = gender diversity; FF = financial flexibility; FI = financial independence; RM = resource management. In this study, the mediating role of financial performance on environmental performance is presented in two parts. The first part investigates the mediation effects of financial performance on the relationships between public sector governance (financial transparency, accountability, and internal control) and environmental performance (Equation 6.3). Table 6.4 above presents that there is a significant total effect of financial transparency (β = .2499, 95%Cl = .0953 to .6775), accountability (β = .1779, 95%CI = .0068 to .8357), and internal control (β = .2216, 95%CI = .0297 to .3846) on environmental performance. As explained in the previous chapter, the total effect is the effect of an independent variable on a dependent variable when the mediating variable is not present in the model. The total effects of the internal control, financial transparency, and accountability on environmental performance are statistically significant after including control variables because 95% bootstrap confidence intervals are entirely above zero. These findings indicate that the increased financial transparency, accountability, and internal control lead to higher environmental performance after including control variables.

Table 6.4 above shows the direct effects of financial transparency, accountability, and internal control on environmental performance. The direct effect is the effect of an independent variable on a dependent variable when the mediator variable is included in the model. After including financial performance in the models, the direct effect of financial transparency (β = .1183, CI = -.0079 to .4439), accountability (β = .0334, CI = -.2966 to .4550), and internal control (β = .1012, CI = -.0635 to .2528) on environmental performance is positive. These results show that if a local government has a higher level of financial transparency, accountability, and internal control than another local government, but they achieve equal financial performance level, then

the local government is estimated to be .118, .033, and .101 units higher in environmental performance than another local government respectively. However, the direct effect of financial transparency, accountability, and internal control on environmental performance is insignificant since the bootstrapped confidence intervals contain zero.

Likewise, Table 6.4 above shows the indirect effects of financial performance on environmental performance. It is found that financial performance mediates the relationship between financial transparency and environmental performance after including control variables indicated by the lower level and upper level of 95% confidence intervals. Zero is not found in the lower level and upper level of 95% confidence intervals of financial transparency ($\beta = .1315, 95\%$ Cl = .0498 to .2182). The positive sign indicates that the environmental performance increases by .1315 units indirectly through financial performance for every unit increase in financial transparency. Likewise, zero does not occur between the lower level and upper level of 95% confidence intervals for indirect effects of accountability ($\beta = .1444, 95\%$ Cl = .0570 to .2382) and internal control ($\beta = .1204, 95\%$ Cl = .0277 to .2176) on environmental performance. These results indicate that the environmental performance increases by .1444 and .1204 units indirectly through financial performance in accountability and internal control, respectively.

The second part of the mediation analysis investigates the mediation effects of financial performance on the relationships between financial management factors (financial flexibility, financial independence, and resource management) and environmental performance after controlling for population size, poverty rate, and

gender diversity (Equation 6.4). Table 6.4 above presents that there are significant total effects of financial flexibility (β = .2268, 95%CI = .0343 to .3063), financial independence (β = .2688, 95%CI = .0987 to .4125), and resource management (β = .1793, 95%CI = .0003 to .0667) without the influence of the mediator variable on environmental performance after including the control variables. The total effects of financial flexibility, financial independence, and resource management on environmental performance are statistically significant since 95% bootstrap confidence intervals are entirely above zero. Thus, an increase in financial flexibility, financial independence, and resource management of environmental performance.

Table 6.4 above also presents the direct effects of financial flexibility, financial independence, and resource management on environmental performance after including control variables and the mediator. The financial performance as a mediating variable and control variables are included in the model to measure the direct effect. The direct effect of financial flexibility (β = .1030, CI = -.0452 to .1999), financial independence (β = .0680, CI = -.0856 to .2149) and resource management (β = .0227, CI = -.0263 to .0347) on environmental performance is positive. These results show that if the local government 'X' has a higher level of financial flexibility than other local governments, but the local government 'X' and the other local governments have equal financial performance level, then the local government 'X' is estimated to be .103 units higher in environmental performance than the other local governments respectively. However, the direct effect of financial flexibility, financial independence, and resource management on environmental performance is performance is insignificant since the bootstrapped confidence intervals contain zero.

Similarly, Table 6.4 above presents that there are the indirect effects of financial flexibility (β = .1238, 95%CI = .0127 to .2285), financial independence (β = .2008, 95%CI = .1029 to .3283), and resource management (β = .1566, 95%CI = .0813 to .2419) on environmental performance via financial performance after including control variables. The indirect effect of financial flexibility, financial independence, and resource management on environmental performance is indicated by the lower and upper-level values that are above zero. The positive sign of the mediating effect of financial performance increases by .1238, .2008, .1566 units indirectly through financial performance for every unit increase in financial flexibility, financial independence, and resource management, respectively. Thus, these findings provide support for Hypothesis 8, suggesting that the effect of financial transparency, accountability, and internal control on environmental performance is mediated by financial performance after including control variables.

5. Robustness Checks

This study employs two types of robustness checks. First, the robustness of the models developed in this study are checked by stepwise regression analysis. The stepwise regression analysis investigates which variables' combinations would best predict environmental performance (Equation 6.1 and 6.2). As shown in Table 6.5 below, there are three models of equations based on the stepwise regression. The first model presents that financial transparency is the single most predictive variable for environmental performance ($R^2 = .169$, F (1,123) = 24.967, *p* < .001). This result shows that the variable of financial transparency accounts for 16.9% of the variation of environmental performance. In the second model, in combination, financial

transparency and internal control account for a significant 22.3% of the variability in environmental performance ($R^2 = .223$, F (2,122) = 17.485, *p* < .001). Financial transparency and internal control significantly predict environmental performance (respectively, $\beta = .273$, *p* < 0.01; $\beta = .270$, *p* < 0.01). The final step of the stepwise multiple regression analysis constitutes the best fit model that indicates three variables in combination are predictors of variance in environmental performance: financial transparency, internal control, and accountability ($R^2 = .253$, F (3,121) = 13,638, *p* < .001). Financial transparency, internal control, and accountability significantly predict environmental performance (respectively, $\beta = .221$, *p* < 0.05; $\beta = .219$, *p* < 0.05; $\beta = .195$, *p* < 0.05). Thus, the results of the robustness checks show the same as the primary findings, which predict a relationship between financial transparency, accountability, internal control, and environmental performance. If the internal control of local governments is improved, then the environmental performance performance of local governments will be enhanced.

Table 6.5:	Stepwise	Multiple	Regression	Analysis	of	Public	Sector	Governance	on
Environme	ental Perfor	rmance							

	Environmental Performance								
Independent Variables	Model 1		Model 2			Model 3			
	В	SE	β	В	SE	β	В	SE	β
Public Sector Governance:									
- Financial Transparency	.635	.127	.411**	.422	.143	.273**	.342	.146	.221*
- Internal Control				.253	.087	.270**	.205	.088	.219*
- Accountability							.461	.210	.195*
R ²	.169			.223			.253		
Sig.	.000			.000			.000		

Note: * p < 0.05; ** p < 0.01

Table 6.6 below shows the results of the robustness checks of Equation 6.2. It is found that financial flexibility is the single most predictive variable for environmental performance ($R^2 = .143$, F (1,123) = 20.546, *p* < .001). This result means that the variable of financial flexibility accounts for 14.3% of the variation of environmental

performance. Financial flexibility significantly predicts environmental performance (β = .378, p < 0.01). In the second model, two variables in combination are predictors of environmental performance: financial flexibility variance in and financial independence (\mathbb{R}^2 = .213, F (2,122) = 16.498, p < .001). Financial flexibility and financial independence significantly predict environmental performance (respectively, β = .292, p < 0.01; β = .278, p < 0.01). The final step of the stepwise multiple regression analysis constitutes the best fit model that indicates three variables in combination are predictors of variance in environmental performance: financial flexibility, financial independence, and resource management (R^2 = .246, F (3,121) = 13.182, p < .001). Financial flexibility, financial independence, and resource management significantly predict environmental performance (respectively, $\beta = .206$, p < 0.05; $\beta = .264$, p < 0.01; $\beta = .204$, p < 0.05). Thus, it can be concluded that the stepwise regression analysis shows the same findings as the primary models.

	Environmental Performance								
Independent Variables	Model 1		Model 2			Model 3			
	В	SE	β	В	SE	β	В	SE	β
Financial Management Factors:									
- Financial Flexibility	.284	.063	.378**	.219	.063	.292**	.155	.068	.206*
- Financial Independence				.264	.080.	.278**	.251	.079	.264**
- Resource Management							.038	.016	.204*
R ²	.143			.213			.246		
Sig.	.000			.000			.000		

Table 6.6: Stepwise Multiple Regression Analysis of Financial Management Factors on Environmental Performance

Note: * p < 0.05; ** p < 0.01

The second type of robustness check is related to the Monte Carlo Simulation of the indirect effects of financial performance on environmental performance. The Monte Carlo Simulation is recommended to check the robustness of the mediation analysis (Rungtusanatham, Miller, and Boyer 2014; Schoemann, Boulton, and Short 2017;

Caron 2019). This study uses the web-based calculator of Monte Carlo Simulation provided as a free and open-source statistical computing software (Preacher 2012). Table 6.7 below shows that zero is not found in the lower level and upper level of 95% confidence intervals of internal control (95%CI= .0237 to .2142). This finding indicates a significant mediating effect of financial performance on the relationship between internal control and environmental performance. Similarly, the lower and upper-level values of financial transparency (95%CI= .0566 to .3755) and accountability (95%CI= .1260 to .5953) are above zero. The Monte Carlo simulation also presents that the lower and upper limit 95%CI of financial flexibility, financial independence, and resource management (respectively, 95%CI = .0257 to .1743; 95%CI = .1033 to .2973; 95%CI = 0109 to .0506) are above zero. These results indicate an indirect effect of financial flexibility, financial independence, and resource management on environmental performance via financial performance. Thus, it can be concluded that the Monte Carlo simulation has the same findings as the primary findings, and Hypothesis 8 is supported.

Dopondont Variable	Indonondont Variables	95% Confidence Interval					
	independent variables	LL	UP				
Environmental Performance Public Sector Governance:							
	- Internal Control	.0237	.2142				
	- Financial Transaprency	.0566	.3755				
	- Accountability	.1260	.5953				
	Financial Management Factors:						
	- Financial Flexibility	.0257	.1743				
	- Financial Independence	.1033	.2973				
	- Resource Management	.0109	.0506				

Table 6.7: Monte Carlo Simulation of the Indirect Effects of Financial Performance on Environmental Performance

6. Conclusion

To conclude, Hypotheses 1, 2, and 3 that predict a positive and significant impact of public sector governance represented by financial transparency, accountability, and internal control on environmental performance are supported after including control variables. These findings show that environmental performance will be enhanced if public sector governance is strengthened. Additionally, this study found that an increase in financial management factors (financial flexibility, financial independence, and resource management) leads to improved environmental performance. Thus, these findings support Hypotheses 4, 5, and 6. Concerning the mediating effects of financial performance, the findings of this study support Hypothesis 8, suggesting that public sector governance and financial management factors have indirect effects on environmental performance via financial performance. The robustness checks also provide the same results as the primary findings. The next chapter presents a discussion of the study.

Chapter 7 Discussion and Conclusion

1. Introduction

This study investigates the relationships between public sector governance, financial management factors, and public service and environmental performance of local government. This study also sought to critically investigate the mediating role of financial performance on public service and environmental performance. The drawing on the institutional theory, this study shows that central government has taken active steps to improve public sector governance and financial management of local governments in Indonesia. This has occurred through the enactment by central government of several regulations, including Indonesian Government Regulation 61/2010, Presidential Regulation 29/2014, Government Regulation 60/2008, Indonesian Law 33/2004, and Law 23/2014, as discussed in Chapter 2. Based on these regulations, the central government has potential coercive power over local governments. There is no market competition within local governments in Indonesia, and this contributes to local governments tending to be highly homogenous in their approaches, without incentive to innovate, be flexible, or strive for excellence of public service delivery. Instead, they tend to simply follow standards, processes, and activities as stated by central government. All subnational governments in Indonesia also still depend heavily on central government for financial resources and awards (Akbar, Pilcher, and Perrin 2012), as discussed in more detailed in the following subsections related to the findings of the study.

2. Service Performance of Local Government

As discussed in Chapter 5, financial transparency significantly predicted service performance, for local governments in Indonesia. It indicates that the more transparency a local government has, the better its service performance. The finding of the present study is in line with those of previous studies, which found that when the local governments provided public-sector financial information to communities, to facilitate monitoring of the local government educational programs, this led to an improvement in the quality of these programs (Reinikka and Svensson 2005; Alcaide Munoz, Rodriguez Bolivar, and Lopez Hernandez 2017; Bauhr and Carlitz 2020). Previous research has indicated that the facilitation of public participation in development planning is a strong example of transparency and an effective way to help meet public needs and expectations (Akbar et al. 2020). Our study's findings align also with those of Bauhr and Carlitz (2020), who found a significant positive relationship between financial transparency of district-level governments in Vietnam, and the quality of their health and education programs.

An efficient tool to increase the financial transparency of local government is to conduct community-based monitoring of public service delivery. Indeed, the availability and accessibility accuracy of disclosure of financial information on local government websites are key components of financial transparency of local government, alongside overall availability and accessibility accuracy of disclosure of financial information. The aim of this requirement is to increase public trust and strengthen the relationship between the local government and communities; and in turn, increase local people's involvement in identifying and prioritising the community's needs. According to the institutional theory, such disclosure of financial

information, constitutes a symbol of trust and modernity adopted by the public sector (Alcaide Munoz, Rodriguez Bolivar, and Lopez Hernandez 2017). Thus, it is expected that when local governments have high financial transparency, this will greatly increase the likelihood of high-quality public service which meets the public's needs.

With respect to accountability, this study also found that accountability significantly predicted service performance for local governments in Indonesia. It means that the more accountable a local government is, the better their service performance. This finding supports those of previous studies which have reported the positive impacts of accountability on public service performance (Kluvers 2003; Hwang 2013; Han and Hong 2016; Ardigó, Martinez, and Jackson 2019). For example, adequate accountability of public sector agencies in Pakistan has been shown to lead to improvements in waste management in that country (Rana et al. 2019). The present study also aligns with the findings of previous studies which have reported that accountability initiatives by government, such as public expenditure tracking surveys, citizen report cards, social audits, and monitoring of government activities by communities, lead to higher-quality health and educational services (Joshi 2013), and increased community satisfaction regarding the provision of public services (Abidin, Singaravello, and Zakaria 2015).

In contrast, low accountability by local government leads to lower quality of public service (Haryanto 2010). Some researchers have proposed that a lack of accountability by local governments may increase citizens' reluctance to pay taxes, given that they do not know what public services their taxes are supporting (Chigwende, Mukuze, and Musasa 2014). Without sufficient finances from taxes,

municipalities are unable to finance important public expenditures such as roads, waste disposal, domestic water supply, and health and education services. In an effort to increase the accountability of local governments, central governments tend to decentralise various responsibilities to lower-level governments. It is hoped that this will reduce unnecessary bureaucracy and increasing efficiencies, local relevance, and overall effectiveness of public services, to better meet local communities' needs (Uhaib and Djani 2018).

Central government's efforts to improve the accountability of local governments include the issuance of Presidential Regulation 29/2014, and the Regulation of the Ministry of Administrative and Bureaucratic Reform 25/2012 and 12/2015. Together these regulations direct all local governments in Indonesia to report on and publish relevant aspects of their activities and performance, such as their strategic plans, annual development programs, standard operating procedures for delivery of services and other processes, and annual performance reports, including progress towards stated goals. The implementation of these accountability measures is assessed by central government, and local governments which achieve adequate levels of organisational accountability will receive public sector awards from central government. Receipt of these awards is considered important for the heads of local government because the awards publicise their achievements regarding the building of public service accountability.

This study also found that internal control significantly predicted service performance for local governments in Indonesia. These findings align with those of previous studies which have reported that more effective internal control associates with better provision of public services (Coram, Ferguson, and Moroney 2006;

Muraleetharan 2011; Vijayakumar and Nagaraja 2012; Badara and Saidin 2013; Tolley, Ridwan, and Kasim 2017; Ong'unya and Abbey 2019). As argued by Nuswantara and Maulidi (2017), effective internal control is expected to improve public service performance because strong internal control reduces the risk of fraud and other corrupt or self-interested behaviour by public officials. Other recent research supports this proposition; for example Salin, Zakaria, and Nawawi (2018) report that fraud and other misuses of funds by public officials is facilitated by a lack of internal control (Salin, Zakaria, and Nawawi 2018). The establishment of mechanisms to ensure adequate internal control within local government is essential for effective implementation of development programs and overall provision of services. This will include effective monitoring and compliance efforts by heads of local government, who have responsibilities to ensure that all their employees follow organisational ethics and standard operating procedures and meet required competencies to perform their duties. It should also include effective capacity to monitor the decisions and actions of senior local government officials; for example, by an adequately resourced and preferably independent 'watch-dog' agency that has strong compliance powers, and by the community and wider public.

With respect to financial management, the present study found that financial flexibility of local governments significantly predicted service performance. The finding of the present study is consistent with those of previous studies (Rubio 2011; Soto, Farfan, and Lorant 2012). Research by Ma and Jin (2016) and Al-Slehat (2019) indicates that financial flexibility leads to better service performance in part by facilitating opportunities of investment, thereby generating economic and social benefits. Financial flexibility can not only generate additional local revenue for local government, it can also increase job opportunities and community well-being

(Kusmayadi 2015). Research by Fiona, Taufik, and Ratnawati (2016) showed that the financial capacity of local government was directly related to their capital expenditure to fulfil community needs. Likewise, a lack of financial flexibility by local government has been linked to difficulties with meeting community needs (Drew and Ryan 2016). Together this indicates that local government flexibility to raise local revenues and other financial resources is an important part of effective service delivery. Financial flexibility empowers the local governments to maintain a sufficient cash balance, which can in turn sustain the provisioning and quality of public services.

The present study also found that the financial independence significantly predicted service performance of local government. This finding aligns with those of Siregar and Pratiwi (2017) and Sri and Suyana (2019), which reported a positive effect of financial independence on service performance, as indicated by the Human Development Index (Niswaty, Manno, and Akib 2015; Sri and Suyana 2019). Similarly, research by Solechah (2013) reported that local governments' independence to manage their local budgets has a significant positive relationship with strong public service outcomes, especially the educational sector.

A range of different types of local revenue can be collected by local government, including from local taxes, retributions, licensing, and income from interest and sale of assets. This revenue can be generated at a relatively low cost, without burdening the private sector. Higher revenue will give local government more capacity to carry out regional developments and deliver services without depending on fiscal transfer from central government or support from other organisations. In contrast, local governments with low self-generation of revenue are more vulnerable (Králik 2019),
especially during financial crises, since they will depend on other resources, such as fiscal transfer and debts, to finance their local expenditures. This financial dependency will limit a local government's capacity and authority to manage its finances (Kotarba and Kolomycew 2014). Financial dependency by local government will increase the asymmetry of power between the of the organisations that are providing the funds, and the local government in question, which may increase the risk that the local government may need to meet the interests of the fund provider at the cost of optimal delivery of public services.

This study's findings also support research hypothesis H6; specifically, this study found that local government's resource management significantly predicted service performance. These findings align with those of previous research reporting that the achievement of budget targets has a positive impact on the efficiency and effectiveness of public service provision (Sarr 2015). Likewise, the present study's findings align with those of Arniwita et al. (2019), who found that budget realisation has a significant positive relationship with organisational performance.

Local governments must plan their expenditures and revenues accurately before the start of the fiscal year to achieve effective resource management. If local governments can achieve their revenue targets and have cash available, this will give them flexibility to deliver their public services appropriately. It is also important for local governments to conduct their public spending within appropriate timeframes, as timely budget spending stimulates local economies and the availability of services. In contrast, slower realisation of budgets may affect economic growth, especially during financial crises (Prasidya 2020), and may delay the procurement of public services; with risks that in the absence of adequate oversight and quality

control, local governments may deliver poor-quality works or services to meet time deadlines. When budget realisation by local government is too slow, this poses two risks; the risk of poorer-quality public services (in the absence of effective quality control), and the downsides of cash sitting idle and depreciating without being used (Abdurrohman and Marsus 2017; Erlina and Muda 2017).

Thus, it can be concluded that many previous studies found that public sector governance and financial management factors were associated with organisational performance in the context of developed countries. However, there has been little research into the relationship between public sector governance, financial performance, and public service performance in developing countries. Therefore, the findings of this current study about these relationships in the context of the Indonesian local governments add a valuable theoretical contribution to the public service performance research in developing countries.

3. Financial Performance of Local Government

This study found that each of the public sector governance measurements examined – financial transparency, accountability, and internal control – significantly predicted financial performance of local government. This finding aligns with those of previous studies (Alt and Lassen 2006; Montes, Bastos, and de Oliveira 2019), which have found that financial transparency contributes to fiscal performance as indicated by low public debt and deficit, with financial transparency enhancing fiscal discipline and facilitating the control of public expenditure. These findings align also with findings by Mashayekhi and Kalhornia (2016) that transparency of financial information leads to investment efficiency. The disclosure of financial reporting can improve decision-making and reduce the risk of under- or over-investment (Mashayekhi and Kalhornia

2016). Public sector transparency has also been found to have a positive effect on Foreign Direct Investment inflows, which are a valuable tool for local government to increase economic development (Seyoum and Manyak 2009) and financial stability (OECD 2003). Other studies have reported similarly that when local governments implement professional and transparent administrative services, there is an increase in the flow of international capital (Dorożyński, Dorożyńska, and Urbaniak 2014).

Financial transparency reduces the risk of fraud, other corruption, and financial scandals, as the public can monitor misuse of budgets and submit reports and complaints if they have concerns. This in turn encourages local governments to act with professionalism and integrity, and put their own monitoring in place to ensure they comply with regulations when undertaking their regional financial management.

Likewise, for local government in Indonesia, this study found that accountability significantly predicted financial performance. This finding aligns with those of the previous studies, which have reported that accountability plays an important role in maintaining financial efficiency in the public sector (Kluvers 2003; Kim and Lee 2010; Han and Hong 2016; Nasution 2018; Rana et al. 2019). The present study also aligned with findings by Nasution (2018) and Karim and Mursalim (2019) that accountability has a positive effect on the financial performance of local government. The higher the level of accountability of local government institutions, the lower the risk of misuse of finances (Sari, Arza, and Taqwa 2019). In contrast, a lack of financial accountability leads to reduced efficiency and effectiveness of financial management (Rabrenović 2007).

With respect to internal control, this study also found that internal control of local government significantly predicted financial performance. This finding aligns with those of earlier studies which have reported positive relationship between internal control and the financial performance of local government (Karim and Mursalim 2019). This finding also supports previous studies (Odoyo et al. 2013; Collins 2014; Channar, Khan, and Shakri 2015; Al-Thuneibat, Al-Rehaily, and Basodan 2015; Florio and Leoni 2017) that have reported positive effects of internal control on earnings, liquidity, return on assets, market value-added, and return on equity. Previous research has also reported that organisations without adequate internal control will face financial challenges (Gamage, Lock, and Fernando 2014).

This study's findings align also with those of Gololo (2018) and Chepsikor (2014), who have both reported that internal control has a positive effect on local government's financial management. Internal control preserves assets and efficiency, and facilitates the accuracy of accounting information (Adekola and Olayinka 2018). Local governments with effective internal control will have the capability to manage their local budgets efficiently, in turn facilitating financial solvency. Mature internal control is supported by well-qualified and experienced internal audit unit staff, who will monitor for compliance with policies and regulations and thereby reduce the risk of financial mismanagement. Sound financial management will, in turn, facilitate the attainment of development program targets and overall sound public service delivery.

Local governments in Indonesia use an electronic internal control system developed by the Finance and Development Supervisory Agency of Indonesia to evaluate and report on the implementation of internal control. This supervisory agency acts as a

representative of the central government has and is responsible for improving the effectiveness of internal control at the level of local government and other public institutions. The internal control units of local governments investigate and evaluate areas of financial management that are likely to have a significant influence on financial performance overall, such as local budgets and asset management. When internal control processes are conducted effectively, this facilitates the achievement of revenue and expenditure targets, and supports overall financial performance.

Similarly, this study found that each of the financial management factors examined – financial flexibility, financial independence, and resource management – significantly predicted financial performance of local government. This finding aligns with those of previous studies which have found that financial flexibility leads to higher levels of cash holdings as well as lower leverage ratios; which in turn supports effective financial performance within an organisation (Yung, Li, and Jian 2015; Embaye and Haile 2019). Likewise, the findings are aligned with previous research which has demonstrated a positive influence of financial flexibility on the organisation's value (Tobin's Q) (Bilyay-Erdogan 2020), as well as other measures of financial performance such as return of assets and return on equity (Embaye and Haile 2019).

When local governments have sufficient financial flexibility, this allows them to take advantage of investment opportunities to increase their local government revenues. They are also more likely to be able to finance their local expenditures without depending on external funds. As a result, they can maintain low levels of leverage, especially during financial crises. A low level of financial leverage is likely to give greater financial flexibility to generate revenue and improve financial performance overall (Embaye and Haile 2019). Financial flexibility is particularly useful during

financial crises because it can act as a buffer against high uncertainty and external financial constraints (Tran 2019). The availability of cash holdings alleviates the need for an organisation to become indebted to external institutions. For local governments, the higher their cash value, the lower their debt; with positive effects on both short- and long-term solvency.

Here again, this study found that financial independence of local government significantly predicted its financial performance. These findings align with those of previous studies which have reported the positive effects of financial independence on financial performance for local government (Antari and Sedana 2018, Ningrat and Supadmi 2019, Apridiyanti 2019, Yuliansyah, Susanti, and Yanti 2019, Prastiwi and Aji 2020). The findings of this study also align with prior studies which have shown that the fiscal independence of local government facilitates efficient spending (Geys, Heinemann, and Kalb 2010; Renyaan et al. 2012; Boetti, Piacenza, and Turati 2012). Financial independence is an important factor for local government to enable funding from local taxes and other local revenue generation, which can be used to finance local expenditures and meet the expectations and needs of local communities without excessive interference from external funding bodies.

As described above, the central government in Indonesia has enacted Laws 33/2004 and 23/2014, to support the autonomy and financial independence of local government to obtain their own local revenues and manage their financial resources. The financial independence of local governments is not absolute; the central government retains functional and fiscal relations with subnational governments; for example, to undertake fiscal transfers to manage financial inequities among local governments. With respect to resource management, this study found that for local

governments in Indonesia, local resource management significantly predicted financial performance. In this study, resource management refers to the capacity of local government, over a fiscal year, to ensure that their approved budget matches their budgetary target achievements for both revenues and expenditures. These findings align with those of prior studies which have reported a positive effect of revenue and expenditures target achievement on local governments' financial performance (Priagung, Sinoeraya, and Mafudi 2016). When revenue targets are achieved, local governments will have short- and long-term solvency to finance their local expenditures, which in turn leads to good financial performance. Likewise, when expenditure targets are met and not over-reached, the associated local budgets and cash holdings will have greater capacity to fulfil service delivery and community needs (Fajar and Arfan 2017). Local governments are encouraged to ensure that their budget targets are met on time (Priagung, Sinoeraya, and Mafudi 2016), to avoid the problem of 'idle cash' that might otherwise arise (Gunarto, Sentri, and Said 2018; Okoye et al. 2019).

Based on the abovementioned findings, it can be concluded that many prior studies found positive impacts of public sector governance and financial management factors on financial performance in the context of private sectors or developed countries. However, there are limited research found these relationships in the context of public sector in developing countries. Therefore, the findings of this current study about these relationships add to the existing knowledge in the areas of financial performance of the public sector in developing countries.

4. Mediating Role of Financial Performance on Service Performance

The relationship between accountability and service performance was likewise found to be mediated by financial performance. Here, local government accountability, for example, compliance with required standards, regulations, and laws, enables the government to maintain adequate financial performance and solvency, which in turn supports service performance. Finally, this study found that the relationship between internal control and service performance is also mediated by financial performance. Here, local governments with strong internal control tend to manage their finances well, which in turn facilitates effective public service performance. The necessity to improve, the internal control of local governments in Indonesia has been recognised by the central government. To this end, as described earlier, the central government has issued several regulations to encourage local governments to undertake effective internal control, which in turn should improve financial performance and ultimately service performance.

Together these findings suggest that local governments in Indonesia which have with high levels of financial transparency, accountability, and internal control will be more likely to deliver sound financial performance, with less risk of fraud and other mismanagement of funds. In turn, the sound financial performance should increase the local governments' capacity to provide effective public services (Batafor 2011; Tamawiwy, Sondakh, and Warongan 2016; Tampubolon 2019). In contrast, when local governments have poor financial performance, they face associated difficulties in to delivering their social, health, and economic development services (Bauducco, Bulíř, and Čihák 2011; Carlson, King, and Lewis 2011; Creel, Hubert, and Labondance 2015). This study's findings that financial performance can act as a mediating variable between public sector governance (financial transparency, accountability, and internal control) and service performance are supported by previous research; a number of previous studies have described an indirect positive relationship between governance factors and organisational performance, through the mediating role of financial performance (Haryono and Paminto 2015; Purnomosidi et al. 2014; Afiani and Bernawati 2019; Fatimah, Mardani, and Wahono 2019; Suryaningtyas and Rohman 2019).

This study also found that for each of the financial management factors – financial flexibility, financial independence, and financial resource management – their relationship with service performance is mediated by financial performance. For example, when financial management is sound and local government has adequate cash holdings available, this can increase their short-term solvency, which can support the provision of public services. Similarly, adequate financial flexibility, financial independence, and achievement of budget targets will enable local government to raise local revenue, spend local expenditures, and manage their short- and long-term solvency as all of which are indicators of sound financial performance and which in turn increase the effectiveness of public service delivery.

Based on these findings, financial flexibility, financial independence, and effective management of financial resources assist with the achievement of revenue targets, which in turn supports the effective delivery of public services by local government. Therefore, financial performance mediates the relationships between each of these financial management factors (financial flexibility, financial independence, and resource management) and service performance. These findings align with those of the prior studies which have reported that when local governments have high levels

of financial flexibility and financial independence, this has a positive effect on their financial performance, and that this in turn improves their public service delivery to communities (Cantarero and Pascual 2008; Geys, Heinemann, and Kalb 2010; Rubio 2011; Soto, Farfan, and Lorant 2012; Meier, Bozec, and Laurin 2013).

Based on the abovementioned findings, it can be concluded that the previous studies have not investigated the mediating role of financial performance on public service performance. Therefore, the findings of this present study enrich the management accounting literature because this current study provides evidence of the relationship between public sector governance, financial management, and service performance through the mediating role of financial performance.

5. Environmental Performance of Local Government

For local government in Indonesia, financial transparency significantly predicted environmental performance: the more transparent the local government, the better its environmental performance. This finding aligns with previous studies which have shown a positive relationship between financial transparency and environmental performance (Bernauer and Koubi 2009; Halkos and Tzeremes 2014; Sahide et al. 2016). As one of the instruments of democracy, financial transparency has been found to have a significant positive effect on air quality (Bernauer and Koubi 2009) and is an important aspect of the sustainable management of forests (Wollenberg et al. 2006). The commitment of local governments to manage their local forests and environments sustainably faces many challenges, particularly given the influence of powerful developers and industries, which are reluctant to adopt sustainable development practices that they perceive may increase costs and reduce marginal net profits.

When local governments make their financial information readily available and accessible to the public, this can empower local communities and relevant stakeholders to monitor local governments' actions towards safeguarding the environment. The commitment of local government to protect the environment can only be assessed accurately by independent monitoring to check for compliance with environmental sustainability principles and practices in development programs and other initiatives; for example, programs focused on forest rehabilitation and revegetation, sustainable irrigation and water management, and environmental education initiatives in schools and communities are all programs that can help to support the environment. Transparency of financial information, for example, the timely and full publication of relevant information on the internet, facilitates this monitoring by local communities, non-governmental organisations, and other key stakeholders.

Here again, for local governments in Indonesia, this study found that accountability significantly predicted environmental performance: higher accountability meant better environmental performance. This finding aligns with those of previous studies which have reported the positive effects of accountability on environmental performance (Han and Hong 2016; Jeremiah 2018; de Silva, Yapa, and Vesty 2020). One author proposes that the influence of the accountability on environmental performance can be explained by Vroom's Expectancy Theory, which states that rewards and punishments act as accountability drivers which influence personal behaviour and performance (Chinander 2001). This highlights the importance of the senior management in an organisation ensuring that all employees are familiar with environmental regulations, products and technologies, and that monitoring is effective to ensure compliance with requirements (Chinander 2001).

With respect to internal control, this study also found that for local governments in Indonesia, internal control significantly predicted environmental performance. This finding aligns with those of the prior studies which have reported that the higher the effectiveness of internal control, the higher the level of environmental sustainability (Enofe and Uwaifo 2016; Li et al. 2018; Liu 2018; Kareem, AlHassani, and Ali 2019). The findings align also with studies which have found that high-quality internal control has a significant positive effect the environmental performance of organisations as measured by environmental information disclosure (Huang and Huang 2020) and sustainable development (Li et al. 2018).

As discussed in Chapter 4, the enactment of Indonesian Law 23/2014 has increased the authority of local governments to manage forest use permits, mining land rent, and mining royalties. However, without effective oversight and management by higher authorities, In Indonesia (and elsewhere) this has led to increased misuse and destruction of forests and the environment, as local elites prioritise self-interest over sustainable environmental practices (Wollenberg et al. 2006; Saito-Jensen et al. 2015). Given these demonstrated and serious problems when local government autonomy is too great, local authorities need to be managed effectively, with sufficient monitoring and compliance efforts to avoid decisions and policies that do not align with environmental sustainability. In Indonesia, most deforestation and other destruction of natural resources is caused by the issuing of resource exploration permits that do not meet the requirements or standards of clean and clear exploration of business permits. Those tasked with issuing permits and making decisions often choose short-term self-interest (such as bribes or cronyism) over longer-term environmental public interest, unless an effective system is in place to minimise this (Wollenberg et al. 2006; Saito-Jensen et al. 2015). Such a system must include both effective internal control by local government and effective independent oversight, as identified by Indonesian Government Regulation 60/2008 and Presidential Decree 9/2014.

Required elements of internal control for local government have been specified and mandated by central government (Government Regulation 60/2008, the President Decree 9/2014). As stated in these regulations, for effective internal control, local governments are obliged to develop (and follow) ethical and environmentally sustainable policies, guidelines and standards, with associated information systems and regular reviews of policies and guidelines. Local governments also need to develop effective mechanisms for risk analysis, communication, and monitoring, to ensure consistency of decisions and actions with environmentally sustainable development. Local governments are also required to support the inspectorates and the internal control task forces within local government units, who are tasked with working with the head of each local government to ensure good governance and environmentally sustainable development programs.

Likewise, this study found that each of the financial management factors significantly predicted environmental performance of local government. These findings align with those of previous studies, which found local governments with high levels of financial flexibility and financial independence are more likely to be concerned about environmental sustainability (García-Sánchez, Frías-Aceituno, and Rodríguez-Domínguez 2013; Alcaraz-Quiles, Navarro-Galera, and Ortiz-Rodríguez 2014; Zhang, Zhang, and Steklova 2020). Other recent research has proposed that when local governments are able to obtain financial resources in a timely manner, this

increases their opportunities to implement their environmental sustainability programs effectively (Zhang, Zhang, and Steklova 2020). Previous research has also identified the importance of the timely spending of budgets on environmental projects; it was found that for many major environmental problems in Indonesia, such as large-scale forest fires, governments did not respond in time to avert disaster; and one of the reasons was a lack of funds by local governments (Trinirmalaningrum et al. 2015).

To conclude, while previous research found that public sector governance and financial management factors in the private sector were positively related to environmental performance, the findings of this current study present a significant relationship between public sector governance, financial management, and environmental performance in the context of the Indonesian local governmental performance in the public sector governance regarding environmental performance in the public sector.

6. Mediating Role of Financial Performance on Environmental Performance

This subsection investigates the mediating role of financial performance in the relationships between public sector governance (financial transparency, accountability, and internal control) and environmental performance. This study found that for each of the relationships of financial transparency, accountability, and internal control with environmental performance, the relationship was mediated by the local government's financial performance. Based on these findings, local governments with high levels of financial transparency, accountability, and internal control are more likely to deliver strong financial performances, and this in turn improves their ability to deliver environmentally sustainable public services.

These findings align with the concept of 'Triple Bottom Line' (Elkington 1998), which states that organisational performance should be measures not only by considering financial performance, but also social and environmental performance; often summarised as 'profit, people and the planet' or 'the three pillars of sustainability'). The Triple Bottom Line model states that organisations should focus not only generating revenues, but also on performing both socially and environmentally; and that these three aspects of performance are often positively associated each other; for example, disclosure of financial information represents a commitment to both social and environmental responsibilities, which in turn reduces the risk that poor social or environmental outcomes will compromise future economic outcomes.

This study's findings regarding the mediating role of financial performance on the relationships that financial transparency, accountability, and internal control each have with environmental performance also align with slack resources theory (Waddock and Graves 1997); which suggests that management of social and environmental needs is influenced by the availability of financial resources. These findings align with those of studies that have reported that financial performance and environmental performance are positively related (Orlitzky, Schmidt, and Rynes 2003; Al-Tuwaijri, Christensen, and Hughes li 2004; Salama 2005). In another study of the environmental performance of Czech pollution-emitting firms, sound financial performance was found to improve environmental performance; the authors stated that this is consistent with the proposal that greater financial liquidity increases capacity to invest resources in environmental management (Esty and Porter 2005). Likewise, this study's finding is commensurate with previous studies that found financial performance has a positive relationship with environmental performance. Good financial performance increases the ability to invest more resources in

alleviating pollution problems (Earnhart and Lizal 2006). Thus, higher levels of public sector governance by local governments will increase their capacity to improve their financial performance, which in turn can positively influence their environmental performance.

While this study identifies that for local governments in Indonesia, financial performance is a mediating variable in the relationships that financial transparency, accountability, and internal control have with environmental performance, it is critical to clarify that this research does not claim that financial performance is the only mediating variable. Nor does this research say anything about the relative influence of financial performance, compared with other mediating variables. Indeed, it is likely that other important mediating variables exist. For example, research across 40+ countries has revealed that at the national level, although a positive relationship exists between financial performance (using GDP) and various indicators of relationship environmental performance, this is not straightforward; and environmental performance is also influenced by a country's regulatory regime and its 'economic and social context', including its extent of good governance, civil and political liberties, and scientific and technological capacity (Esty and Porter 2005). These caveats do not, however, negate the finding that financial performance plays a mediating role on environmental performance.

Similarly, this study found that for each of these financial management factors, their relationship with service performance is mediated by financial performance. When the local government has appropriate financial flexibility, financial independence, and financial resource management, this can improve its financial capacity, which can in turn support their capacity to deliver effective environmental performance. Financial

flexibility and financial independence can enable local governments to raise and spend revenue and spend those revenues, and enhance their financial solvency, in turn increasing their capacity to undertake development programs that support environmental sustainability. Claims that organisations must make choose between financial performance and environmental performance are an example of a 'false dichotomy' (Aras and Crowther 2008) because financial performance can increase the opportunities of environmental sustainability programs (Wahba and Elsayed 2015); strong environmental performance is a factor in strong financial performance (Salama 2005). and likewise strong financial performance can improve environmental performance (Wahba and Elsayed 2015) as this study demonstrates.

Similar to the caveat mentioned earlier, while this study identifies that for local governments in Indonesia, financial performance is a mediating variable in the relationship between key financial management factors and environmental performance, it is critical to clarify that this research does not claim that financial performance is the only mediating variable. Nor does this research say anything about the relative influence of financial performance, compared with other mediating variables. As described above, research across a large number of countries has found that environmental performance at the national level is correlated with national wealth (GDP), but not always; and that environmental performance is also influenced by the effectiveness of environmental regulatory regimes, and by the extent of a country's civil and political liberties and scientific/technological capacities (Esty and Porter 2005). With these caveats, however, it remains clear that financial performance.

While the previous research found a direct effect of financial performance on environmental performance, the findings of the present study contribute to the body of knowledge about the influences of public sector governance and financial management on the environmental performance of local governments through the mediating role of financial performance. Thus, this study is one of the first to have examined the mediating role of financial performance on environmental performance of local governments.

7. Theoretical and Practical Contributions of the Study

Scholars and practitioners have investigated the relationships between public sector governance, financial management factors, and service performance of the public sector in many developed countries. However, much less research into these relationships has been conducted in developing countries. Likewise, with much of the environmental performance literature focusing on the private sector, there remains a gap in research on environmental performance in the public sector, especially at the local-government level. This study addresses these gaps and adds to the existing knowledge in the areas of service and environmental performance of local governments in developing countries; by providing both theoretical and practical contributions. Details of these contributions are discussed in the following sections.

First, this study contributes to relevant theory by supporting the concept that institutional isomorphism acts as the underpinning framework to explain the practices of public sector governance (financial transparency, accountability, internal control), and financial management factors (financial flexibility, financial independence, and

resource management) in the public sector. This study's findings support those of previous researchers (e.g., Fogarty 1996; Modell 2001; Akbar, Pilcher, and Perrin 2012) that coercive isomorphism is a major factor influencing the implementation of public sector governance and financial management factors by local governments. There is no market competition within local governments in Indonesia; instead, local governments tend to prioritise organisational homogenisation, and try to maintain institutional legitimacy through political power and their compliance with regulations issued by central government.

A second contribution of this study to the theory is its contributions to support slack resources theory proposed by Waddock and Graves (1997), which suggests that organizational performance is influenced by the availability of financial resources. The availability of spare or 'slack' resources empowers local governments to invest more resources in improving organizational performance (Stanwick and Stanwick 1998; Earnhart and Lizal 2006; Qi et al. 2014). As previously mentioned, the empirical results show the mediating role of financial performance on the relationships between public sector governance, financial management factors, and service and environmental performance. These findings align with the previous studies that have reported the association between financial performance and environmental performance (Orlitzky, Schmidt, and Rynes 2003; Al-Tuwaijri, Christensen, and Hughes li 2004; Salama 2005); although in the former paper, the authors were interested in the potential influence of environmental performance on financial performance on the potential performance of environmental performance on the section section between financial performance on financial performance interested in the potential influence of environmental performance on financial performance rather than vice versa.

A third contribution of this study is to contribute to the growing body of literature on public sector governance, and public financial performance, by performing a

simultaneous analysis of the effects of public sector governance (financial transparency, accountability, internal control) on the financial performance of local government. The study found that the disclosure of financial information via the internet is an effective and efficient tool to empower local communities in Indonesia to monitor local government budget processes. Specifically, in agreement with Benito and Bastida (2009) and Tejedo-Romero and de Araujo (2018), this financial transparency reduces the asymmetry of information between local government and the community. Moreover, this study supports the findings of Rabrenović (2007), Harrison and Horngren (2008), Kim and Lee (2010), and Han and Hong (2016) that financial transparency and accountability reduce the risk of corruption and other misuses of power and finances at local-government level, and improve local government efficiency, financial productivity, and overall effectiveness. This study also shows that internal control facilitates financial performance, for example by fostering efficient and effective revenue collection and expenditure while reducing financial mismanagement.

A fourth contribution of this study is to add to the body of knowledge regarding the relationships that key financial management factors – financial flexibility, financial independence, and financial resource management – have with financial performance for local governments in Indonesia. This study's findings suggest that local government capacity to preserve cash holdings can minimise their dependency on external institutions; and support Petkovska (2011) and Scutariu and Scutariu (2015), who suggest that adequate levels of local government autonomy are necessary to support their capacity to raise their own revenue and spend this revenue in ways that meet local community needs.

A fifth contribution of this study is to enrich the management accounting literature; as this study develops theoretical models which investigate the mediating effect of financial performance on the relationships between public sector governance, financial management factors, and service performance. Despite the hypothesised role of financial performance in mediating the relationships between public sector governance, financial management factors, and service performance, such hypotheses have not previously been tested empirically. This research provides empirical evidence of the relationships between, on the one hand, public sector governance, and key financial management factors, and service performance on the other hand; as well as empirical evidence of the mediating role of financial performance. This is a valuable theoretical contribution to the field of financial and service performance research.

A sixth contribution of this study is to add to the limited literature regarding environmental performance in the public sector. While many previous studies have investigated environmental performance in the private sector, including how public sector governance and financial management factors in the private sector are related to environmental performance, there has been little research into these relationships in the public sector. This study makes a useful contribution by presenting findings about these relationships in the context of the Indonesian public sector, with likely implications for our understanding of these relationships in local governments in other developing countries.

A seventh contribution of this study to the theory in this field is its contribution to understanding the relationships between public sector governance (financial transparency, accountability, and internal control), financial management factors

(financial flexibility, financial independence, and resource management), financial performance, and environmental performance. Although the variables of public sector governance, financial management factors, and financial performance are necessary components of sound environmental performance by local government, this study is one of the first to have investigated relationships among these variables, particularly in the context of a developing country. This study has also developed a theoretical model that explores the influences of multiple variables on the environmental performance of organisations through the mediating role of financial performance.

This study makes a practical contribution to efforts to improve government and corporate environmental performance, by highlighting the positive effects of public sector governance and financial management factors on financial performance and, thereby, increasing the capacity for sound environmental performance. This study demonstrates that strong financial performance can help to improve the environmental performance of local government, specifically when combined with adequate levels of financial transparency, accountability, internal control, financial flexibility, financial independence, and resource management.

Broadly, this study's findings have direct practical applications for central government, policymakers, and local government authorities in developing countries to improve local governments' service and environmental performance. This study's findings indicate that to improve service and environmental performance, local government authorities should strengthen their public sector governance and financial management to strengthen their financial performance and, ultimately, their service and environmental performance should equate

to short- and long-term solvency and both budgetary and service-level solvency. This study proposes the central government policymakers should give local governments sufficient autonomy to raise local revenue without dependency on fiscal transfer from the central government; and this autonomy should always be made available in the context that local government authorities can demonstrate effective internal control and disclosure of financial information. This study's findings that high levels of internal control and disclosure of information make a positive impact on local government's performance have immediate practical applications for local governments, central governments, and the communities they serve.

8. Limitations of the Study

While This study's findings offer several theoretical and practical contributions to the field, as outlined above, these findings should be considered in the context of the study's limitations. One limitation is that this study's empirical analyses have been conducted using data from local governments in Indonesia. The findings may not be applicable to some other countries, depending on government and other contextual similarities and differences. A second potential limitation of this study used data from a single year (2016), and therefore reflects a single point in time. It would be expected that as policies and other contextual factors change, so too may the findings. While using panel data or other longitudinal data would have provided a more nuanced understanding of the links between public sector governance (financial transparency, accountability, internal control), financial management factors (financial flexibility, financial independence, and resource management),

financial performance, and service and environmental performance, this was not feasible within the time constraints of PhD research.

A third limitation of this study is that time and funding constraints prevented the inclusion of both quantitative and qualitative research approaches. Using a combination of both quantitative and qualitative methods would have provided a greater understanding of the institutional constraints that influence local governments' ability to optimise their service and environmental performance. A final limitation of the study is suggested by the values of the R² in the models, which indicates the existence of other factors that are not identified or included in this study. It is important to recognise that other key explanatory variables are likely to be present that may influence service and environmental performance as indicated by the literature (e.g., Esty and Porter 2005). A larger study could identify additional potential factors and add them to the models to test their influence; however, this was not possible in this current research due to the time and funding constraints of PhD research.

9. Recommendations for Future Research

Several future research directions have been identified which could mitigate the limitations of this study and extend the research. First, it would be useful to test the theoretical models in other contexts – including contexts that are both like and different from the context of this research – to identify the extent to which this study's empirical insights may be relevant and applicable in other contexts. A second recommendation is to undertake a longitudinal study to investigate in more detail the short- and long-term effects of public sector governance, financial management

factors, and financial performance on service and environmental performance. Analysing the same set of samples over time would help to identify and understand any changes and patterns that occur over an extended period.

A third recommendation is that further research could include both quantitative and qualitative methodologies. The qualitative research could include in-depth semistructured interviews with a wide range of government officials and other relevant stakeholders to examine the institutional and other constraints that influence local government's ability to provide effective service and environmental performance. Another recommendation is that future research could, first, identifies key strengths and weaknesses of local governments' financial, service, and environmental performance in a particular country or context, and use this information alongside other relevant information from peer-reviewed literature and other reliable sources to design and test a set of guidelines for effective local government performance. Finally, future studies could investigate other variables that may affect financial, service, and environmental performance; including, for example, innovation, managerial styles, and the sets of factors identified by previous research as being important for environmental performance at the national level, including the local regulatory regime and its economic and social context (Esty and Porter 2005).

10. Conclusions

This study is a contribution to the understanding of local government performance in a developing country: Indonesia. This conclusion section summarises the previous chapters, which have presented findings regarding the relationships between public sector governance (financial transparency, accountability, internal control), financial

management factors (financial flexibility, financial independence, and resource management), and the service and environmental performance, for local government in Indonesia. First, the literature review for this thesis identified and discussed existing gaps in this research field, including about the need to investigate service and environmental performance in the public sector, especially in developing countries. This research aimed to address this research gap by developing conceptual models to help understand the relationships between public sector governance, financial management factors, and service and environmental performance of local governments in Indonesia, through the mediating role of financial performance. This thesis has discussed how local governments in Indonesia currently have relatively extensive political, administrative, and financial autonomy to conduct their responsibilities, as central government has largely decentralised its authority to local government (while retaining control of foreign policy, national defence and security, the judicial system, fiscal and monetary policy, and religious affairs. However, central government still influences local governments in other areas through laws and other regulations, with the aim of overseeing and improving local government performance.

This thesis used multiple regression analysis and bootstrapping methods to test the four conceptual models and four research questions developed for this research. In the first conceptual model, public sector governance represented by financial transparency, accountability, and internal control are each positively associated with the local government's financial and service performance. This thesis noted full disclosure by local governments of financial and information and government decisions, plans and actions via the internet, and a demonstrated commitment to

accountability, reduces the asymmetry of information between local government and communities, which in turn empowers the public and other stakeholders to conduct community-based monitoring of public service delivery. Likewise, a demonstrated commitment to effective internal control by local government improves service performance, by reducing the risk of corruption and other government mismanagement and underperformance at the local level. The establishment of bestpractice internal control measures, including policies and guidelines regarding organisational ethics; employee competency and training; standard operating procedures; risk assessments; and effective monitoring and compliance, are essential for effective organisational performance. This research also highlights the role of financial performance in mediating the relationships that public sector governance has with service performance.

The findings from this study's second conceptual model indicate that for local government in Indonesia, financial management factors measured by financial flexibility, financial independence and financial resource management are each positively associated with financial and service performance. Local governments with greater financial flexibility and better achievement of financial targets will have more opportunities to increase their local revenue and cash holdings, thereby reducing their dependency on external funds. Adequate levels of flexibility and independence of local governments to manage their financial budgets are important to enable them to meet community needs. This study showed that local governments' financial performance is a mediating variable that helps to explain the positive influence of financial management factors on service performance.

The findings from the third conceptual model of this study showed that for local governments in Indonesia, public sector governance represented by financial transparency, accountability, and internal control are each positively associated with environmental performance. With respect to transparency, when local governments' financial and other relevant information is publicly available and accessible, this facilitates local communities to monitor government decisions and actions for environmental sustainability. With respect to accountability, a key component of sound environmental performance for local government is to ensure that all employees have an adequate understanding of environmental values, risks, regulations, products, and technologies; and that effective monitoring and enforcement or other compliance systems are in place. With respect to internal control, the presence of an effective inspectorate and internal control task force within local government units will facilitate local governments' compliance with the requirements of environmentally sustainable development. Here again, the research found that financial performance acts as a mediating variable in the positive influence of financial transparency, accountability, and internal control on environmental performance. This indicates that the maintenance by local governments of their short- and long-term, budgetary, and service-level solvency will increase their capacity to deliver a strong environmental outcome.

In the last conceptual model, this study investigated the relationships between several key financial management factors and environmental performance. This study found that the three financial management variables investigated – financial flexibility, financial independence, and financial resource management –influence the financial performance of local government directly and influence environmental performance indirectly via financial performance. When local governments have

adequate flexibility and independence to meet revenue and expenditure targets and manage their budgets well, this increases their capacity for programs that support biodiversity conservation and manage the environment sustainably. Together, this study's findings indicate that central and local government authorities should optimise the financial flexibility, financial independence, and financial resource management of local governments (with appropriate monitoring) to improve local government financial performance, as an essential component of effective environmental performance.

References

- Abdallah, Saamah, Juliet Michaelson, S Shah, L Stoll, and N Marks. 2012. *The Happy Planet Index: 2012 Report. A global Index of Sustainable Well-Being.* London.: NewEconomics Foundation. (NEF).
- Abdullah, Syukriy, and Neza Muthia. 2019. "Determinan Saldo Kas Akhir Tahun Pada Pemerintah Daerah di Aceh." *Jurnal Akuntansi Dan Auditing Indonesia* 21 (2):130.
- Abdullahi, Muhammad, and Al Chikaji. 2017. "Local Government and Development: Challenges and Way Forward." *Maiduguri Journal of Arts and Social Sciences* 14 (1):1-20.
- Abdurrohman, Maman, and Soffan Marsus. 2017. "Factor Analysis for Slow Budget Realization." *International Journal of Innovation and Economic Development* 3 (1):28-50.
- Abidin, Nor Zaini Zainal, Kuppusamy Singaravello, and Zaherawati Zakaria. 2015. "Diminishing Obligations of Local Government: Effect on Accountability and Public Trust." *Procedia-Social and Behavioral Sciences* 211:255-259.
- Acharyya, Madhusudan. 2010. "The Role of Operational Risk And Strategic Risk In The Enterprise Risk Management Framework of Financial Services Firms." International Journal of Services Sciences 3 (1):79-102.
- Adams, Carol A, Stephen Muir, and Zahirul Hoque. 2014. "Measurement of Sustainability Performance in The Public Sector." *Sustainability Accounting, Management and Policy Journal.*
- Adams, Charles F. 1981. Urban Fiscal Stress: A Comparative Analysis of 66 U.S. Cities. edited by James M Howell and Charles F Stamm: Academy of Political Science.
- Adams, Dawda, Subhan Ullah, Pervaiz Akhtar, Kweku Adams, and Samir Saidi. 2019. "The Role of Country-Level Institutional Factors in Escaping the Natural Resource Curse: Insights from Ghana." *Resources Policy* 61:433-440.
- Adekola, Omotayo Ajibike, and Segun Wale Olayinka. 2018. "Internal Control And Financial Effectiveness In Tertiary Institutions In Ogun State." 4th ICAN International Academic Conference Proceedings, Ota, Nigeria.
- Adelin, Vani, and Eka Fauzihardani. 2013. "Pengaruh Pengendalian Internal, Ketaaatan pada Aturan Akuntansi dan Kecenderungan Kecurangan Terhadap Perilaku Tidak Etis." *Wahana Riset Akuntansi* 1 (2).
- Adiputra, I Made Pradana, Sidharta Utama, and Hilda Rossieta. 2018. "Transparency of Local Government In Indonesia." *Asian Journal of Accounting Research* 3 (1):123-138.
- Adriana, Anissa, and Irwan Taufiq Ritonga. 2018. "Analysis of Local Financial Management Transparency Based on Websites on Local Government in Java." *Jurnal Dinamika Akuntansi* 10 (1):13-26.
- Afiani, R Khaira, and Yustrida Bernawati. 2019. "Mediating Effect of Financial Performance on Association Between Good Corporate Governance and Firm Value." *KnE Social Sciences*:604–621-604–621.
- Afriyanti, Dwi, Harpanto Guno Sabanu, and Fahrizal Noor. 2015. "Penilaian indeks akuntabilitas instansi pemerintah." *Jurnal Tata Kelola & Akuntabilitas Keuangan Negara* 1 (1):21-42.

- Agler, Robert, and Paul De Boeck. 2017. "On The Interpretation And Use of Mediation: Multiple Perspectives On Mediation Analysis." *Frontiers in psychology* 8:1984.
- Agol, Dorice, Agnieszka E Latawiec, and Bernardo BN Strassburg. 2014. "Evaluating Impacts of Development And Conservation Projects Using Sustainability Indicators: Opportunities And Challenges." *Environmental Impact* Assessment Review 48:1-9.
- Agtini, Magdarina D, Rooswanti Soeharno, Murad Lesmana, Narain H Punjabi, Cyrus Simanjuntak, Ferry Wangsasaputra, Dazwir Nurdin, Sri Pandam Pulungsih, Ainur Rofiq, and Hari Santoso. 2005. "The Burden Of Diarrhoea, Shigellosis, And Cholera In North Jakarta, Indonesia: Findings From 24 Months Surveillance." *BMC Infectious Diseases* 5 (1):89.
- Ahmad, Ehtisham, and Jon Craig. 1997. "Intergovernmental transfers." In *Fiscal Federalism in Theory and Practice*, edited by Teresa Ter-Minassian, 73-107. Washington: Internationla Monetary Fund.
- Ahmed, Khalid, Muhammad Shahbaz, Ahmer Qasim, and Wei Long. 2015. "The Linkages Between Deforestation, Energy and Growth for Environmental Degradation in Pakistan." *Ecological Indicators* 49:95-103.
- Ahyaruddin, Muhammad, and Rusdi Akbar. 2018. "Indonesian Local Government's Accountability and Performance: the Isomorphism Institutional Perspective." *Jurnal Akuntansi dan Investasi* 19 (1):1-11.
- Aikins, Stephen Kwamena. 2011. "An Examination of Government Internal Audits' Role in Improving Financial Performance." *Public Finance and Management* 11 (4):306-337.
- Akbar, Aulia, Johannes Flacke, Javier Martinez, and Martin FAM van Maarseveen. 2020. "Participatory Planning Practice in Rural Indonesia: A Sustainable Development Goals-Based Evaluation." *Community Development*:1-18.
- Akbar, Rusdi. 2011. "Performance Measurement and Accountability in Indonesian Local Government." Curtin University.
- Akbar, Rusdi, Robyn Pilcher, and Brian Perrin. 2012. "Performance measurement in Indonesia: The Case of Local Government." *Pacific Accounting Review* 24 (3):262-291.
- Akbar, Rusdi, and Anthonius H Citra Wijaya. 2013. "The Influence of Information, Organizational Objectives and Targets, and External Pressure Towards the Adoption of Performance Measurement System in Public Sector." *Journal of Indonesian Economy and Business* 28 (1).
- Al-Slehat, Zaher Abdel Fattah. 2019. "The Impact of the Financial Flexibility on the Performance: An Empirical Study on a Sample of Jordanian Services Sector Firms in Period (2010-2017)." International Journal of Business and Management 14 (6):1-11.
- Al-Thuneibat, Ali A., Awad S. Al-Rehaily, and Yousef A. Basodan. 2015. "The Impact of Internal Control Requirements On Profitability of Saudi Shareholding Companies." *International Journal of Commerce and Management* 25 (2):196.
- Al-Tuwaijri, Sulaiman A, Theodore E Christensen, and KE Hughes Ii. 2004. "The relations among environmental disclosure, environmental performance, and economic performance: a simultaneous equations approach." *Accounting, organizations and society* 29 (5-6):447-471.

- Al Breiki, Mariam, and Haitham Nobanee. 2019. "The Role of Financial Management in Promoting Sustainable Business Practices and Development." *Available at SSRN 3472404*.
- Alaerts, GJ. 2020. "Adaptive Policy Implementation: Process and Impact of Indonesia's National Irrigation Reform 1999–2018." *World Development* 129:104880.
- Alam, A B M Mahbub 2019. "Management Control Systems and Financial Conditions: A Study of Municipal Organizations in a Developing Country." School of Business and Governance, Murdoch University.
- Alawattage, Chandana, Trevor Hopper, and Danture Wickramasinghe. 2007. "Introduction to Management Accounting in Less Developed Countries." *Journal of Accounting & Organizational Change* 3 (3):183-191.
- Albrecht, Johan AE. 1998. "Environmental costs and competitiveness. A productspecific test of the porter hypothesis."
- Alcaide Munoz, Laura, Manuel Pedro Rodriguez Bolivar, and Antonio Manuel Lopez Hernandez. 2017. "Transparency in Governments: a Meta-Analytic Review of Incentives for Digital Versus Hard-Copy Public Financial Disclosures." The American Review of Public Administration 47 (5):550-573.
- Alcaraz-Quiles, Francisco José, Andrés Navarro-Galera, and David Ortiz-Rodríguez. 2014. "Factors Influencing The Transparency of Sustainability Information in Regional Governments: An Empirical Study." *Journal of Cleaner Production* 82:179-191.
- Aldridge, C Richard, and Janet L Colbert. 1994. "Management' s Report on Internal Control, and the Accountant' s Response." *Managerial Auditing Journal* 9 (7):21-28.
- Alesina, Alberto, Ricardo Hausmann, Rudolf Hommes, and Ernesto Stein. 1999. "Budget institutions and fiscal performance in Latin America." *Journal of development Economics* 59 (2):253-273.
- Alexander, Anggono, William John Jackson, and Audrey Sadie Paterson. 2014. "An Institutional Theory Perspective On The Impacts Of Central Government's Policies on Local Government: A Case Study Of Indonesian Local Governments." European Accounting Association, 37th Annual Congress, Tallinn, Estonia.
- Alexander, Elizabeth A. 2012. "The Effects of Legal, Normative, and Cultural-Cognitive Institutions on Innovation in Technology Alliances." *Management International Review* 52 (6):791-815.
- Allen, Peter James , Kellie Bennett, and Brody Heritage. 2019. SPSS Statistics: A Practical Guide. 4th ed. South Melbourne, Victoria: Cengage.
- Almeida, Heitor, and Murillo Campello. 2007. "Financial Constraints, Asset Tangibility, and Corporate Investment." *The Review of Financial Studies* 20 (5):1429-1460.
- Alonso, Jose M, and Rhys Andrews. 2019. "Fiscal Decentralisation and Local Government Efficiency: Does Relative Deprivation Matter?" *Environment and Planning C: Politics and Space* 37 (2):360-381.
- Alpenberg, Jan, Tomasz Wnuk-Pel, and Amanda Henebäck. 2018. "Environmental orientation in Swedish local governments." *Sustainability* 10 (2):459.
- Alqooti, Amal Abdulla. 2020. "Public Governance in The Public Sector: Literature Review." *International Journal of Business Ethics and Governance* 3 (3):14-25.

- Alt, James E, and David Dreyer Lassen. 2006. "Fiscal Transparency, Political Parties, and Debt In OECD Countries." *European Economic Review* 50 (6):1403-1439.
- ANAO. 2014. Public Sector Governance Strengthening Performance Through Good Governance. Barton ACT: Commonwealth of Australia.
- Andoko, Effendi. 2019. "Analysis of Indonesia'Government Strategy for Rural Development through Agriculture." *FFTC Agricultural Policy Platform (FFTC-AP)*.
- Andrews, Lorraine, Agnes Higgins, Michael Waring Andrews, and Joan G Lalor. 2012. "Classic Grounded Theory to Analyse Secondary Data: Reality and Reflections." *Grounded Theory Review* 11 (1).
- Andrews, Rhys. 2012. "Social Capital and Public Service Performance: A Review of The Evidence." *Public Policy and Administration* 27 (1):49-67.
- Andriati, Reny. 2019. "Factors Affect Budget Absorption in Government Institutions of Indonesia." *Journal of Governance and Public Policy* 6 (2):101-109.
- Angus, Ian, and Simon Butler. 2011. *Too Many People?: Population, Immigration, and the Environmental Crisis*: Haymarket Books.
- APEC. 2011. Good Practice Guide on Public Sector Governance. Singapore: Asia Pacific Economic Cooperation Secretariat.
- Aras, Güler, and David Crowther. 2008. "Governance and Sustainability: An Investigation into The Relationship between Corporate Governance and Corporate Sustainability." *Management Decision* 46 (3):433-448.
- Araujo, Joaquim Filipe Ferraz Esteves, and Francisca Tejedo-Romero. 2018. "Does Gender Equality Affect Municipal Transparency: The Case Of Spain." *Public Performance & Management Review* 41 (1):69-99.
- Ardigó, Iñaki Albisu, Roberto Martinez, and David Jackson. 2019. "Local Government Accountability Mechanisms." *Transparency International*.
- Arifin, Saru. 2016. "Penyelesaian Sengketa Batas Daerah Menggunakan Pendekatan Regulasi." *Jurnal Hukum IUS QUIA IUSTUM* 23 (3):439-460.
- Arimura, Toshi H, Nicole Darnall, Rama Ganguli, and Hajime Katayama. 2016. "The effect of ISO 14001 on Environmental Performance: Resolving Equivocal Findings." *Journal of Environmental Management* 166: 556-566.
- Armstrong, Anona, Xinting Jia, and Vicky Totikidis. 2005. "Parallels in Private and Public Sector." GovNet Annual Conference: Contemporary Issues in Governance, Victoria, Australia.
- Arshad, Zeeshan, Margarita Robaina, Muhammad Shahbaz, and Anabela Botelho Veloso. 2020. "The Effects of Deforestation and Urbanization on Sustainable Growth in Asian Countries." *Environmental Science and Pollution Research*:1-22.
- Arslan-Ayaydin, Özgür, Chris Florackis, and Aydin Ozkan. 2014. "Financial Flexibility, Corporate Investment and Performance: Evidence From Financial Crises." *Review of Quantitative Finance and Accounting* 42 (2):211-250.
- Asadi, Ali, Morteza Akbari, Hossain Shabanali Fami, Hoshang Iravani, Farahnaz Rostami, and Abolhasan Sadati. 2008. "Poverty Alleviation and Sustainable Development: The Role of Social Capital." *Journal of Social Sciences* 4 (3):202-215.
- Asaolu, Taiwo Olufemi, Samuel Adebayo Adedokun, and James Unam Monday. 2016. "Promoting Good Governance Through Internal Audit Function (IAF): The Nigerian Experience." *International Business Research* 9 (5):196-204.

- Ashworth, Rachel, George Boyne, and Rick Delbridge. 2009. "Escape from the iron cage? Organizational change and isomorphic pressures in the public sector." *Journal of public administration research and theory* 19 (1):165-187.
- Aspinall, Edward, and Greg Fealy. 2010. Soeharto's New Order and Its Legacy: Essays in honour of Harold Crouch: ANU Press.
- Asselin, Lynette. 1995. "Integrated Financial Management in Latin America, as of 1995." *LATPS Occasional Paper World Bank, Series* (18).
- Asteria, Donna, Agus Brotosusilo, and I Wayan Agus Apriana. 2018. "The Role of Women As Environmental Activist in Resolution of Environmental Conflicts for Sustainability of City." *KnE Social Sciences* 3 (10):174-193.
- Atici, U. 2011. "Prediction of The Strength Of Mineral Admixture Concrete Using Multivariable Regression Analysis And An Artificial Neural Network." *Expert Systems With Applications* 38 (8).
- Atuoye, Kilian Nasung, Jenna Dixon, Andrea Rishworth, Sylvester Zackaria Galaa, Sheila A Boamah, and Isaac Luginaah. 2015. "Can She Make It? Transportation Barriers to Accessing Maternal and Child Health Care Services in Rural Ghana." *BMC Health Services Research* 15 (1):1-10.
- Awwaliyah, Noor Farieda, Ratno Agriyanto, and Dessy Noor Farida. 2019. "The Effect of Regional Original Income and Balance Funding on Regional Government Financial Performance." *Journal of Islamic Accounting and Finance Research–Vol* 1 (1).
- Ayagre, Philip, and Julius Aidoo-Buameh. 2017. "Ethical Values, Integrity and Internal Controls in Public Sector Organisations: A Developing Country Perspective." *Management & Accounting Review (MAR)* 16 (2):41-58.
- Aye, Goodness C, and Prosper Ebruvwiyo Edoja. 2017. "Effect of Economic Growth on CO2 Emission in Developing Countries: Evidence from a Dynamic Panel Threshold Model." *Cogent Economics & Finance* 5 (1):1379239.
- Badara, Mu'azu Saidu, and Siti Zabedah Saidin. 2013. "Impact of the effective internal control system on the internal audit effectiveness at local government level." *Journal of Social and Development Sciences* 4 (1):16-23.
- Bahl, Roy 1984. *Financing State and Local Government in the 1980's*. New York: Oxford University Press.
- Bahrun, Andi, and Dedi Erawan. 2015. "Perwujudan Swasembada Pangan Melalui Perbaikan Teknologi Pengelolaan Air." Prosiding Seminar Nasional Agribisnis.
- Bailey, Stephen James. 1991. "Fiscal Stress: The New System Of Local Government Finance In England." *Urban Studies* 28 (6):889-907.
- Baiocchi, Gianpaolo. 2001. "Participation, Activism, and Politics: The Porto Alegre Experiment and Deliberative Democratic Theory." *Politics & Society* 29 (1):43-72.
- Baker, Marzena, Muhammad Ali, and Erica French. 2019. "The Impact of Women's Representation on Performance in Project-Based and Non-Project-Based Organizations." *International Journal of Project Management* 37 (7):872-883.
- Bakker, Karen. 2007. "Trickle Down? Private Sector Participation And The Pro-Poor Water Supply Debate In Jakarta, Indonesia." *Geoforum* 38 (5):855-868.
- Ball, Amanda, and Jan Bebbington. 2008. "Accounting and Reporting for Sustainable Development in Public Service Organizations." *Public Money and Management* 28 (6):323-326.
- Ball, Amanda, and Suzana Grubnic. 2011. "Sustainability Accounting And Accountability In The Public Sector." In Sustainability accounting and

accountability, edited by Jeffrey Unerman, Jan Bebbington and Brendan O'Dwyer, 176. New York: Routledge.

- Baloyi, Wiston Mbhazima, and Lourens Johannes Erasmus Beyers. 2019. "The Impact of Financial Malpractice on Service Delivery: A Conceptual Examination of Literature." The 4th Annual International Conference on Public Administration and Development Alternatives Johannesburg, South Africa.
- Baltaci, Mustafa, and Serdar Yilmaz. 2006. "Keeping An Eye on Subnational Governments: Internal Control and Audit at Local Levels." *World Bank Institute, Washington, DC.*
- Bancel, Franck, and Usha R Mittoo. 2011a. "Financial Flexibility and The Impact of The Global Financial Crisis: Evidence From France." *International Journal of Managerial Finance*.
- Bancel, Franck, and Usha R Mittoo. 2011b. "Financial Flexibility And The Impact of The Global Financial Crisis: Evidence From France." *International Journal of Managerial Finance* 7 (2):179-216.
- BAPPENAS, and UNDP. 2008. *Studi Evaluasi Dampak Pemekaran Daerah*. Jakarta: BRIDGE.
- Barbera, Anthony J, and Virginia D McConnell. 1990. "The impact of environmental regulations on industry productivity: direct and indirect effects." *Journal of environmental economics and management* 18 (1):50-65.
- Batafor, Gregorius Gehi. 2011. "Evaluasi Kinerja Keuangan dan Tingkat Kesejahteraan Masyarakat Kabupaten Lembata-Provinsi NTT." *Program PascaSarjana: Universitas Udayana*.
- Bauducco, Sofía, Aleš Bulíř, and Martin Čihák. 2011. "Monetary policy rules with financial instability." *Finance a Uver Czech Journal of Economics and Finance* 61 (6):545-565.
- Bauhr, Monika, and Ruth Carlitz. 2020. "When Does Transparency Improve Public Services? Street-level Discretion, Information, and Targeting." *Public Administration*:1–17.
- Bekker, Simon , and Anthea Jeffery. 1989. "Local Government In Urban South Africa: A Normative Approach." *Local Government Working Paper* (1).
- Belal, Ataur Rahman. 2015. "Social and Environmental Accountability in Developing Countries." In *Development Oriented CSR* edited by Dima Jamali, C Karam and M Blowfield, 153-166. Sheffield: Green Leaf.
- Bell, Simon, and Stephen Morse. 2012. Sustainability Indicators: Measuring The Immeasurable?: Routledge.
- Benedek, Mária, Klára Tubak Szenténé, and Daniel Beres. 2014. "Internal Controls in Local Governments." *Public Finance Quarterly* 59 (3):296-309.
- Benito, Bernardino, and Francisco Bastida. 2009. "Budget Transparency, Fiscal Performance, and Political Turnout: An International Approach." *Public Administration Review* 69 (3):403-417.
- Berger, Peter L., and Thomas Luckmann. 1979. *The Social Construction of Reality : A Treatise in The Sociology of Knowledge*. Harmondsworth, Eng: Penguin Books.
- Bernauer, Thomas, and Vally Koubi. 2009. "Effects of political institutions on air quality." *Ecological economics* 68 (5):1355-1365.
- Bernick, E Lee, Jonathan M Birds, Katheryn Brekken, Al G Gourrier, and Leander D Kellogg. 2014. "Explaining County Government Fiscal Transparency In An Age Of E-Government." *State and Local Government Review* 46 (3):173-183.

- Berrone, Pascual, Jordi Surroca, and Josep A Tribó. 2007. "Corporate ethical identity as a determinant of firm performance: A test of the mediating role of stakeholder satisfaction." *Journal of Business Ethics* 76 (1):35-53.
- Beschel, Robert P, Blair James Cameron, Jana Kunicova, and C Bernard Myers. 2018. Improving Public Sector Performance: Through Innovation and Inter-Agency Coordination. Malaysia.
- Bhuiyan, Shahjahan H, and Francis Amagoh. 2011. "Public Sector Reform in Kazakhstan: Issues and Perspectives." International Journal of Public Sector Management.
- Bilyay-Erdogan, Seda. 2020. "Does Financial Flexibility Enhance Firm Value? A Comparative Study Between Developed and Emerging Countries." *Business: Theory and Practice* 21 (2):723-736.
- Birčiaková, Naďa, Jana Stávková, and Veronika Antošová. 2015. "Evaluating Living Standard Indicators." *DANUBE: Law and Economics Review* 6 (3):175-188.
- Bird, Richard M, and Andrey V Tarasov. 2004. "Closing The Gap: Fiscal Imbalances And Intergovernmental Transfers In Developed Federations." *Environment and Planning C: Government and Policy* 22 (1):77-102.
- Bjorkman, Martina, and Jakob Svensson. 2007. Power to the people: evidence from a randomized field experiment of a community-based monitoring project in Uganda. The World Bank.
- Bland, J Martin, and Douglas G Altman. 1997. "Statistics Notes: Cronbach's Alpha." BMJ 314 (7080):572.
- Boetti, Lorenzo, Massimiliano Piacenza, and Gilberto Turati. 2012. "Decentralization and Local Governments' Performance: How Does Fiscal Autonomy Affect Spending Efficiency?" *FinanzArchiv/Public Finance Analysis*:269-302.
- Bonett, Douglas G, and Thomas A Wright. 2015. "Cronbach's Alpha Reliability: Interval Estimation, Hypothesis Testing, and Sample Size Planning." *Journal* of organizational behavior 36 (1):3-15.
- Booth, Anne. 1977. "Irrigation in Indonesia Part II." *Bulletin of Indonesian Economic Studies* 13 (2):45-77.
- Booysen, Frederik. 2002. "An Overview And Evaluation of Composite Indices of Development." *Social Indicators Research* 59 (2):115-151.
- Bostan, Ionel. 2010. "Performance of Financial Control." *Theoretical and Applied Economics* XVII (7).
- Bovens, Mark. 2007a. "Analysing and Assessing Accountability: A Conceptual Framework." *European Law Journal* 13 (4):447-468.
- Bovens, Mark. 2007b. "Public Accountability." In *The Oxford Handbook of Public Management*, edited by Ewan Ferlie, Laurence E Lynn Jr and Christopher Pollitt. Oxford, United Kingdom: Oxford University Press.
- Bowman, Marilyn L. 2002. "The Perfidy of Percentiles." Archives of Clinical Neuropsychology 17 (3):295-303.
- Boyne, George A. 2003. "Sources of Public Service Improvement: A Critical Review and Research Agenda." *Journal of Public Administration Research and Theory* 13 (3):367-394.
- Boyne, George, and Julian Gould-Williams. 2003. "Planning and Performance in Public Organizations an Empirical Analysis." *Public Management Review* 5 (1):115-132.
- BPS. 2018. Statistical Yearbook of Indonesia 2018. Jakarta: CV. Dharmaputra.
- Brännlund, Runar, Rolf Färe, and Shawna Grosskopf. 1995. "Environmental Regulation and Profitability: An Application to Swedish Pulp and Paper Mills." *Environmental and Resource Economics* 6 (1):23-36.
- Brignall, Stan, and Sven Modell. 2000. "An Institutional Perspective on Performance Measurement and Management in The 'New Public Sector'." *Management accounting research* 11 (3):281-306.
- Brinkerhoff, Derick W, and Jennifer M Brinkerhoff. 2015. "Public Sector Management Reform in Developing Countries: Perspectives Beyond NPM Orthodoxy." *Public Administration and Development* 35 (4):222-237.
- Broadbent, Jane, and Richard Laughlin. 2003. "Control and Legitimation In Government Accountability Processes: The Private Finance Initiative In The UK." *Critical Perspectives on Accounting* 14 (1):23-48.
- Brodjonegoro, Bambang. 2018. "Fiscal Decentralization in Indonesia." In *Governance in Indonesia: Challenges Facing the Megawati Presidency*, edited by Hadi Soesastro, Anthony L Smith and Hang Mui Lin. Singapore: Institute of Southeast Asian Studies.
- Brown, Ken W. 1993. "The 10-Point Test Of Financial Condition: Toward An Easy-To-Use Assessment Tool For Smaller Cities." *Government Finance Review* 9:21-21.
- Brown, Lawrence D., and Marcus L. Caylor. 2006. "Corporate governance and firm valuation." *Journal of Accounting and Public Policy* 25 (4):409-434.
- Brown, Nerissa C, C Pott, and Andreas Wömpener. 2008. "The Effect of Internal Control Regulation on Earnings Quality: Evidence From Germany." *University* of Southern California, Los Angeles, CA.
- Bruns, Bryan. 2004. "From Voice to Empowerment: Rerouting Irrigation Reform in Indonesia." *The Politics of Irrigation Reform: Contested Policy Formulation and Implementation in Asia, Africa, and Latin America. Hans, UK: Ashgate.*
- Budijanto, Didik. 2018. *Data Dasar Puskesmas Kondisi Desember 2016*. Jakarta: Pusat Data dan Informasi Kemenkes RI.
- Burck, Jan, Christoph Bals, and Simone Ackermann. 2009. *The Climate Change Performance Index: Background And Methodology*: Germanwatch.
- Burkholder, Thomas J, and Richard L Lieber. 1996. "Stepwise regression is an alternative to splines for fitting noisy data." *Journal of Biomechanics* 29 (2):235-238.
- Burritt, Roger L, and Stephen Welch. 1997. "Accountability for Environmental Performance of the Australian Commonwealth Public Sector." *Accounting, Auditing & Accountability Journal* 10 (4):532-561.
- Busch, Jonah, Kalifi Ferretti-Gallon, Jens Engelmann, Max Wright, Kemen G Austin, Fred Stolle, Svetlana Turubanova, Peter V Potapov, Belinda Margono, and Matthew C Hansen. 2015. "Reductions in Emissions from Deforestation from Indonesia's Moratorium on New Oil Palm, Timber, and Logging Concessions." *Proceedings of the National Academy of Sciences* 112 (5):1328-1333.
- Butarbutar, Iwan Richard, and Irfan Sofi. 2017. "Efektivitas Hibah Air Minum dari Pendapatan Dalam Negeri dalam Mencapai Target Sambungan Rumah Tangga." *Jurnal Defis* 1 (1):48-58.
- Butt, Babar Zaheer, Ahmed Imran Hunjra, and Kashif Ur Rehman. 2010. "Financial Management Practices and Their Impact on Organizational Performance." *World Applied Sciences Journal* 9 (9):997-1002.

- Caamano-Alegre, J., S. Lago-Penas, F. Reyes-Santias, and A. Santiago-Boubeta. 2013. "Budget Transparency in Local Governments: An Empirical Analysis." *Local Government Studies* 39 (2):182-207.
- Cabaleiro, Roberto, Enrique Buch, and Antonio Vaamonde. 2013. "Developing A Method To Assessing The Municipal Financial Health." *The American Review* of Public Administration 43 (6):729-751.
- Camagni, Roberto. 2017. "Sustainable Urban Development: Definition and Reasons for a Research Programme." In *Seminal Studies in Regional and Urban Economics*, 261-282. Springer.
- Cameron, A Colin, and Frank AG Windmeijer. 1997. "An R-Squared Measure Of Goodness Of Fit For Some Common Nonlinear Regression Models." *Journal* of Econometrics 77 (2):329-342.
- Cameron, Lisa, Manisha Shah, and Susan Olivia. 2013. "Impact Evaluation of A Large-Scale Rural Sanitation Project in Indonesia." *Policy Research Working Paper* 6360 (83).
- Campello, Murillo, John R Graham, and Campbell R Harvey. 2010. "The Real Effects of Financial Constraints: Evidence From A Financial Crisis." *Journal of Financial Economics* 97 (3):470-487.
- Cantarero, D., and Marta Pascual. 2008. "Analysing the Impact of Fiscal Decentralization on Health Outcomes: Empirical Evidence from Spain." *Applied Economics Letters* 15 (2):109-111.
- Carlson, Mark A, Thomas King, and Kurt Lewis. 2011. "Distress in the financial sector and economic activity." *The BE Journal of Economic Analysis & Policy* 11 (1).
- Carmeli, Abraham. 2008. "The Fiscal Distress of Local Governments In Israel: Sources And Coping Strategies." *Administration & Society* 39 (8):984-1007.
- Caron, Pier-Olivier. 2019. "A comparison of the type I error rates of three assessment methods for indirect effects." *Journal of Statistical Computation and Simulation* 89 (8):1343-1356.
- Carpenter, Vivian L, and Ehsan H Feroz. 2001. "Institutional Theory and Accounting Rule Choice: An Analysis of Four Us State Governments' Decisions to Adopt Generally Accepted Accounting Principles." *Accounting, organizations and society* 26 (7-8):565-596.
- Casal, Roberto Cabaleiro, and Enrique Buch Gómez. 2011. "Impact of Size and Geographic Location on the Financial Condition of Spanish Municipalities." *Transylvanian Review of Administrative Sciences* 7 (34):22-39.
- Cavalluzzo, Ken S, and Christopher D Ittner. 2004. "Implementing Performance Measurement Innovations: Evidence From Government." *Accounting, Organizations and Society* 29 (3-4):243-267.
- Cerrillo-i-Martínez, Agustí. 2017. "The Principles of Good Governance." In *Global Encyclopedia of Public Administration, Public Policy, and Governance*, edited by A Farazmand. Cham: Springer.
- Chaney, Barbara A, Dean Michael Mead, and Kenneth R Schermann. 2002. "The New Governmental Financial Reporting Model: What It Means For Analyzing Government Financial Condition." *The Journal of Government Financial Management* 51 (1):26.
- Channar, Zahid Ali, Maha Khan, and Irfan Haider Shakri. 2015. "Internal Control Effectiveness & Its Relationship With Financial Performance." *IBT Journal of Business Studies* 11 (2):92-107.

Chapman, Jeffrey I. 1999. Local Government, Fiscal Autonomy And Fiscal Stress: The Case Of California: Lincoln Institute of Land Policy Cambridge.

Chatterjee, Samprit, and Ali S Hadi. 2015. *Regression Analysis By Example*: John Wiley & Sons.

Chatterjee, Samprit, and Jeffrey S Simonoff. 2013. *Handbook of Regression Analysis*. Vol. 5: John Wiley & Sons.

Chattopadhyay, Soumyadip. 2013. "Decentralised Provision of Public Services in Developing Countries: A Review of Theoretical Discourses and Empirical Evidence." *Social Change* 43 (3):421-441.

Cheema, G.S. 2007. "Devolution with Accountability: Learning from Good Practices." In *Decentralizing Governance: Emerging Concepts and Practices*, edited by G.S Cheema and D.A Rondinelli. Washington: Brookings Institution Press.

Chen, Xia, and Jianmin Liu. 2020. "Fiscal Decentralization and Environmental Pollution: A Spatial Analysis." *Discrete Dynamics in Nature and Society* 2020.

Cheng, Joseph LC, and Idalene F Kesner. 1997. "Organizational Slack and Response to Environmental Shifts: the Impact of Resource Allocation Patterns." *Journal of management* 23 (1):1-18.

Chenhall, Robert H. 2003. "Management Control Systems Design Within Its Organizational Context: Findings From Contingency-Based Research And Directions For The Future." *Accounting, Organizations and Society* 28 (2):127-168.

Chenhall, Robert H., and Kim Langfield-Smith. 2007. "Multiple Perspectives of Performance Measures." *European Management Journal* 25 (4):266-282.

Cheung, Gordon W., and Rebecca S. Lau. 2008. "Testing Mediation and Suppression Effects of Latent Variables: Bootstrapping With Structural Equation Models." *Organizational Research Methods* 11 (2):296-325.

Chigwende, Shylet, Abel Mukuze, and Stephen Tawanda Musasa. 2013. "The Impact of Transparency and Accountability on Service Delivery: An Analysis of Municipalities of Zimbabwe." *International Journal of Governance* 4 (1):26-38.

Chigwende, Shylet, Abel Mukuze, and Stephen Tawanda Musasa. 2014. "The Impact of Transparency and Accountability on Service Delivery: An Analysis of Municipalities of Zimbabwe." *International Journal of Governance* 4 (1).

Chinander, Karen R. 2001. "Aligning Accountability and Awareness for Environmental Performance in Operations." *Production and Operations Management* 10 (3):276-291.

Chiu, Yi-Chia, and Yi-Ching Liaw. 2009. "Organizational Slack: Is More or Less Better?" *Journal of Organizational Change Management* 22 (3):321-342.

Choi, NakHyeok. 2021. "Analyzing Local Government Capacity and Performance: Implications for Sustainable Development." *Sustainability* 13 (7):1-14.

Chokkalingam, Unna. 2006. One century of Forest Rehabilitation In The Philippines: CIFOR.

Chong, Y, Kumi Abeysuriya, Lenny Hidayat, Hery Sulistio, and JR Willetts. 2016. "Strengthening Local Governance Arrangements For Sanitation: Case Studies of Small Cities in Indonesia." *Aquatic Procedia*.

Christensen, Tom, and Per Lægreid. 2015. "Performance and Accountability—A Theoretical Discussion and an Empirical Assessment." *Public Organization Review* 15 (2):207-225.

Chudi, Ibekwe Perpetus. 2010. "Healthcare Problems in Developing Countries." *Medical Practice and Reviews* 1 (1):9-11.

- Church, Russell M. 2002. "The Effective Use Of Secondary Data." *Learning and motivation* 33 (1):32-45.
- Chye Koh, Hian, and E. Sah Woo. 1998. "The Expectation Gap in Auditing." *Managerial Auditing Journal* 13 (3):147-154.
- Cigu, Elena. 2014. "An Aproach of Local Financial Autonomy and Implication Over Sustainable Development in the Knowledge Society." *Journal of Public Administration, Finance and Law* (06):44-53.
- Cohen, Jacob, Patricia Cohen, Stephen G West, and Leona S Aiken. 2013. *Applied Multiple Regression/Correlation Analysis For The Behavioral Sciences*. Mahwah, N.J: Routledge.
- Cole, Matthew A, and Eric Neumayer. 2004. "Examining the impact of demographic factors on air pollution." *Population and Environment* 26 (1):5-21.
- Collins, Oyoo Otieno. 2014. "Effect of Internal Control on Financial Performance of Micro-Finance Institutions in Kisumu Central Constituency, Kenya." *Journal of Scientific Research and Essay* 3:139-155.
- Coram, Paul, Colin Ferguson, and Robyn Moroney. 2006. "The value of internal audit in fraud detection." *Journal of Accounting and Finance* 48 (4):543-59.
- Corfee-Morlot, Jan, Lamia Kamal-Chaoui, Michael G Donovan, Ian Cochran, Alexis Robert, and Pierre-Jonathan Teasdale. 2009. "Cities, Climate Change and Multilevel Governance." *OECD Environmental Working Papers* 14.
- COSO. 2013. Internal Control Integrated Framework. Durham: Committee of Sponsoring Organizations of the Treadway Commission.
- Coste, Andreea-Ioana, and Adriana Tiron Tudor. 2013. "Service Performance-Between Measurement and Information in The Public Sector." *Procedia-Social and Behavioral Sciences* 92:215-219.
- Cowton, Christopher J. 1998. "The Use Of Secondary Data In Business Ethics Research." *Journal of Business Ethics* 17 (4):423-434.
- Coy, David, Mary Fischer, and Teresa Gordon. 2001. "Public Accountability: A New Paradigm For College And University Annual Reports." *Critical Perspectives on Accounting* 12 (1):1-31.
- Cramer, James C, and Robin P Cheney. 2000. "Lost in the ozone: population growth and ozone in California." *Population and Environment* 21 (3):315-338.
- Creel, Jérôme, Paul Hubert, and Fabien Labondance. 2015. "Financial stability and economic performance." *Economic Modelling* 48:25-40.
- Da Cruz, Nuno Ferreira, and Rui Cunha Marques. 2014. "Revisiting the Determinants of Local Government Performance." *Omega* 44:91-103.
- Damania, Richard. 2002. "Environmental Controls With Corrupt Bureaucrats." *Environment and Development Economics*:407-427.
- Dandara-Tabacaru, Daniela, and Marius Ioan Danila. 2017. "Financial System Stability and Social Welfare." *Revista de Cercetare si Interventie Sociala* 56 (MARCH):153-168.
- Dani, Andreia Carpes, Jaime Dagostim Picolo, and Roberto Carlos Klann. 2019. "Gender Influence, Social Responsibility and Governance in Performance." *RAUSP Management Journal* 54 (2):154-177.
- Darmawan, Rachmad Erland Danny. 2008. "The Practices of Decentralization In Indonesia And Its Implication On Local Competitiveness." University of Twente.
- Darnall, Nicole, and Younsung Kim. 2012. "Which types of environmental management systems are related to greater environmental improvements?" *Public Administration Review* 72 (3):351-365.

- Darweesh, Mohamed. 2015. "Correlations Between Corporate Governance, Financial Performance, and Market Value." Dissertation/Thesis, ProQuest Dissertations Publishing.
- Das, Keya Rani, and AHMR Imon. 2016. "A Brief Review of Tests For Normality." American Journal of Theoretical and Applied Statistics 5 (1):5-12.
- Daud, Sitti Bulkis. 2006. "Community Empowerment and Irrigation Management: A Case of Water Users Association in South Sulawesi, Indonesia." *Potential of Social Capital for Community Development*:41-66.
- de Jong, Wil. 2010. "Forest Rehabilitation And Its Implication For Forest Transition Theory." *Biotropica* 42 (1):3-9.
- de Lima, Ismar Borges, and Leszek Buszynski. 2011. "Local Environmental Governance, Public Policies and Deforestation in Amazonia." *Management of Environmental Quality: An International Journal* 22 (3).
- De Sherbinin, Alex, David Carr, Susan Cassels, and Leiwen Jiang. 2007. "Population and Environment." *Annual Review of Environment and Resources* 32 (1):345-373.
- de Silva, Keshara, Prem W Senarath Yapa, and Gillian Vesty. 2020. "The Impact of Accountability Mechanisms on Public Sector Environmental Sustainability Performance: A Case Study of Sri Lanka." *Australasian Accounting, Business and Finance Journal* 14 (3):38-55.
- Deegan, Craig Michael. 2009. *Financial accounting theory*. 3rd ed. North Ryde, N.S.W: McGraw-Hill.
- DeFries, Ruth, Frédéric Achard, Sandra Brown, Martin Herold, Daniel Murdiyarso, Bernhard Schlamadinger, and Carlos de Souza Jr. 2007. "Earth Observations for Estimating Greenhouse Gas Emissions from Deforestation in Developing Countries." *Environmental science & policy* 10 (4):385-394.
- Del Sol, Daniel Albalatel. 2013. "The Institutional, Economic and Social Determinants of Local Government Transparency." *Journal of Economic Policy Reform* 16 (1):90-107.
- DEPIKD. 2018. *Ringkasan APBD 2018*. Jakarta: Direktorat Jenderal Perimbangan Keuangan.
- Dermawan, Ahmad, Heru Komarudin, and Sian McGrath. 2006. "Decentralization in Indonesia's Forestry Sector–Is It Over? What Comes Next." Eleventh Biennial Global Conference of the International Association for the Study of Common Property (IASCP) on the theme 'Survival of the Commons: Mounting Challenges and New Realities', Bali.
- Devas, Nick, and Ursula Grant. 2003. "Local Government Decision-Making—Citizen Participation and Local Accountability: Some Evidence From Kenya and Uganda." *Public Administration and Development* 23 (4):307-316.
- Dewata, Evada, Hadi Jauhari, Siska Aprianti, and Eka Nurfa Hijria. 2018. "The Effects of Local Government Characteristics and Audit Opinion on the Performance of District and City Governments in Indonesia." *Jurnal Dinamika Akuntansi dan Bisnis Vol* 5 (2):151-162.
- Dharmawardena, Priyadarshana, Raphel Ouseph Thattil, and Sembakutti Samita. 2017. "Scale Independent Principal Component Analysis and Factor Analysis with Preserved Inherent Variability of the Indicators." *American Journal of Theoretical and Applied Statistics* 6 (2):90-94.
- Dick-Sagoe, Christopher. 2020. "Decentralization for Improving the Provision of Public Services in Developing Countries: A Critical Review." Cogent Economics & Finance 8 (1):1804036.

- Digdowiseiso, Kumba, Eko Sugiyanto, and Djumadin Zainul. 2018. "Implementation of Irrigation Policy in the Decentralized Government: A Case Study of West Java, Indonesia." *Journal of Environmental Management and Tourism* 9 (3 (27)):411-422.
- DiMaggio, P. J, and W. W Powell. 1983. "The Iron Cage Revisited: Institutional Isomorphism and Collective Rationality in Organizational Field." *American Sociological Review* 48:147-160.
- DiMaggio, P.J, and W Powell. 1991. "The Iron Cage Revisited: Institutional Isomorphism and Collective Rationality in Organisational Fields." In *The New Institutionalism in Organizational Analysis*, edited by W.W Powell and P.J DiMaggio. Chicago: University of Chicago Press.
- Dionne, Georges. 2013. "Risk Management: History, Definition, And Critique." *Risk Management and Insurance Review* 16 (2):147-166.

Directorate General of Fiscal Balance. 2017. Dana Bagi Hasil (DBH) Pajak.

- Dizdaroglu, Didem. 2015. "Developing Micro-Level Urban Ecosystem Indicators For Sustainability Assessment." *Environmental Impact Assessment Review* 54:119-124.
- DJPK. 2019. *Ringkasan APBD 2016.* Jakarta: Direktorat Jenderal Perimbangan Keuangan.
- Dollery, Brian, Lin Crase, and Joel Byrnes. 2006. "Local Government Failure: Why Does Australian Local Government Experience Permanent Financial Austerity?" *Australian Journal of Political Science* 41 (3):339-353.
- Donelson, D. C., M. S. Ege, and J. M. McInnis. 2017. "Internal Control Weaknesses and Financial Reporting Fraud." *AUDITING-A JOURNAL OF PRACTICE & THEORY* 36 (3):45-69.
- Doolan, Daniel M., Daniel M. Doolan, Erika S. Froelicher, and Erika S. Froelicher. 2009. "Using An Existing Data Set To Answer New Research Questions: A Methodological Review." *Research and theory for nursing practice* 23 (3):203-215.
- Dorożyński, Tomasz, Agnieszka Dorożyńska, and Wojciech Urbaniak. 2014. "The Role of Local Government Units in Attracting FDI: the Case of the Lodz Region." *Business and Economic Horizons* 10 (4):281-304.
- Drezner, Zvi, Ofir Turel, and Dawit Zerom. 2010. "A Modified Kolmogorov-Smirnov Test for Normality." *Communications in Statistics - Simulation and Computation* 39 (4):693-704.
- Du Plessis, Anel. 2015. "A Role For Local Government in Global Environmental Governance and Transnational Environmental Law from A Subsidiarity Perspective." *Comparative and International Law Journal of Southern Africa* 48 (2):281-316.
- Dubbink, Wim, Johan Graafland, and Luc Van Liedekerke. 2008. "CSR, Transparency and the Role of Intermediate Organisations." *Journal of business ethics* 82 (2):391-406.
- Dubnick, Melvin. 2005. "Accountability and the Promise of Performance: in Search of the Mechanisms." *Public Performance & Management Review* 28 (3):376-417.
- Dyllick, Thomas, and Kai Hockerts. 2002. "Beyond the Business Case for Corporate Sustainability." *Business strategy and the environment* 11 (2):130-141.
- Earnhart, Dietrich, and Lubomir Lizal. 2006. "Effects of Ownership and Financial Performance on Corporate Environmental Performance." *Journal of Comparative Economics* 34 (1):111.

- Edvardsson, Bo, Anders Gustafsson, and Inger Roos. 2005. "Service Portraits in Service Research: a Critical Review." *International journal of service industry management*.
- Edwards, Jeffrey R, and Lisa Schurer Lambert. 2007. "Methods For Integrating Moderation And Mediation: A General Analytical Framework Using Moderated Path Analysis." *Psychological methods* 12 (1):1.
- Eisenhardt, K. M. 1988. "Agency and Institutional Theory Explanations: The Case of Retail Sales Compensation." *Academy of Management Journal* 31 (3):488-511.
- Eldridge, Stephen, Jos van Iwaarden, Ton van der Wiele, and Roger Williams. 2014. "Management Control Systems For Business Processes In Uncertain Environments." *International Journal of Quality & Reliability Management*.
- Elkington, John. 1998. "Partnerships From Cannibals With Forks: The Triple Bottom Line of 21st-Century Business." *Environmental Quality Management* 8 (1):37-51.
- Embaye, Semere Simon, and Fessehaie Haile. 2019. "The Effect of Financial Flexibility on Firm Performance-Evidence From North African Countries." *European Journal of Research* 5.
- Engel, Susan, and Anggun Susilo. 2014. "Shaming and Sanitation in Indonesia: A Return To Colonial Public Health Practices?" *Development and Change* 45 (1):157-178.
- Enofe, A.O, and N.F Uwaifo. 2016. "The Impact of Internal Control on Corporate Sustainability and Economic Development." *Journal of Policy and Development Studies* 10 (4).
- Erlina, A Saputra, and Iskandar Muda. 2017. "The Analysis of the Influencing Factors of Budget Absorption." *International Journal of Economic Research* 14 (12):287-300.
- Esty, Daniel C, and Michael E Porter. 2005. "National Environmental Performance: An Empirical Analysis of Policy Results and Determinants." *Environment and development economics*:391-434.
- Everard, Mark, Paul Johnston, David Santillo, and Chad Staddon. 2020. "The Role of Ecosystems in Mitigation and Management of Covid-19 and Other Zoonoses." *Environmental Science & Policy* 111:7-17.
- Eyraud, Luc, and Lusine Lusinyan. 2013. "Vertical Fiscal Imbalances And Fiscal Performance In Advanced Economies." *Journal of Monetary Economics* 60 (5):571-587.
- Eysink, Wim , and Leen Paape. 2016. Good Governance Driving Corporate Performance? A Meta-Analysis of Academic Research & Invitation to Engage in The Dialogue. Netherland: Deloitte.
- Fafurida, Fafurida, and Erliz Nindi Pratiwi. 2017. "Financial Independence of Regencies and Cities in Central Java." *Economic Journal of Emerging Markets* 9:199-209.
- Fajar, Nurul, and Muhammad Arfan. 2017. "Analisis Penyerapan Anggaran Pada Satuan Kerja Perangkat Aceh." *Jurnal Telaah dan Riset Akuntansi* 10 (2):95-102.
- Farrugia, Nadia. 2007. "Conceptual Issues In Constructing Composite Indices." International Conference On Small States And Economic Resilience Valletta, Malta
- Fatimah, Fatimah, Ronny Malavia Mardani, and Budi Wahono. 2019. "Pengaruh Good Corporate Governance Terhadap Nilai Perusahaan Dengan Kinerja

Keuangan Sebagai Variabel Intervening (Studi Kasus Pada Perusahaan Manufaktur Sektor Barang)." *Jurnal Ilmiah Riset Manajemen* 8 (15).

- Fernandes, Riyan 2019. "Jurnalisme Warga Dan Gerakan Sadar Berbahasa Indonesia Di Ruang Publik." In *Pengutamaan Bahasa Negara di Ruang Publik: Perkuat Pengawasan*. Jakarta: Pusat Pembinaan Bahasa Dan Sastra Badan Pengembangan Bahasa Dan Perbukuan.
- Field, Andy. 2013. *Discovering Statistics Using IBM SPSS Statistics*. Thousand Oaks (CA): Sage.
- Field, Andy P. 2009. *Discovering Statistics Using SPSS: And Sex And Drugs And Rock 'N' Roll*. 3rd, [Students] ed. London: SAGE.
- Florio, Cristina, and Giulia Leoni. 2017. "Enterprise risk management and firm performance: The Italian case." *The British Accounting Review* 49 (1):56-74.
- Fogarty, Timothy J. 1996. "The Imagery and Reality of Peer Review in the U.S.: Insights from Institutional Theory." *Accounting, Organizations and Society* 21 (2):243-267.
- Fölscher, Alta. 2006. "Introduction: African Experience with Budget Reform." OECD Journal on Budgeting 6 (2):1.
- Fölscher, Alta, Warren Krafchik, and Isaac Shapiro. 2000. "Transparency and Participation in the Budget Process." *IDASA, Budget Information Service and the International Budget Project.*
- Freudenberg, Michael. 2003. "Composite Indicators Of Country Performance: A Critical Assessment." *OECD Science, Technology and Industry Working Papers* 16.
- Fricker, Alan. 1998. "Measuring Up To Sustainability." Futures 30 (4):367-375.
- Fujii, Hidemichi, and Shunsuke Managi. 2016. "Trends in Corporate Environmental Management Studies and Databases." *Environmental Economics and Policy Studies* 18 (2):265-272.
- Fulazzaky, Mohamad Ali. 2017. "Participation of Farmers in Irrigation Water Management in Indonesia: A Review." *Irrigation and Drainage* 66 (2):182-191.
- Galera, Andrés Navarro, Araceli de los Ríos Berjillos, Mercedes Ruiz Lozano, and Pilar Tirado Valencia. 2014. "Transparency of Sustainability Information in Local Governments: English-Speaking and Nordic Cross-Country Analysis." *Journal of Cleaner Production* 64:495-504.
- Gamage, CT, Kevin Low Lock, and AAJ Fernando. 2014. "A proposed research framework: Effectiveness of internal control system in state commercial banks in Sri Lanka." *International Journal of Scientific Research and Innovative Technology* 1 (5):25-44.
- Gamayuni, Rindu Rika. 2018. "The Effect of Internal Audit Function Effectiveness and Implementation of Accrual Based Government Accounting Standard on Financial Reporting Quality." *Review of Integrative Business and Economics Research* 7 (1):46-58.
- Gamayuni, Rindu Rika. 2019. "The Effect of Management Accounting Information System Application on Information Quality and Its Implication on Good Government Governance." *International Journal Of Scientific & Technology Research* 8 (8):1247-1254.
- Gamba, Andrea, and Alexander Triantis. 2008. "The Value of Financial Flexibility." *The journal of finance* 63 (5):2263-2296.

- García-Sánchez, Isabel-María, José-Valeriano Frías-Aceituno, and Luis Rodríguez-Domínguez. 2013. "Determinants of Corporate Social Disclosure in Spanish Local Governments." *Journal of Cleaner Production* 39:60-72.
- Garcia-Tabuyo, Manoli, Alejandro Sáez-Martín, and María Del Carmen Caba-Pérez. 2016. "Mandatory Versus Voluntary Disclosures: Drivers of Proactive Information Provision By Local Governments In Central America." *Information Development* 32 (4):1199-1215.
- Garson, G David. 2012. "Testing Statistical Assumptions." Asheboro, NC: Statistical Associates Publishing.
- Gaspar, Abeid Francis, and Tausi Ally Mkasiwa. 2015. "Managing Performance or Legitimacy? A Case Study of the Tanzanian Local Government Authorities." *Journal of Accounting in Emerging Economies*.
- Gaventa, John, and Gregory Barrett. 2010. "So what difference does it make? Mapping the outcomes of citizen engagement." *IDS Working Papers* 2010 (347):01-72.
- Gaventa, John, and Rosemary McGee. 2013. "The Impact of Transparency and Accountability Initiatives." *Development Policy Review* 31:s3-s28.
- Gberevbie, Daniel, Segun Joshua, Nchekwube Excellence-Oluye, and Adeola Oyeyemi. 2017. "Accountability for Sustainable Development and The Challenges of Leadership in Nigeria, 1999-2015." Sage Open 7 (4):2158244017742951.
- George, Darren, and Paul Mallery. 2010. SPSS for Windows Step-By-Step: A Simple Guide and Reference : 18.0 Update. 11th ed. London;Boston, Mass;: Allyn and Bacon.
- Gesuele, Benedetta, Concetta Metallo, and Sergio Longobardi. 2018. "The Determinants of E-Disclosure Attitude: Empirical Evidences From Italian And Spanish Municipalities." *International Journal of Public Administration* 41 (11):921-934.
- Geys, Benny, Friedrich Heinemann, and Alexander Kalb. 2010. "Voter Involvement, Fiscal Autonomy and Public Sector Efficiency: Evidence From German Municipalities." *European Journal Of Political Economy* 26 (2):265-278.
- Ghani, Intan Martina Md, and Sabri Ahmad. 2010. "Stepwise Multiple Regression Method to Forecast Fish Landing." *Procedia - Social and Behavioral Sciences* 8:549-554.
- Ghozali, Imam. 2016. *Aplikasi Analisis Multivariete Dengan Program IBM SPSS* 23. Semarang: Badan Penerbit Universitas Diponegoro.
- Glavey, Sarah, Oliver Haas, Eddy Perez, Nila Mukherjee, Jacqueline Devine, Deviariandy Setiawan, and Craig Kullmann. 2015. The World Bank.
- Gopel, Maja. 2016. The Great Mindshift: How a New Economic Paradigm and Sustainability Transformations go Hand in Hand. Vol. 2. Cham: Springer International Publishing.
- Gorenflo, LJ, Catherine Corson, Kenneth M Chomitz, Grady Harper, Miroslav Honzák, and Berk Özler. 2011. "Exploring the association between people and deforestation in Madagascar." In *Human population*, 197-221. Springer.
- Goryakin, Yevgeny, Paul Revill, Andrew Mirelman, Rohan Sweeney, Jessica Ochalek, and Marc Suhrcke. 2017. "Public Financial Management and Health Service Delivery: A Literature Review." *London: Overseas Development Institute*.
- Graham, John, Timothy Wynne Plumptre, and Bruce Amos. 2003. "Principles for Good Governance In the 21st Century." *Policy Brief* 15.

- Gravetter, Frederick J. author, and Larry B. Wallnau. 2017. *Statistics for the Behavioral Sciences*. 10. ed. Boston, MA: Cengage Learning.
- Greenstein, Theodore N. 2006. *Methods of Family Research*. 2nd ed. London: SAGE.
- Greiling, Dorothea, Albert Anton Traxler, and Sandra Stötzer. 2015. "Sustainability Reporting in the Austrian, German and Swiss Public Sector." *International Journal of Public Sector Management*.
- Grignon, Julie L. 2019. "Predicting Relationship Satisfaction with Postformal thought and Adult Attachment Tendencies." Dissertation/Thesis, ProQuest Dissertations Publishing.
- Grimmelikhuijsen, Stephan. 2012. "Linking transparency, knowledge and citizen trust in government: an experiment." *International Review of Administrative Sciences* 78 (1):50-73.
- Groves, Sanford M, W Maureen Godsey, and Martha A Shulman. 1981. "Financial Indicators For Local Government." *Public Budgeting & Finance* 1 (2):5-19.
- Grüner, Andreas, and Ingeborg Raastad. 2018. "Financial Slack and Firm Performance during Economic Downturn." *GSL Journal of Business Management and Administration Affairs* 2 (1):1-7.
- Guillamón, Maria-Dolores, Francisco Bastida, and Bernardino Benito. 2011. "The Determinants of Local Government's Financial Transparency." *Local Government Studies* 37 (4):391-406.
- Gujarati, Damodar N., and Dawn C. Porter. 2009. *Basic Econometrics*. 5th, international ed. Boston: McGraw-Hill.
- Gunarto, Toto, Ahmad Sentri, and Muhammad Said. 2018. "Local Government Expenditures and Economic Growth in a New Autonomous in Indonesia " *Academy of Accounting and Financial Studies Journal* 22 (4).
- Guntur, Muhammad. 2017. "Implementation of Good Governance Principles Based on Transparency, Accountability and Public Participation in Indonesia." International Conference ADRI - 5 "Scientific Publications toward Global Competitive Higher Education, Makassar.
- Hadihardaja, Iwan K, and Neil S Grigg. 2011. "Decision Support System For Irrigation Maintenance In Indonesia: A Multi-Objective Optimization Study." *Water Policy* 13 (1):18-27.
- Hagquist, Curt, and Magnus Stenbeck. 1998. "Goodness of Fit In Regression Analysis–R 2 And G 2 Reconsidered." *Quality and Quantity* 32 (3):229-245.
- Hair, Joseph F., William C. Black, and Barry J. Babin. 2010. *Multivariate Data Analysis: A Global Perspective*. 7th ed. London: Pearson Education.
- Hair, Joseph F., Mary Wolfinbarger Celsi, Arthur H Money, Phillip Samouel, and Michael J Page. 2011. Essentials of Business Research Methods. 2nd ed. Armonk, N.Y: M.E. Sharpe.
- Hakkio, Craig S., and William R. Keeton. 2009. "Financial Stress: What Is It, How Can It Be Measured, and Why Does It Matter?" *Federal Reserve Bank of Kansas City Economic Review* 94 (2):5-50.
- Halim, Abdul. 2012. *Akuntansi Sektor Publik Akuntansi Keuangan Daerah*. Jakarta: Salemba Empat.
- Halkos, George E, and Nickolaos G Tzeremes. 2014. "Public Sector Transparency and Countries' Environmental Performance: A Nonparametric Analysis." *Resource and Energy Economics* 38:19-37.

- Halonen, Petteri. 2014. "Risk-Based Evaluation Of Internal Controls In Case Company's Sales Process-Case Company X." Department of Accounting, Aalto University.
- Han, Y., and S. Hong. 2016. "The Impact of Accountability on Organizational Performance in the U.S. Federal Government: The Moderating Role of Autonomy." *Review of Public Personnel Administration*.
- Han, Yousueng, and Sounman Hong. 2019. "The Impact of Accountability on Organizational Performance in the U.S. Federal Government: The Moderating Role of Autonomy." *Review of Public Personnel Administration* 39 (1):3-23.
- Handa, Rekha. 2018. "Does Corporate Governance Affect Financial Performance: A Study of Select Indian Banks." *Asian Economic and Financial Review* 8 (4):478.
- Handoyo, Sofik. 2018. "The Role of Public Governance in Environmental Sustainability." *Jurnal Ilmiah Peuradeun* 6 (2):161-178.
- Harrison, Walter T., and Charles T. Horngren. 2008. *Financial accounting*. 7th ed. Upper Saddle River, N.J: Pearson Prentice Hall.
- Harsanto, Bambang Tri, and Simin. 2006. "Desentralisasi Irigasi: Studi Kasus Pengelolaan Irigasi Di Daerah Irigasi Tajum Kabupaten Banyumas " *Jurnal Pembangunan Pedesaan* 6 (2).
- Harun, Harun, and Haryono P. Kamase. 2012. "Accounting Change and Institutional Capacity: The Case of a Provincial Government in Indonesia." *Australasian Accounting Business & Finance Journal* 6 (2):35-50.
- Haryanto, Agus. 2010. "Improving Service Quality of District Education Offices in Indonesian Decentralized Education System." PhD, School of Management, RMIT University
- Haryanto, Joko Tri 2015. Desentralisasi Fiskal Seutuhnya.
- Haryono, Untung, and Ardi Paminto. 2015. "Corporate Governance and Firm Value: The Mediating Effect of Financial Performance and Firm Risk." *European Journal of Business and Management* 7 (35):18-24.
- Havenga, Belinda 2002. "The Restructuring of Local Government with Specific Reference to The City of Tshwane." The Faculty Humanity, University of Pretoria.
- Hayes, Andrew F. author. 2018. Introduction To Mediation, Moderation, And Conditional Process Analysis: A Regression-Based Approach. Second ed. New York, NY: Guilford Press.
- He, Xiaoping, and Shuo Jiang. 2019. "Does Gender Diversity Matter for Green Innovation?" *Business Strategy and the Environment* 28 (7):1341-1356.
- Heaton, Janet. 2004. Reworking Qualitative Data. London: SAGE.
- Helpap, David J. 2015. "Explaining The Use of Recommended Practices And Guidelines: The Case Of Public Budgeting." *Public Administration Quarterly* 39 (2):259.
- Hidayati, Firda, Yogi Vidyattama, and Cameron Gordon. 2016. "Local Government Forestry Expenditure and Forest Land Cover: A Preliminary Lesson from Decentralized Indonesia." *Economics and Finance in Indonesia* 62 (3):127-140.
- Hinton, PR, C Brownlow, I Mcmurray, and B Cozens. 2011. SPSS Explained. Using Spss To Analyse Questionnaires: Reliability. Routledge Taylor & Francis Group, London.

- Hirsch, Werner Z., and Evan Osborne. 2000. "Privatization of Government Services: Pressure-Group Resistance and Service Transparency." *Journal of Labor Research* 21 (2):315-326.
- Hofman, Bert, and Kai Kaiser. 2004. "The Making of the 'Big Bang'and Its Aftermath: A Political Economy Perspective." In *Reforming intergovernmental fiscal relations and the rebuilding of Indonesia: The 'Big Bang'program and its economic consequences*, edited by Alm James, Jorge Martinez-Vasquez and Sri Mulyani Indrawati, 15-46. Cheltenham: Edward Elgar Publishing, Inc.
- Hofstad, Hege, and Trond Vedeld. 2021. "Exploring City Climate Leadership in Theory and Practice: Responding to the Polycentric Challenge." *Journal of Environmental Policy & Planning*:1-14.
- Honadle, Beth Walter. 2003. "The States' Role In Us Local Government Fiscal Crises: A Theoretical Model And Results Of A National Survey." *International Journal of Public Administration* 26 (13):1431-1472.
- Honadle, BW, JM Costa, and BA Cigler. 2004. "Fiscal Health for Local Governments: An Introduction to Concepts." *Practical Analisys and Strategies, San Diego California USA*.
- Hood, Christopher. 2007. "Public Service Management By Numbers: Why Does It Vary? Where Has It Come From? What Are The Gaps and The Puzzles?" *Public Money and Management* 27 (2):95-102.
- Hope, Sr Kempe Ronald. 2005. "Toward Good Governance and Sustainable Development: The African Peer Review Mechanism." *Governance* 18 (2):283-311.
- Hoque, Zahirul, and Maybelle Chia. 2012. "Competitive Forces and the Levers of Control Framework in A Manufacturing Setting." *Qualitative Research in Accounting & Management*.
- Howell, James M, and Charles F Stamm. 1979. Urban Fiscal Stress: A Comparative Analysis of 66 U.S. Cities the United States: Lexington.
- Hox, Joop J, and Hennie R Boeije. 2005. "Data Collection, Primary Versus Secondary." In *Encyclopedia of Social Measurement* 593–599.
- Hsu, Angel, Ainsley Lloyd, and John W Emerson. 2013. "What progress have we made since Rio? Results from the 2012 Environmental Performance Index (EPI) and Pilot Trend EPI." *Environmental Science & Policy* 33:171-185.
- Huang, Rongbing, and Yubo Huang. 2020. "Does Internal Control Contribute to a Firm's Green Information Disclosure? Evidence from China." *Sustainability* 12 (8):3197.
- Hudaya, Muhammad. 2014. "Understanding the Practice of Accountability Reporting: A Case of Indonesian Local Government." PhD, School of Accounting and Finance, University of Wollongong.
- Hudaya, Muhammad, Ciorstan Smark, Ted Watts, and Parulian Silaen. 2015. "The Use of Accountability Reports and the Accountability Forum: Evidence from an Indonesian Local Government." *Australasian Accounting Business & Finance Journal* 9 (4):57.
- Huka, Happiness Anton, Alban Dismas Mchopa, and Johnson James Kimambo. 2014. "Analysis of Procurement Procedures in Local Government Authorities: Experience After Procurement Reforms and Case Analysis from Selected Councils in Kilimanjaro Tanzania." *European Journal of Business and Management* 6 (18):16-22.

- Hussain, Intizar, Deeptha Wijerathna, Sigit S Arif, and Agnes Mawarni. 2006. "Irrigation, Productivity and Poverty Linkages in Irrigation Systems in Java, Indonesia." *Water Resources Management* 20 (3):313-336.
- Hutama, Rifqi Subhanullah, and Ivan Yudianto. 2019. "The Influence of Budget Participation, Budget Goals Clarity and Internal Control Systems Implementation on Local Government Performance." *Journal of Accounting Auditing and Business* 2 (2):58-76.
- Hutton, Guy, and Claire Chase. 2016. "The Knowledge Base for Achieving the Sustainable Development Goal Targets on Water Supply, Sanitation and Hygiene." *International Journal of Environmental Research and Public Health* 13 (6):536.
- Hwang, Kwangseon. 2013. "The Impact of Accountability and Accountability Management on Performance at the Street Level." 75, ProQuest Dissertations Publishing (Dissertation/Thesis).
- Hyman, David N. 2014. *Public Finance: A Contemporary Application of Theory to Policy*: Cengage Learning.
- IFAC. 2001. *Governance in the Public Sector: A Governing Body Perspective*. New York: International Federation of Accountants.
- IFAC. 2012. *Evaluating and Improving Internal Control in Organizations*. New York, NY: the International Federation of Accountants (IFAC).
- Indreswari, Adisti Dini 2016. "KPK: 3.982 Izin Usaha Pertambangan Bermasalah." *Kontan.*
- Industry Commission. 1997. Performance Measures for Councils: Improving Local Government Performance Indicators. Melbourne.
- Iqbal, Muhammad. 2017. "Kedudukan Gubernur Dalam Hubungannya Terhadap Bupati/Wali Kota Sebagai Perwakilan Pemeirntah Pusat (Ditinjau Dari Undang-Undang Nomor 5 Tahun 1974 Tentang Pokok-Pokok Pemerintahan Di Daerah Dengan Undang-Undang Nomor 22 Tahun 1999 Dengan Undang Undang Nomor 32 Tahun 2004 Dengan Undang-Undang Nomor 23 Tahun 2014 Tentang Pemerintahan Daerah)." Law Faculty, University of North Sumatra
- Islamiyah, Yunita, Eko Ganis Sukoharsono, and Rosidi. 2017. "Public Sector's Managerial and Financial Accountability." *The International Journal of Accounting and Business Society* 25 (1):1-18.
- Iwata, Hiroki, and Keisuke Okada. 2011. "How Does Environmental Performance Affect Financial Performance? Evidence From Japanese Manufacturing Firms." *Ecological Economics* 70 (9):1691-1700.
- Jamali, Dima. 2010. "MNCs and international accountability standards through an institutional lens: Evidence of symbolic conformity or decoupling." *Journal of Business Ethics* 95 (4):617-640.
- Järvenpää, Marko. 2009. "The Institutional Pillars of Management Accounting Function." *Journal of Accounting & Organizational Change*.
- Jasch, Christine. 2000. "Environmental Performance Evaluation And Indicators." Journal of Cleaner Production 8 (1):79-88.
- Jeremiah, Mfon Solomon. 2018. "Corporate Social Responsibility Contribution to Environmental Sustainability in Developing Countries: The Accountability Perspective." *International Journal of Business and Management* 13 (1).
- Jha, Shikha, Juzhong Zhuang, and Pilipinas Quising. 2013. "Governance and Public Service Delivery." In Asian Development Outlook 2013 Update. Governance

and Public Service Delivery, edited by Takehiko Nakao and Changyong Rhee. Mandaluyong City, Philippines: Asian Development Bank.

- Johnston, Mel. 2013. "Mimetic, Coercive and Normative Influences and the Decision of National Sport Organisations to Bid for World Championship Events." PhD, Auckland University of Technology.
- Johnston, Melissa P. 2017. "Secondary data analysis: A method of which the time has come." *Qualitative and quantitative methods in libraries* 3 (3):619-626.
- Jokipii, Annukka. 2010. "Determinants And Consequences of Internal Control In Firms: A Contingency Theory Based Analysis." *Journal of Management and Governance* 14 (2):115-144.
- Jones, LG. 2006. "Environmental Key Performance Indicator, Reporting Guidelines for UK Business." *Queen's Printer and Controller. London*.
- Jones, Mark P. 2009. "Gender Quotas, Electoral Laws, and the Election of Women: Evidence From the Latin American Vanguard." *Comparative Political Studies* 42 (1):56-81.
- Jones, Stewart, and Robert G Walker. 2007. "Explanators of Local Government Distress." *Abacus* 43 (3):396-418.
- Joshi, Anuradha. 2013. "Do They Work? Assessing The Impact of Transparency and Accountability Initiatives in Service Delivery." *Development Policy Review* 31 (s1).
- JRC. 2008. Handbook On Constructing Composite Indicators: Methodology And User Guide: OECD publishing.
- Jütting, Johannes, Elena Corsi, Céline Kauffmann, Ida McDonnell, Holger Osterrieder, Nicolas Pinaud, and Lucia Wegner. 2005. "What Makes Decentralisation in Developing Countries Pro-Poor?" *European Journal of Development Research* 17 (4):626-648.
- Kamnikar, Judith A, Edward G Kamnikar, and Keren H Deal. 2006. "Assessing A State's Financial Condition." *The Journal of Government Financial Management* 55 (3):30.
- Karagiorgos, Theofanis, George Drogalas, and Nikolaos Giovanis. 2011. "Evaluation of The Effectiveness Of Internal Audit In Greek Hotel Business." *International Journal of Economic Sciences and Applied Research* 4 (1):19-34.
- Kareem, Akeel Dakheel, Waad Hadi Abd AlHassani, and Raid Hassan Ali. 2019.
 "The Role of Internal Auditing in Raising the Performance of Institutions and Improving the Environment an Applied Study in one of the Environmental Departments." *International Journal of Innovation, Creativity and Change* 8 (3).
- Karim, Rusdi Abdul, and Mursalim Mursalim. 2019. "Pengaruh Akuntabilitas, Transparansi dan Sistem Pengendalian Intern Terhadap Kinerja Keuangan." *Jurnal Ilmu Ekonomi* 2 (1):39-49.
- Kaufmann, Wesley, and Anne Lafarre. 2020. "Does Good Governance Mean Better Corporate Social Performance? A Comparative Study of OECD Countries." *International Public Management Journal*:1-30.
- Kearns, Kevin P. 1994. "The Strategic Management Of Accountability In Nonprofit Organizations: An Analytical Framework." *Public administration review*:185-192.
- Kearns, Kevin P. 1996. *Managing For Accountability: Preserving The Public Trust In Public And Nonprofit Organizations*: Jossey-Bass.

- Kesuma, Sinar Indra, Yusak Maryunianta, and Iskandar Muda. 2018. "Evaluation of Irrigation System to Support Implementation of Food Security Policy." International Journal of Civil Engineering and Technology 9 (9):600-614.
- Kewo, Cecilia Lelly. 2017. "The influence of Internal Control Implementation and Managerial Performance on Financial Accountability Local Government in Indonesia." *International Journal of Economics and Financial Issues* 7 (1).
- Khairudin, Rosmiati Tarmizi, Indrayenti, Aminah, and Kamaruzzaman Muhammad.
 2018. "Financial Performance of Local Government and Social Welfare: The Case in Indonesia." *The Journal of Social Sciences Research*:1035-1039: 5.
 Khanna, Madhu, and Surender Kumar. 2011. "Corporate Environmental
- Khanna, Madhu, and Surender Kumar. 2011. "Corporate Environmental Management and Environmental Efficiency." *Environmental and Resource Economics* 50 (2):227-242.
- Khotami. 2017. "The Concept of Accountability In Good Governance." International Conference on Democracy, Accountability and Governance Riau.
- Kilfoyle, Eksa, and Alan J Richardson. 2011. "Agency and Structure in Budgeting: Thesis, Antithesis and Synthesis." *Critical Perspectives on Accounting* 22 (2):183-199.
- Kim, Hae-Young. 2013. "Statistical Notes for Clinical Researchers: Assessing Normal Distribution (2) Using Skewness and Kurtosis." *Restorative Dentistry* & *Endodontics* 38 (1):52.
- Kim, Seok Eun, and Jung Wook Lee. 2010. "Impact of Competing Accountability Requirements on Perceived Work Performance." *The American Review of Public Administration* 40 (1):100-118.
- King, Lester O. 2016. "Functional Sustainability Indicators." *Ecological Indicators* 66:121-131.
- Kioko, Sharon N, Justin Marlowe, David ST Matkin, Michael Moody, Daniel L Smith, and Zhirong J Zhao. 2011. "Why Public Financial Management Matters." *Journal of Public Administration Research and Theory* 21 (suppl_1):i113-i124.
- Kiringai, JA. 2002. "Budget Reforms and Medium Term Expenditure Framework in Kenya." *Kenya Institute for Public Policy, Research and Analysis.*
- Kis-Katos, Krisztina, and Bambang Suharnoko Sjahrir. 2017. "The Impact of Fiscal and Political Decentralization on Local Public Investment in Indonesia." *Journal of Comparative Economics* 45 (2):344-365.
- Kloha, Philip, Carol S Weissert, and Robert Kleine. 2005a. "Someone To Watch Over Me State Monitoring of Local Fiscal Conditions." *The American Review* of *Public Administration* 35 (3):236-255.
- Kloha, Philip, Carol S. Weissert, and Robert Kleine. 2005b. "Developing and Testing a Composite Model to Predict Local Fiscal Distress." *Public Administration Review* 65 (3):313-323.
- Kloot, Louise. 1999. "Performance Measurement and Accountability in Victorian Local Government." *International Journal of Public Sector Management* 12 (7):565-584.
- Kluvers, Ron. 2003. "Accountability for Performance in Local Government." *Australian Journal of Public Administration* 62 (1):57-69.
- Kohn, Alfie. 1999. Punished by Rewards:: The Trouble with Gold Stars, Incentive Plans, A's, Praise, and Other Bribes: Houghton Mifflin Harcourt.
- Kolstad, Ivar, and Arne Wiig. 2009. "Is Transparency The Key To Reducing Corruption in Resource-Rich Countries?" *World development* 37 (3):521-532.
- Konar, Shameek, and Mark A Cohen. 2001. "Does the market value environmental performance?" *Review of economics and statistics* 83 (2):281-289.

Kotarba, Boguslaw, and Anna Kolomycew. 2014. "Financial Independence of Local Government Units in Poland." *Journal of Universal Excellence* 3 (4):18-35.

- Králik, Lóránd-István. 2019. "Financing and Financial Independence of Local Governments from Valea Ierului and Valea Barcăului." *Forum on Economics and Business* 22 (140):59-75.
- Krell, Katharina, Sabine Matook, and Fiona Rohde. 2009. "The effects of regulatory pressure on information system adoption success: An institutional theory perspective."
- Krishnamohan, T. 2016. "The Local Government System in Indonesia and Sri Lanka: A Comparative Overview." International Journal in Commerce, IT & Social Sciences 3 (12).

Kristiansen, Stein, Agus Dwiyanto, Agus Pramusinto, and Erwan Agus Putranto. 2009. "Public Sector Reforms and Financial Transparency: Experiences from Indonesian Districts." *Contemporary Southeast Asia* 31 (1):64-87.

Kristiansen, Stein, and Lambang Trijono. 2005. "Authority and Law Enforcement: Local Government Reforms and Security Systems in Indonesia." *Contemporary Southeast Asia* 27 (2):236-254.

- Kristianto, M. Agus 2017. "Penerapan Goods and Services Tax di Indonesia." *Jurnal Defis Edisi 1* 1 (1):62-70.
- Kuepie, Mathias, David Shapiro, and Michel Tenikue. 2015. "Access to Schooling and Staying in School in Selected Sub-Saharan African Countries." *African Development Review* 27 (4):403-414.
- Kulshreshtha, Praveen. 2008. "Public Sector Governance Reform: The World Bank's Framework." International Journal of Public Sector Management.
- Kuncoro, H. 2012. "Does the Local Economic Governance in Indonesia Perform an Improvement?" *Bulletin of Monetary Economic and Banking* 83-107.
- Kuncoro, Mudrajad. 2018. *Perencanaan Pembangunan Daerah: Teori dan Aplikasi.* Jakarta: PT Gramedia Pustaka Utama.

Kusmana, Cecep. 2011. "Forest Resources And Forestry In Indonesia." *Forest Science and Technology* 7 (4):155-160.

Kusmayadi, Rudy Catur Rohman. 2015. "Kewenangan Pemerintah Daerah Dalam Penyelenggaraan Investasi Daerah." *Jurnal Pusaka* 3 (1):65-80.

Kyriacou, Andreas P, Leonel Muinelo-Gallo, and Oriol Roca-Sagalés. 2017. "Regional Inequalities, Fiscal Decentralization And Government Quality." *Regional Studies* 51 (6):945-957.

Lachowicz, Mark J., Kristopher J. Preacher, and Ken Kelley. 2018. "A Novel Measure of Effect Size for Mediation Analysis." *Psychological Methods* 23 (2):244-261.

- Laffan, Brigid. 2001. "The European Union Polity: A Union of Regulative, Normative and Cognitive Pillars." *Journal of European Public Policy* 8 (5):709-727.
- Laily, Elida Imro'atin Nur. 2015. "Partisipasi Masyarakat Dalam Perencanaan Pembangunan Partisipatif." *Jurnal Kebijakan dan Manajemen Publik* 3 (2).
- Laksono, Agung Dwi, Ilham Akhsanu Ridlo, and Ernawaty Ernawaty. 2020. "Distribution Analysis of Doctors in Indonesia." *Jurnal Administrasi Kesehatan Indonesia* 8 (1).
- Lancee, Jaap, Maarten C Eisma, Annemieke van Straten, and Jan H Kamphuis. 2015. "Sleep-Related Safety Behaviors And Dysfunctional Beliefs Mediate The Efficacy of Online Cbt For Insomnia: A Randomized Controlled Trial." *Cognitive Behaviour Therapy* 44 (5):406-422.

- Lang, Harald. 2016. "Elements of regression analysis." Stockholm: KTH Mathematics.
- Lanin, Dasman, and Nailuredha Hermanto. 2019. "The Effect of Service Quality Toward Public Satisfaction and Public Trust on Local Government in Indonesia." *International Journal of Social Economics*.
- Lannai, Darwis, and Asbi Amin. 2020. "Factors Affect Budget Absorption in Government Institutions of South Sulawesi." *Jurnal Ilmiah Akuntansi* 3 (2):113-119.
- Laswad, Fawzi, Richard Fisher, and Peter Oyelere. 2005. "Determinants of Voluntary Internet Financial Reporting By Local Government Authorities." *Journal of Accounting and Public Policy* 24 (2):101-121.
- Laughlin, Richard C. 1990. "A Model of Financial Accountability and The Church of England." *Financial Accountability and Management* 6 (2):93-114.
- Lawton, Alan, David McKevitt, and Michelle Millar. 2000. "Developments: Coping With Ambiguity: Reconciling External Legitimacy and Organizational Implementation in Performance Measurement." *Public Money and Management* 20 (3):13-20.
- Lawton, Alan, David McKevitt, and Michelle Millar. 2010. "Developments: Coping with Ambiguity: Reconciling External Legitimacy and Organizational Implementation in Performance Measurement." *Public Money and Management* 20 (3):13-20.
- Leblois, Antoine, Olivier Damette, and Julien Wolfersberger. 2017. "What Has Driven Deforestation in Developing Countries Since the 2000s? Evidence from New Remote-Sensing Data." *World Development* 92:82-102.
- Lekipiouw, Sherlock Halmes. 2020. "Konstruksi Penataan Daerah dan Model Pembagian Urusan Pemerintahan." SASI 26 (4):557-570.
- Leung, Philomena, Barry J Cooper, and Luckmika Perera. 2011. "Accountability Structures And Management Relationships of Internal Audit." *Managerial Auditing Journal* 26 (9):794-816.
- Levina, Maria. 2015. "The Effect of Tax Extensification, Taxpayer Compliance, and Tax Arrears Search To Acceptance of Personal Income Tax." *Jurnal Ilmiah Wahana Akuntansi* 10 (2):140-153.
- Lewis, Blane D. 2017a. "Does Local Government Proliferation Improve Public Service Delivery? Evidence From Indonesia." *Journal of Urban Affairs* 39 (8):1047-1065.
- Lewis, Blane D. 2017b. "Local Government Spending and Service Delivery in Indonesia: The Perverse Effects of Substantial Fiscal Resources." *Regional Studies* 51 (11):1695-1707.
- Lewney, Richard. 2021. "Environmental Policies to Save the Planet." In *Economic Policies for a Post-Neoliberal World*, edited by Philip Arestis and Malcolm Sawyer, 179-223. Switzerland Palgrave Macmillan, Cham.
- Li, Ji, Jane Moy, Kevin Lam, and WL Chris Chu. 2008. "Institutional Pillars and Corruption at The Societal Level." *Journal of Business Ethics* 83 (2):327-339.
- Li, Xiao, Chunmei Zheng, Gang Liu, and Muhammad Safdar Sial. 2018. "The Effectiveness of Internal Control and Corporate Social Responsibility: Evidence from Chinese Capital Market." *Sustainability* 10 (11):4006.
- Liou, Kuotsai Tom. 2011. "Public Budgeting And Finance Reforms in China: Introduction." *Journal of Public Budgeting, Accounting & Financial Management* 23 (4):535.

- Liu, Jau-Yang. 2018. "An Internal Control System that Includes Corporate Social Responsibility for Social Sustainability in the New Era." *Sustainability* 10 (10):3382.
- Liu, Lei, Tong Wu, Shaoshuang Li, Martin de Jong, and Yanhong Sun. 2017. "The Drivers of Local Environmental Policy in China: An Analysis of Shenzhen's Environmental Performance Management System, 2007–2015." *Journal of Cleaner Production* 165:656-666.
- Lockwood, Chondra M, and David P MacKinnon. 1998. "Bootstrapping The Standard Error Of The Mediated Effect." Proceedings of the 23rd annual meeting of SAS Users Group International.
- Lockwood, Michael. 2010. "Good Governance For Terrestrial Protected Areas: A Framework, Principles and Performance Outcomes." *Journal of environmental management* 91 (3):754-766.
- Lockwood, Michael, Julie Davidson, Allan Curtis, Elaine Stratford, and Rod Griffith. 2010. "Governance Principles For Natural Resource Management." *Society and natural resources* 23 (10):986-1001.
- Lodhia, Sumit, Kerry Jacobs, and Yoon Jin Park. 2012. "Driving public sector environmental reporting: the disclosure practices of Australian commonwealth departments." *Public Management Review* 14 (5):631-647.
- Lohrey, Robin C, Claire A Horner, Belinda R Williams, and Trevor D Wilmshurst. 2019. "Coming to Grips With Corporate Governance in Local Government." *Australian Journal of Public Administration* 78 (4):596-612.
- Lopez Subires, Maria Deseada, Laura Alcaide Munoz, Andres Navarro Galera, and Manuel Pedro Rodriguez Bolivar. 2019. "The Influence of Socio-Demographic Factors on Financial Sustainability of Public Services: A Comparative Analysis in Regional Governments and Local Governments." *Sustainability* 11 (21):1-18.
- Lotz, Joergen R, and RA Bahl. 1981. "Fiscal Problems and Issues In Scandinavian Cities." *Urban Government Finance: Emerging Trends*:221-243.
- Lowatcharin, Grichawat, and Charles E Menifield. 2015. "Determinants of Internet-Enabled Transparency At The Local Level: A Study of Midwestern County Web Sites." *State and Local Government Review* 47 (2):102-115.
- Lu, Jing, and Irene M. Herremans. 2019. "Board Gender Diversity and Environmental Performance: An Industries Perspective." *Business Strategy and the Environment* 28 (7):1449-1464.
- Lubowski, Ruben, Max Wright, Kalifi Ferretti-Gallon, A.Javier Miranda Arana, Marc Steininger, and Jonah Busch. 2014. Mexico Deforestation Vulnerability Analysis and Capacity Building. Mexico.
- Mackay, Peter, and Sara B. Moeller. 2007. "The Value of Corporate Risk Management." *The Journal of Finance* 62 (3):1379-1419.
- MacKinnon, David P. 2011. "Integrating Mediators And Moderators In Research Design." *Research on social work practice* 21 (6):675-681.
- Magee, Tracy, Susan M Lee, Karen K Giuliano, and Barbara Munro. 2006. "Generating New Knowledge From Existing Data: The Use of Large Data Sets For Nursing Research." *Nursing Research* 55 (2):S50-S56.
- Mahadeen, Buthayna, R Al-Dmour, Bader Yousef Obeidat, and Ali Tarhini. 2016.
 "Examining The Effect of The Organization's Internal Control System on Organizational Effectiveness: A Jordanian Empirical Study." *International Journal of Business Administration* 7 (6):22-41.

- Maksimovska, Aleksandra, and Aleksandar Stojkov. 2016. "Social Responsiveness and Service Performance of Local Government: The Case of Macedonia." *Croat. & Comp. Pub. Admin.* 16:219.
- Malalgoda, Chamindi, Dilanthi Amaratunga, and Richard Haigh. 2016. "Overcoming Challenges Faced by Local Governments in Creating A Resilient Built Environment in Cities." *Disaster Prevention and Management: An International Journal* 25 (5):628-648.
- Manaf, Asnawi, Budi Setiyono, Imam Wahyudi, Micah Fisher, and Hendri Yuzal. 2016. "Implementing Pro-Poor Planning and Budgeting: A Case Study of Government-Community Poverty Alleviation Partnership in Pekalongan City, Indonesia." *International Journal of Sustainable Society* 8 (4):302-317.
- Manafe, Mesri Welhelmina Nisriani, and Rusdi Akbar. 2014. "Accountability and Performance: Evidence from Local Government." *Journal of Indonesian Economy and Business* 29 (1):56-73.
- Marchica, Maria-Teresa, and Roberto Mura. 2010. "Financial Flexibility, Investment Ability, and Firm Value: Evidence from Firms with Spare Debt Capacity." *Financial Management* 39 (4):1339-1365.
- Mardiasmo, Diaswati, Paul H Barnes, and Yuka Sakurai. 2008. "Implementation of Good Governance By Regional Governments In Indonesia: The Challenges." Twelfth Annual Conference of the International Research Society for Public Management, Brisbane.
- Marhani, Atika Ritmelina. 2018. "The Utilization of Drones For Smart Governance Implementation: Tax Extensification Strategy For Transfer of Ownership In Real Estate Sales At Cibinong Primary Tax Office." 2018 International Conference on ICT for Smart Society (ICISS).
- Marlissa, Elsyan Rienette, and Jhon Urasti Blesia. 2018. "Fiscal Dependence in a Special Autonomy Region: Evidence from a local government in Eastern Indonesia." *Journal of Economic Development, Environment and People* 7 (1):55-72.
- Martinez-Vazquez, Jorge, Santiago Lago-Peñas, and Agnese Sacchi. 2017. "The Impact of Fiscal Decentralization: A Survey." *Journal of Economic Surveys* 31 (4).
- Martinez-Vazquez, Jorge, and Bob Searle. 2007. "Challenges in The Design of Fiscal Equalization and Intergovernmental Transfers." In *Fiscal Equalization*, 3-10. Springer.
- Martinez-Vazquez, Jorge, and Andrey Timofeev. 2010. "Choosing Between Centralized and Decentralized Models of Tax Administration." *International Journal of Public Administration* 33 (12-13):601-619.
- Martinez-Vazquez, Jorge, and Francois Vaillancourt. 2011. "Decentralization in developing countries: Global perspectives on the obstacles to fiscal devolution." In.
- Masanja, Ndalahwa Musa. 2018. "The Impact of Internal Control Challenges on Organizational Financial Performance for Selected Local Government Authorities in Arumeru District, Arusha Tanzania." *International Journal o Research and Innovation In Social Science*:206-211.
- Mashayekhi, Bita, and Hamid Kalhornia. 2016. "Relationship between Financial Reporting Transparency and Investment Efficiency: Evidence from Iran." International Journal of Economics and Management Engineering 10 (7):2433-2437.

Masiya, Tyanai, Yul D Davids, and Mary S Mangai. 2019. "Assessing Service Delivery: Public Perception of Municipal Service Delivery in South Africa."

- Masron, Tajul Ariffin, and Yogeeswari Subramaniam. 2019. "Does Poverty Cause Environmental Degradation? Evidence From Developing Countries." *Journal* of Poverty 23 (1):44-64.
- Maurel, Christophe, David Carassus, Christophe Favoreu, and Damien Gardey. 2014. "Characterization and Definition of Public Performance: An Application to Local Government Authorities." *Gestion et management public* 2 (1):23-44.
- Mayston, David. 1993. "Principals, Agents and The Economics of Accountability In The New Public Sector." *Accounting, Auditing & Accountability Journal* 6 (3):68.
- Mbecke, Paulin. 2014. "Corporate Municipal Governance For Effective And Efficient Public Service Delivery in South Africa." *Journal of Governance and Regulation* 3 (4):98-106.
- McCarthy, John F. 2001. "Decentralization, Local Communities And Forest Management In Barito Selatan." *Centre for International Forestry Research.*
- Mees, Heleen, and Peter Driessen. 2019. "A Framework for Assessing the Accountability of Local Governance Arrangements for Adaptation to Climate Change." *Journal of Environmental Planning and Management* 62 (4):671-691.
- Meier, Iwan, Yves Bozec, and Claude Laurin. 2013. "Financial Flexibility and the Performance During the Recent Financial Crisis." *International Journal of Commerce and Management* 23 (2):79-96.
- Meliala, Andreasta, Krishna Hort, and Laksono Trisnantoro. 2013. "Addressing The Unequal Geographic Distribution Of Specialist Doctors In Indonesia: The Role Of The Private Sector And Effectiveness Of Current Regulations." *Social science & medicine* 82:30-34.
- Menlhk. 2018. *Status Hutan & Kehutanan Indonesia 2018*. Jakarta: The Ministry of Environment and Forestry of the Republic of Indonesia.
- Mera, Koichi. 2000. "Decentralization and capacity building: Selecting modes of training for Indonesia."
- Meule, Adrian. 2019. "Contemporary Understanding of Mediation Testing." *Meta-Psychology* 3.
- Meyer, J. W., and B Rowan. 1977. "Institutionalized Organizations: Formal Structure as Myth and Ceremony." *American Journal of Sociology* 83 (2):340-363.
- Mihaiu, Diana Marieta, Alin Opreana, and Marian Pompiliu Cristescu. 2010. "Efficiency, Effectiveness and Performance of the Public Sector." *Romanian Journal of Economic Forecasting* 4 (1):132-147.
- Mimba, Ni Putu SH, G Helden, and Sandra Tillema. 2007. "Public Sector Performance Measurement In Developing Countries: A Literature Review and Research Agenda." *Journal of Accounting and Organizational Change* 3 (3):192-208.
- Mishkin, Frederic S. 1990. Asymmetric information and financial crises: a historical perspective. National Bureau of Economic Research.
- Modell, Sven. 2012. "Strategy, political regulation and management control in the public sector: institutional and critical perspectives." *Management Accounting Research* 23 (4):278-295.
- Modlin, Steve, and La Shonda M Stewart. 2012. "Cash Management Practices Among Southeastern County Governments: Proper Utilization or Excessive Caution." *Public Finance & Management* 12 (2).

Mohrman, Mary Beth. 1996. "The Use of Fixed Gaap Provisions In Debt Contracts." Accounting Horizons 10 (3):78.

- Montes, Gabriel Caldas, Júlio Cesar Albuquerque Bastos, and Ana Jordânia de Oliveira. 2019. "Fiscal Transparency, Government Effectiveness and Government Spending Efficiency: Some International Evidence Based on Panel Data Approach." *Economic Modelling* 79:211-225.
- Mood, Carina, and Jan O Jonsson. 2016. "The Social Consequences of Poverty: An Empirical Test on Longitudinal Data." *Social indicators research* 127 (2):633-652.
- Morudu, Hlabi Dunstan. 2017. "Service Delivery Protests in South African Municipalities: an Exploration Using Principal Component Regression and 2013 Data." *Cogent Social Sciences* 3 (1):1329106.
- Mulgan, Richard. 2000. "Accountability': An Ever-Expanding Concept?" *Public Administration* 78 (3):555-573.
- Mulumba, Jane Nabunnya, Christophe Nothomb, Alana Potter, and Marielle Snel. 2014. "Striking the Balance: What Is the Role of the Public Sector in Sanitation as a Service and as a Business?" *Waterlines* 33 (3):195-210.
- Mulyani, Sri. 2017a. Sosialiasi Transfer ke Daerah dan Dana Desa TA 2018 & Penganugrahan Dana Rakca Award. Jakarta: Directorat General of Fiscal Balance.
- Mulyani, Sri Mulyani. 2017b. Penggunaan, Pemantauan, Dan Evaluasi Dana Bagi Hasil Cukai Hasil Tembakau. In *PMK Number 222/PMK.07/2017*. Jakarta: The Ministry of Finance.
- Mulyani, Sri Mulyani. 2017c. Penggunaan, Pemantauan, dan Evaluasi Dana Bagi Hasil Sumber Daya Alam Kehutanan Dana Reboisasi. In *PMK Number* 230/PMK.07/2017. Jakarta: The Ministry of Finance.
- Muraleetharan, P. 2011. "Internal control and impact of financial performance of the organizations (special reference public and private organizations in jaffna district)." International Conference on Business & Information [C/OL].
- Muskanan, Moni. 2014. "The Effectiveness of the Internal Controlling System Implementation in Fiscal Decentralization Application." *Procedia, Social and Behavioral Sciences* 164:180-193.
- Mutiha, Arthaingan H. 2018. "The Effect of Financial Ratio to Human Development Index (Based on the Study in Districts and Cities of West Java)." The 2nd International Conference on Vocational Higher Education (ICVHE) "The Importance on Advancing Vocational Education to

Meet Contemporary Labor Demands".

- Nasution, Anwar. 2014. *Macroeconomic Policies in Indonesia: Indonesia economy since the Asian financial crisis of 1997*: Routledge.
- Nasution, Anwar. 2017. "The Government Decentralization Program In Indonesia." In *Central and Local Government Relations in Asia*. Jakarta: Edward Elgar Publishing.
- Nasution, Dito Aditia Darma. 2018. "Analisis Pengaruh Pengelolaan Keuangan Daerah, Akuntabilitas Dan Transparansi Terhadap Kinerja Keuangan Pemerintah." *Jurnal Studi Akuntansi & Keuangan* 2 (3):149-162.
- Nawir, Ani Adiwinata, and Lukas Rumboko. 2007. *Forest Rehabilitation In Indonesia: Where To After More Than Three Decades?* Jakarta: Center for International Forestry Research (CIFOR).

- Nguyen, Cuong Viet, Long Thanh Giang, Anh Ngoc Tran, and Huyen Thanh Do. 2019. "Do Good Governance and Public Administration Improve Economic Growth and Poverty Reduction? The Case of Vietnam." *International Public Management Journal*:1-27.
- Nikijuluw, Ruth. 2012. "Analisis Pengaruh Transfer Pemerintah Pusat terhadap Upaya Pemungutan Pajak Kabupaten/Kota di Indonesia." *Jurnal Ekonomi dan Pembangunan Indonesia* 12 (2):168-191.
- Ningsih, Nurdiana, Andi Kusumawati, and Mursalim Nohong. 2019. "Moderation of the Internal Control System on the Effect of Unethical Behavior on Accounting Fraud in the Enrekang Regency." *International Journal of Advanced Research* 7 (5):437-450.
- Niswaty, Risma, J Manno, and Haedar Akib. 2015. "An Analysis of The Public Service Performance Based on Human Development Index in Makassar City, Indonesia." International Journal of Applied Business and Economic Research (IJABER) 13 (6):4421-4429.
- Nollenberger, Karl, Sanford Groves, and Maureen G Valente. 2003. *Evaluating Financial Condition: A Handbook for Local Government*. Washington, DC: International City/County Management Association.
- Nopyandri, Nopyandri. 2014. "Penerapan Prinsip Good Enviromental Governance Dalam Rangka Perlindungan Dan Pengelolaan Lingkungan Hidup." *Jurnal Ilmu Hukum Jambi* 5 (2):43278.
- Nouri, Masoud, and Seyedeh Mahboobeh Jafari. 2016. "The Impact of Financial Flexibility on Investment Efficiency (Over-Investment and Under-Investment) With Respect to Managerial Ownership in the Firms Listed in Tehran Stock Exchange." *Economics and Finance* 3 (2):18-22.
- Nunnally, Jum C., and Ira H. Bernstein. 1994. *Psychometric theory*. 3rd ed. New York: McGraw-Hill.
- Nuswantara, Dian Anita, and Ach Maulidi. 2017. "The Efficacy of Control Environment as Fraud Deterrence in Local Government." *Management & Marketing. Challenges for the Knowledge Society* 12 (4):591-613.
- O'Donnell, Owen. 2007. "Access to Health Care in Developing Countries: Breaking Down Demand Side Barriers." *Cadernos de saude publica* 23 (12):2820-2834.
- O'Leary, Conor, Errol Iselin, and Divesh Sharma. 2006. "The Relative Effects of Elements Of Internal Control on Auditors' Evaluations of Internal Control." *Pacific Accounting Review* 18 (2):68-95.
- Octovido, Irsandy, Nengah Sudjana, and Devi Farah Azizah. 2014. "Analisis Efektivitas Dan Kontribusi Pajak Daerah Sebagai Sumber Pendapatan Asli Daerah Kota Batu (Studi Pada Dinas Pendapatan Daerah Kota Batu Tahun 2009-2013)." *Jurnal Administrasi Bisnis* 15 (1).
- Odagiri, Mitsunori, Zainal Muhammad, Aidan A Cronin, Michael E Gnilo, Aldy K Mardikanto, Khaerul Umam, and Yameha T Asamou. 2017. "Enabling Factors For Sustaining Open Defecation-Free Communities In Rural Indonesia: A Cross-Sectional Study." *International Journal of Environmental Research and Public Health* 14 (12):1572.
- Odoyo, Collins Otien, Oginda Moses, Mark Obura John, Fredrick O Aila, Patrick B Ojera, and Elijah M Siringi. 2013. "Effect of Information Systems on Revenue Collection by Local Authorities in Homa Bay County, Kenya." Universal Journal of Accounting and Finance 1 (1):29-33.

- OECD. 2003. Public Sector Transparency and the International Investor. Paris: OECD.
- OECD. 2017. Internal Control and Risk Management for Public Integrity in the Middle East and North Africa: OECD Publishing Paris.
- Okoye, Lawrence Uchenna, Alexander Ehimare Omankhanlen, Johnson I Okoh, Ese Urhie, and Ado Ahmed. 2019. "Government Expenditure And Economic Growth: The Case Of Nigeria." International Conference on Education, Social Sciences and Humanities, Istanbul, Turkey.
- Ong'unya, Gordon Otieno, and Kalenzi Abbey. 2019. "Internal Control and Quality Service Delivery in A Public Health Sector: A Case Study of a Local Government in Uganda." *African Journal of Business Management* 13 (16):557-563.
- OPHI. 2018. Global Multidimensional Poverty Index 2018: The Most Detailed Picture To Date of the World's Poorest People. UK: University of Oxford.
- Opler, Tim C., and Sheridan Titman. 1994. "Financial Distress and Corporate Performance." *The Journal of Finance* 49 (3):1015-1040.
- Orcan, Fatih. 2020. "Parametric or Non-Parametric: Skewness to Test Normality for Mean Comparison." International Journal of Assessment Tools in Education 7 (2):255-265.
- Orlitzky, Marc, Frank L. Schmidt, and Sara L. Rynes. 2003. "Corporate Social and Financial Performance: A Meta-Analysis." *Organization Studies* 24 (3):403-441.
- Ostwald, Kai, Yuhki Tajima, and Krislert Samphantharak. 2016. "Indonesia's Decentralization Experiment: Motivations, Successes, And Unintended Consequences." *Journal of Southeast Asian Economies (JSEAE)* 33 (2):139-156.
- Ott, Katarina, Mihaela Bronić, and Miroslav Petrušić. 2014. "Budget Transparency of Croatian Counties, Cities and Municipalities in 2013/2014." *Newsletter-An Occasional Publication of the Institute of Public Finance* 16 (87):1-11.
- Oulasvirta, Lasse, and Maciej Turała. 2005. *Measuring the Financial Autonomy of Local Governments With A Local Autonomy Index*: Caledonian Business School.
- Oulasvirta, Lasse, and Maciej Turała. 2007. "Financial Autonomy of Local Governments: Case Studies of Finland and Poland." In *Regionalisation, Growth, and Economic Integration*, 251-267.
- Palmer, Charles. 2001. "The Extent and Causes of Illegal Logging: An Analysis of A Major Cause of Tropical Deforestation In Indonesia." CSERGE Working Papers.
- Palmer, Charles, and Stefanie Engel. 2007. "For Better Or For Worse? Local Impacts of The Decentralization of Indonesia's Forest Sector." *World Development* 35 (12):2131-2149.
- Palthe, Jennifer. 2014. "Regulative, Normative, and Cognitive Elements of Organizations: Implications For Managing Change." *Management and organizational studies* 1 (2):59-66.
- Pandjaitan, Edison 2016. Indonesia, Educational Statistics in Brief, 2015/2016. Jakarta: Ministry of Education and Culture.
- Panya, Napawan, Chamlong Poboon, Wisakha Phoochinda, and Ratiporn Teungfung. 2018. "The Performance of The Environmental Management of Local Governments in Thailand." *Kasetsart Journal of Social Sciences* 39 (1):33-41.

- Pardo, Antonio, and Marta Román. 2013. "Reflections on the Baron and Kenny Model Of Statistical Mediation." *Anales De Psicologia* 29 (2):614-623.
- Parke, Carol. 2013. "Module 5: Identifying and Addressing Outliers." In *Essential First Steps to Data Analysis: Scenario-Based Examples Using SPSS* edited by Carol Parke. Thousand Oaks: SAGE Publications, Inc.
- Parker, Lee, and Graeme Gould. 1999. "Changing Public Sector Accountability: Critiquing New Directions." *Accounting Forum* 23 (2):109-135. doi: 10.1111/1467-6303.00007.
- Pasandaran, Effendi. 2010. "Historical Perspective of Irrigation Management In Indonesia: Toward A New Wave of Development." ICID Seminar on History of Irrigation in Eastern Asia.
- Pasha, GR. 2002. "Selection of Variables in Multiple Regression Using Stepwise Regression." *Population* 1 (1):2.
- Patamasiriwat, D. 2012. "Fiscal Inequality and Grant Allocation: Provincial Analysis of Thailand's Local Government Finance." In *Local Governments in A Global Context*, edited by Peerasit Kamnuansilpa and Bonnie Pacala Brereton, 91-116. Thailand: College of Local Administration, Khon Kaen University.
- Patton, James M. 1992. "Accountability and Governmental Financial Reporting." *Financial Accountability & Management* 8 (3):165-180.
- Patunru, Arianto A. 2015. "Access to Safe Drinking Water and Sanitation in Indonesia." Asia & the Pacific Policy Studies 2 (2):234-244.
- PDSPK. 2016. Educationla Statistics. Jakarta: Kemdikbud.
- Pérez, Carmen Caba, Antonio M López Hernández, and Manuel Pedro Rodríguez Bolívar. 2005. "Citizens' access to on-line governmental financial information: Practices in the European Union countries." *Government Information Quarterly* 22 (2):258-276.
- Petkovska, S. 2011. "An Analysis of Financial Autonomy In Macedonian Higher Education ", University of Aveiro.
- Pickering, J, G Brokaw, P Harnden, and A Gardner. 2014. *Building High-Performance Local Governments*. Austin: River Grove Books.
- Piland, Scott G, Robert W Motl, Michael S Ferrara, and Connie L Peterson. 2003. "Evidence For The Factorial And Construct Validity of A Self-Report Concussion Symptoms Scale." *Journal of Athletic Training* 38 (2):104.
- Pina, Vicente, Patricia Bachiller, and Lara Ripoll. 2020. "Testing the Reliability of Financial Sustainability. The Case of Spanish Local Governments." *Sustainability* 12 (17):1-20.
- Playán, Enrique, Juan Antonio Sagardoy, and Rosendo Castillo. 2018. "Irrigation Governance in Developing Countries: Current Problems and Solutions." *Water* 10 (9):1118.
- Pollanen, Raili M. 2005. "Performance Measurement in Municipalities." International Journal of Public Sector Management 18 (1).
- Poniatowicz, Marzanna, Ryta Dziemianowicz, and Aneta Kargol-Wasiluk. 2020. "Good Governance and Institutional Quality of Public Sector: Theoretical and Empirical Implications." *European Research Studies Journal* XXIII (2):529-556.
- Popescu Ljungholm, Doina. 2015. "The Impact of Transparency in Enhancing Public Sector Performance." *Contemporary Readings in Law and Social Justice* VII (1):172-178.

- Potts, Tavis. 2004. "Triple Bottom Line Reporting: A Tool For Measuring, Communicating and Facilitating Change In Local Communities." Sustainability and Social Science: Round Table: Proceedings of Conference Sydney.
- Poulsen, Birgitte. 2009. "Competing Traditions of Governance and Dilemmas of Administrative Accountability: The Case of Denmark." *Public Administration* 87 (1):117-131.
- Pradeep, HUS, and Ranjith Dickwella. 2012. "Problems and Challenges Faced by the Local Government Authorities in Delivering Health Services in Sri Lanka: A Case Study of Matara Municipal Council." *International Journal of Humanities and Social Sciences.* 2 (3):239-258.
- Prasetijo, Hari. 2012. "Studi Pemberdayaan Lembaga Pengelola Jaringan Irigasi Di Tingkat Desa." *Jurnal Teknik Pengairan* 1 (1):27-36.
- Prasetyo, Iwan Agung. 2015. "Indonesia's Local Government In Risk: A Study On Post-Decentralized Local Government Finance." The Australasian Conference on Business and Social Sciences, Sydney.
- Prawitt, Douglas F, Jason L Smith, and David A Wood. 2009. "Internal audit quality and earnings management." *The Accounting Review* 84 (4):1255-1280.
- Preacher, K. J. 2012. "Monte Carlo Method For Assessing Correlations: An Interactive Tool for Creating Confidence Intervals for Correlation Coefficients [Computer Software]."
- Preacher, Kristopher J., and James P. Selig. 2012. "Advantages of Monte Carlo Confidence Intervals for Indirect Effects." *Communication Methods and Measures* 6 (2):77-98.
- Priagung, Dimas Estu, Edward K Sinoeraya, and Mafudi Mafudi. 2016. "Pengaruh Penyerapan Anggaran, Budaya Organisasi, Kepuasan Kerja Dan Pengawasan Inspektorat Terhadap Kinerja Keuangan Organisasi Publik (Studi pada Pemerintah Daerah Kabupaten Banjarnegara)." *Jurnal Ekonomi dan Bisnis Islam* 2 (1):1-15.
- Priono, Hero, Indrawati Yuhertiana, Siti Sundari, and Devy Sylvia Puspitasari. 2019. "Role of Financial Management in the Improvement of Local Government Performance." *Humanities & Social Sciences Reviews* 7 (1):77-86.
- Productivity Commission. 2008. Assessing Local Government Revenue Raising Capacity: Looking to the Future. Canberra: Productivity Commission.
- Psycharis, Yannis, Maria Zoi, and Stavroula Iliopoulou. 2016. "Decentralization and Local Government Fiscal Autonomy: Evidence from the Greek Municipalities." *Environment and Planning C: Government and Policy* 34 (2):262-280.
- Pujiono, Dodik Slamet, Hari Sukarno, and Novi Puspitasari. 2016. "Pengaruh Sistem Pengendalian Intern Terhadap Pengelolaan Keuangan Daerah Serta Kinerja Pemerintah Daerah." *BISMA: Jurnal Bisnis dan Manajemen* 10 (1):68-81.
- Pumkaew, Darunee. 2019. "Local Finance: Financial Resources Problem in Thailand." *Journal of Governance and Political Social UMA* 7 (2):148-159.
- Purba, Rasita Ekawati. 2011. "Public Participation in Development Planning: A Case Study of Indonesian Musrenbang." *International Journal of Interdisciplinary Social Sciences* 5 (12):265-277.
- Purba, Winda Sartika, Pramudya Ajeng Safitri, and Riska Andianti. 2017. *Environment Statistics of Indonesia 2017*. Jakarta: Badan Pusat Statistik.
- Purnomosidi, Lukman, Suhadak, Hermanto Siregar, and M Dzulkirom. 2014. "Mediating Effect of Financial Performance on the Influence of Size of Company, Capital Structure, Good Corporate Governance and Macro

Fundamentals on Corporate Value." *Journal of Research in Business and Management* 2 (10):41-47.

- Putri, Lovina Eka, and Khoirul Aswar. 2020. "Performance Measurement of Local Government in Indonesia: A Conceptual Study." *Information Management and Business Review* 12 (1):41-44.
- Qi, Guo You, Sai Xing Zeng, Jonathan J Shi, XH Meng, Han Lin, and QX Yang. 2014. "Revisiting the Relationship Between Environmental and Financial Performance in Chinese Industry." *Journal of environmental management* 145:349-356.
- Quinn, Andrew, and Barry R. Schlenker. 2002. "Can Accountability Produce Independence? Goals as Determinants of the Impact of Accountability on Conformity." *Personality and Social Psychology Bulletin* 28 (4):472-483.
- Rabrenović, Aleksandra. 2007. "Financial Accountability as A Condition for EU Membership." PhD, University of Glasgow.
- Rahman, Abdul, Abdul Rasyid Sahar, Fahdiansyah Putra, and Rahmadi Diliawan.
 2018. "Does Leadership Background Matter In Performance Of Local Government?" 2018 Annual Conference of Asian Association for Public Administration:" Reinventing Public Administration in a Globalized World: A Non-Western Perspective"(AAPA 2018).
- Rahman, Yustisia, Citra Hartati, M Maulana, Henri Subagiyo, and R Alam Surya Putra. 2013. Indeks Kelola Hutan dan Lahan Daerah: Kinerja Pemerintah Daerah Dalam Pengelolaan Hutan dan Lahan di Indonesia (Studi Kasus pada 9 Kabupaten). Jakarta: Sekretariat Nasional Forum Indonesia Untuk Transparasi Anggaran.
- Raman, Krishnamurthy K. 1982. "Alternative Accounting Measures As Predictors Of Municipal Financial Distress." *Journal of Accounting, Auditing and Finance* 6 (1).
- Ramos, Tomás B, Inés Alves, Rui Subtil, and João Joanaz de Melo. 2009. "The State of Environmental Performance Evaluation In The Public Sector: The Case of The Portuguese Defence Sector." *Journal of Cleaner Production* 17 (1):36-52.
- Rana, Fauzia, Arfan Ali, Waqas Riaz, and Asmara Irfan. 2019. "Impact of Accountability on Public Service Delivery Efficiency." *Journal of Public Value and Administrative Insights* 2 (1):7-9.
- Rasool, Faiz, and Charles A Ogunbode. 2015. "Socio-Demographic Differences in Environmental Concern and Willingness to Pay for Addressing Global Climate Change in Pakistan." *Asian Journal of Social Science* 43 (3):273-298.
- Recatalá, Luis, and Daniel Sacristán. 2014. "A Minimum Indicator Set For Assessing Resources Quality And Environmental Impacts At Planning Level In A Representative Area Of The European Mediterranean Region." *Ecological Indicators* 45:160-170.
- Reginato, Elisabetta, Isabella Fadda, and Aldo Pavan. 2010. "Italian Municipalities' NPFM Reforms: An Institutional Theory Perspective." *Pecvnia* (11):153-175.
- Reinikka, Ritva, and Jakob Svensson. 2005. "Fighting Corruption to Improve Schooling: Evidence from a Newspaper Campaign in Uganda." *Journal of the European economic association* 3 (2-3):259-267.
- Rejekiningrum, Popi, and Budi Kartiwa. 2018. "Institutional Development of Irrigation Management Based on The Local Wisdom in Indonesia." *Journal of Socioeconomics and Development* 1 (1):52-62.

- Renyaan, James Paul Alfred, Salim Ubud, M.S Idrus, and Djumahir. 2012. "Effect of Fiscal Autonomy and Economic Growth on Local Financial Performance (A Study on Local Government Of Papua Province)." International Journal of Business and Management Invention 1 (1).
- Resosudarmo, Ida Aju Pradnja. 2004. "Closer to People and Trees: Will Decentralisation Work for the People and the Forests of Indonesia?" *The European Journal of Development Research* 16 (1):110-132.
- Rezaee, Zabihollah, Rick Elam, and Ahmad Sharbatoghlie. 2001. "Continuous Auditing: The Audit Of The Future." *Managerial Auditing Journal* 16 (3):150-158.
- Ribot, Jesse C, Arun Agrawal, and Anne M Larson. 2006. "Recentralizing While Decentralizing: How National Governments Reappropriate Forest Resources." *World development* 34 (11):1864-1886.
- Ricks, Jacob Isaac, and Sigit Supadmo Arif. 2012. "Local Success in Encouraging Participatory Irrigation Management: Policy Lessons From Indonesia." *Journal of International Development* 24 (7):946-949.
- Risti, Lucia Camelia, Cristina Nicolaescu, and Diana Tagaduan. 2013. "Budget Deficits Effects on Economic Growth." *Journal of Economics and Business Research* 19 (1):162-170.
- Ritonga, Irwan Taufiq. 2014a. "Developing a Measure of Local Government's Financial Condition." *Journal of Indonesian Economy and Business: JIEB.* 29 (2):142.
- Ritonga, Irwan Taufiq 2014b. "Modelling Local Government Financial Conditions In Indonesia." Doctor of Philosophy, Victoria University.
- Ritonga, Irwan Taufiq, Colin Clark, and Guneratne Wickremasinghe. 2012. "Assessing Financial Condition Of Local Government In Indonesia: An Exploration." *Public and Municipal Finance* 1 (2):37-50.
- Rittenberg, Larry E., and Bradley J. Schwieger. 2005. *Auditing: Concepts For A Changing Environment.* 5th ed. Australia: Thomson/South-Western.
- Riyadi, Slamet. 2009. "Analisis Pemekaran Wilayah Dan Potensi Konflik di Kabupaten Donggala." *Academica* 1 (2).
- Robinson, Rick. 2013. *Foundations of Forensic Vocational Rehabilitation*: Springer Publishing Company.
- Rogers, Matama. 2006. "Corporate governance and financial performance of selected commercial banks in Uganda." *Makerere University Business School, Faculty of Commerce. East Africa: Kampala Uganda.*
- Rondhi, Mohammad, Ahmad Fatikhul Khasan, Yasuhiro Mori, and Takumi Kondo. 2020. "Absence of legislation and the quest for an effective mode of governance in agricultural water management: An insight from an irrigation district in central java, indonesia." *Irrigation and Drainage*.
- Rubin, Irene. 2007. "Budget Formats: Choices and Implications." In *Public Sector Governance and Accountability Series: Local Budgeting*, edited by Anwar Shah. Washington, D.C.: The World Bank.
- Rubio, Dolores Jiménez. 2011. "The Impact of Decentralization of Health Services on Health Outcomes: Evidence from Canada." *Applied Economics* 43 (26):3907-3917.
- Rungtusanatham, M., J. W. Miller, and K. K. Boyer. 2014. "Theorizing, testing, and concluding for mediation in SCM research: Tutorial and procedural recommendations." *Journal of Operations Management* 32 (3):99-113.

- Russo, Paschoal Tadeu, Claudio Parisi, Evandir Megliorini, and Claudiane Barbosa de Almeida. 2012. "Evidence of Institutionalizing Elements in The Balanced Scorecard in The Book Strategy in Action: A View Based on Institutional Theory." *Revista Contabilidade & Finanças* 23 (58):7-18.
- Rustinsyah, Rustinsyah, and Ratna A Prasetyo. 2019. "Stakeholder Engagement in A Water User Association for Agricultural Irrigation Management in the Villages in Indonesia." *Journal of Water and Land Development* 40 (1):181-191.
- Saad, Ilyas. 2001. "Indonesia's Decentralization Policy: The Budget Allocation And Its Implications For The Business Environment." The Third EUROSEAS Conference Panel on Decentralization and Democratization in Southeast Asia, London.
- Safitra, Nur. 2019. "Strategi Peningkatan Pendapatan Asli Daerah (Pad) Melalui Intensifikasi Dan Ekstensifikasi Penerimaan Pajak Dan Retribusi Daerah Di Kabupaten Enrekang." *JAKPP (Jurnal Analisis Kebijakan & Pelayanan Publik)* 1 (1):37-48.
- Sahide, Muhammad Alif K, S Supratman, A Maryudi, Y-S Kim, and L Giessen. 2016.
 "Decentralisation Policy as Recentralisation Strategy: Forest Management Units and Community Forestry in Indonesia." *International Forestry Review* 18 (1):78-95.
- Saito-Jensen, Moeko, Thomas Sikor, Y Kurniawan, Michael Eilenberg, EP Setyawan, and SJ Kustini. 2015. "Policy Options for Effective REDD+ Implementation in Indonesia: The Significance of Forest Tenure Reform." International Forestry Review 17 (1):86-97.
- Sakib, Nurul Huda. 2020. "Institutional Isomorphism." In *Global Encyclopedia of Public Administration, Public Policy, and Governance*, edited by Ali Farazmand, 1-7. Cham: Springer International Publishing.
- Salama, Aly. 2005. "A Note on the Impact of Environmental Performance on Financial Performance." *Structural Change and Economic Dynamics* 16 (3):413-421.
- Salin, Ahmad Saiful Azlin Puteh, Khairul Mizan Zakaria, and Anuar Nawawi. 2018. "The Impact of Weak Internal Controls on Fraud." 1st International Conference on Religion, Social Sciences and Technological Education, Malaysia.
- Salman, Abdulsalam Abiodun, Janneke Mostert, and Tinashe Mugwisi. 2018. "The Governance and Management of Public Library Services in Nigeria." *Library management* 39 (6-7):389-401.
- Santos, J. Řeynaldo À. . 1999. "Cronbach's Alpha: A Tool for Assessing the Reliability of Scales " *Journal of Extension* 37 (2).
- Sari, Melisa Eka, Fefri Indra Arza, and Salma Taqwa. 2019. "Pengaruh Akuntabilitas, Kesesuaian Kompensasi Dan Pengendalian Intern Terhadap Potensi Kecurangan Dana Desa." *Jurnal Eksplorasi Akuntansi* 1 (3):1443-1457.
 Sarkis, Joseph, and James J Cordeiro. 2001. "An Empirical Evaluation of
- Sarkis, Joseph, and James J Cordeiro. 2001. "An Empirical Evaluation of Environmental Efficiencies and Firm Performance: Pollution Prevention Versus End-Of-Pipe Practice." *European Journal of Operational Research* 135 (1):102-113.
- Sarr, Babacar 2015. "Credibility and Reliability of Government Budgets: Does Fiscal Transparency Matter?" *Working Papers* 5.

- Sato, Shuichi, Eiji Yamaji, and Takeshi Kuroda. 2011. "Strategies and Engineering Adaptions To Disseminate SRI Methods in Large-Scale Irrigation Systems in Eastern Indonesia." *Paddy and water environment* 9 (1):79-88.
- Satria, Yuki, Bambang Juanda, and Mamun Sarma. 2018. "Strategi Peningkatan Kualitas Belanja Daerah Provinsi Banten." *Jurnal Manajemen Pembangunan Daerah* 10.
- Schleicher, Andreas. 2018. PISA 2018: Insights and Interpretations. Paris, France: OECD.
- Schneider, Arnold, and Bryan K. Church. 2008. "The Effect Of Auditors' Internal Control Opinions On Loan Decisions." *Journal of Accounting and Public Policy* 27 (1):1-18.
- Schoemann, Alexander M., Aaron J. Boulton, and Stephen D. Short. 2017. "Determining Power and Sample Size for Simple and Complex Mediation Models." *Social Psychological and Personality Science* 8 (4):379-386.
- Scott, W Richard. 1987. "The Adolescence of Institutional Theory." Administrative science quarterly:493-511.
- Scott, W Richard. 2008. "Lords of the Dance: Professionals as Institutional Agents." Organization Studies 29 (2):219-238.
- Scott, W Richard. 2013. Institutions and Organizations. California: SAGE Publications.
- Scott, W Richard 2005. "Institutional Theory: Contributing To A Theoretical Research Program." In *Great Minds in Management: The Process of Theory Development*, edited by K Smith and M Hitt. Oxford: Oxford University Press.
- Scott, W. Richard, and John W. Meyer. 1991. "The Organization of Societal Sectors: Propositions and Early Evidence." In *The New Institutionalism in Organizational Analysis*, edited by Walter W. Powell and oPaul J. Dimaggi. London, UK: University of Chicago Press.
- Scutariu, Adrian Liviu, and Petronela Scutariu. 2015. "The Link Between Financial Autonomy and Local Development. The Case of Romania." *Procedia Economics and Finance* 32:542-549.
- SDA. 2018. Program Pengembangan dan Pengelolaan Sistem Irigasi di Indonesia. Jakarta: Direktorat Jenderal Sumber Daya Air, Kementerian Pekerjaan Umum dan Perumahan Rakyat.
- SDA. 2019. *Modul Kelembagaan Pengelola Irigasi*. Jakarta: Direktorat Jenderal Sumber Daya Air, Kementerian Pekerjaan Umum dan Perumahan Rakyat.
- Sekaran, Uma, and Roger Bougie. 2010. *Research Methods For Business: A Skill-Building Approach*. 5th ed. Chichester: Wiley.
- Sekiguchi, Shunsuke. 2019. "An Analysis of the Efficiency of Local Government Expenditure and the Minimum Efficient Scale in Vietnam." *Urban Science* 3 (3):77.
- Seyoum, Belay, and Terrell G Manyak. 2009. "The Impact of Public and Private Sector Transparency on Foreign Direct Investment in Developing Countries." *Critical perspectives on international business.*
- Shah, Anwar. 2006. *Local Governance in Developing Countries*. Washington: The International Bank for Reconstruction and Development.
- Shah, Anwar, and Sana Shah. 2006. "The New Vision of Local Governance and the Evolving Roles of Local Governments." In *Local Governance in Developing Countries*, edited by Anwar Shah. Washington, D.C: The International Bank for Reconstruction and Development.

- Shahib, Habib Muhammad, Eko Ganis Sukoharsono, M Achsin, and Yeney Widya Prihatiningtias. 2020. "Developing Local Government's Socioenvironmental Accountability: Insights from Indonesian Socioenvironmental NGOs' Annual Reports." In *Environmentalism and NGO accountability*. Emerald Publishing Limited.
- Shen, Li-Yin, J Jorge Ochoa, Mona N Shah, and Xiaoling Zhang. 2011. "The Application Of Urban Sustainability Indicators–A Comparison Between Various Practices." *Habitat International* 35 (1):17-29.
- Shirali, Gholamabbas, Mohammad Shekari, and Kambiz Ahmadi Angali. 2018. "Assessing Reliability And Validity Of An Instrument For Measuring Resilience Safety Culture In Sociotechnical Systems." Safety and Health at Work 9 (3):296-307.
- Shrestha, Sangam, Foyya Yusufu Aquino, and Vishnu P Pandey. 2014. "Performance evaluation of the Telagasari Irrigation Scheme (TIS) of Karawang Regency, Indonesia." *Journal of Food, Agriculture & Environment* 12 (3&4):187-197.
- Signé, Landry, and Koiffi Korha. 2016. "Horizontal Accountability and the Challenges for Democratic Consolidation in Africa: Evidence from Liberia." *Democratization* 23 (7):1254-1271.
- Simangunsong, Rosma. 2014. "The Impact of Internal Control Effectiveness and Internal Audit Role Toward the Performance of Local Government." *Research Journal of Finance and Accounting* 5 (7):50-58.
- Simatupang, Rentanida Renata. 2009. "Evaluation of decentralization outcomes in Indonesia: Analysis of health and education sectors."
- Simson, Rebecca, and Bryn Welham. 2014. "Incredible Budgets: Budget Credibility In Theory And Practice." *Working and Discussion Papers* 400.
- Sinervo, Lotta-Maria. 2020. "Financial Sustainability of Local Governments in the Eyes of Finnish Local Politicians." *Sustainability* 12 (23):1-16.
- Singer, B. 2016. "Financing Sustainable Forest Management in Developing Countries: The Case For A Holistic Approach." *International Forestry Review* 18 (1):96-109.
- Siombo, Marhaeni Ria. 2014. "Tanggung Jawab Pemda terhadap Kerusakan Lingkungan Hidup Kaitannya dengan Kewenangan Perizinan di Bidang Kehutanan dan Pertambangan." *Jurnal Dinamika Hukum* 14 (3):394-405.
- Sitek, Paweł. 2014. "The Idea Of The Local Government's The Autonomy And Financial Independence." *Law Journal Library* 17.
- Skoufias, Emmanuel, AmbDasgupta Narayan, Basab, and Kai Kaiser. 2011. "Electoral Accountability, Fiscal Decentralization and Service Delivery in Indonesia." *Policy Research Working Paper* WPS5614.
- Slukhai, Sergii. 2003. *Dilemmas and Compromises: Fiscal Equalization in Transition Countries*. Budapest, Hungary: Local Government and Public Service Reform Intiative, Open Society Institute.
- Smedby, Nora, and Maj-Britt Quitzau. 2016. "Municipal Governance and Sustainability: The Role of Local Governments in Promoting Transitions." *Environmental Policy and Governance* 26 (5):323-336.
- Smith, Alexander K, John Z Ayanian, Kenneth E Covinsky, Bruce E Landon, Ellen P McCarthy, Christina C Wee, and Michael A Steinman. 2011. "Conducting High-Value Secondary Dataset Analysis: An Introductory Guide And Resources." *Journal of General Internal Medicine* 26 (8):920-929.

- Smith, Malcolm. 2005. Performance Measurement And Management: A Strategic Approach To Management Accounting. London: SAGE Publications Ltd.
- Smith, Malcolm, Normah Haji Omar, Syed Iskandar Zulkarnain Sayd Idris, and Ithnahaini Baharuddin. 2005. "Auditors' Perception Of Fraud Risk Indicators: Malaysian Evidence." *Managerial Auditing Journal* 20 (1):73-85.
- Smoke, P, and G Loffler. 2013. The Role of Decentralization/Devolution in Improving Development Outcomes at the Local Level: Review of the Literature and Selected Cases. New York.
- Soenen, Luc. 2003. "Cash Holdings: A Mixed Blessing?" AFP Exchange 5:54-57.
- Sofyani, Hafiez, Hosam Alden Riyadh, and Heru Fahlevi. 2020. "Improving Service Quality, Accountability and Transparency of Local Government: The Intervening Role of Information Technology Governance." *Cogent Business & Management* 7 (1):1-20.
- Solomon, J. 2013. *Corporate Governance and Accountability*. Fourth ed. Chichester, West Sussex, United Kingdom: John Wiley and Sons, Inc.
- Sorunke, Olukayode Abayomi. 2016. "Internal Audit and Fraud Control in Public Institutions in Nigeria: A Survey of Local Government Councils in Osun State." *International Journal of Academic Research in Business and Social Sciences* 6 (2):153-158.
- Soto, Victoria Eugenia, Maria Isabel Farfan, and Vincent Lorant. 2012. "Fiscal Decentralisation and Infant Mortality Rate: the Colombian Case." *Social Science & Medicine* 74 (9):1426-1434.
- Spangenberg, Joachim H. 2004. "Reconciling Sustainability and Growth: Criteria, Indicators, Policies." *Sustainable development* 12 (2):74-86.
- Sri, Lestari, and Utama Made Suyana. 2019. "The Effect of Local Government Own Revenue And Revenue Sharing Funds on Economic Performance and Community Welfare Through Capital Expenditure of Regency/City in Bali Province, Indonesia." *Russian Journal of Agricultural and Socio-Economic Sciences* 91 (7).
- Stål, Herman. 2017. "Institutional Maintenance Within the Regulative Pillar." Academy of Management (AOM) Meeting.
- Stanwick, Peter A., and Sarah D. Stanwick. 1998. "The Relationship between Corporate Social Performance, and Organizational Size, Financial Performance, and Environmental Performance: An Empirical Examination." *Journal of Business Ethics* 17 (2):195-204.
- Statistics Indonesia. 2014. *Statistical Yearbook of Indonesia 2014*. Jakarta: Badan Pusat Statistik.
- Steccolini, Ileana. 2004. "Is the Annual Report an Accountability Medium? An Empirical Investigation into Italian Local Governments." *Financial Accountability & Management* 20 (3):327-350.
- Stine, Robert A., and Dean P. Foster. 2011. *Statistics for Business: Decision Making And Analysis*. International ed. Boston: Addison-Wesley.
- Stockburger, David. 2001. Introductory Statistics: Concepts, Models, And Applications: Atomic Dog Pub.
- Streib, Gregory D, and Theodore H Poister. 1999. "Assessing the Validity, Legitimacy, and Functionality of Performance Measurement Systems in Municipal Governments." *The American Review of Public Administration* 29 (2):107-123.
- Štreimikienė, Dalia. 2015. "Environmental Indicators for The Assessment of Quality of Life." *Intelektinė ekonomika* 9 (1):67-79.

Studenmund, A. H. author. 2014. Using econometrics: a practical guide. Sixth, Pearson new international ed. Harlow: Pearson Education.

- Styles, Alan K., and Mack Tennyson. 2007. "The accessibility of financial reporting of u.s. municipalities on the internet." *Journal of Public Budgeting, Accounting & Financial Management* 19 (1):56-92.
- Subhan, Subhan. 2019. "Governance: Perubahan Paradigma Pada Penyelenggaraan Pemerintahan." SOSFILKOM 13 (01):33-47.
- Sudsomboon, Seerungrat, and Phapruke Ussahawanitchakit. 2009. "Professional Audit Competencies: The Effects On Thai's CPAS Audit Quality, Reputation, And Success." *Review of Business Research* 9 (3):66-85.
- Suhardi, Didik. 2017. *Rangkuman Statistik Pendidikan Dasar dan Menengah* 2017/2018. Jakarta: Pusat Data dan Statistik Pendidikan dan Kebudayaan.
- Sujarwoto, Sujarwoto. 2012. "Political decentralization and local public services performance in Indonesia." *Journal of Public Administration and Governance* 2 (3):55-84.
- Sukandar, Sukandar, Masimin Masimin, and Eldina Fatimah. 2018. "Sosial Kapital Masyarakat Dalam Operasional Dan Pemeliharaan Daerah Irigasi Krueng Aceh." *Jurnal Teknik Sipil* 1 (4):995-1004.
- Sukasni, Agnes, and Hady Efendy. 2017. "The Problematic of Education System In Indonesia And Reform Agenda." *International Journal of Education* 9 (3):183-199.
- Sulistyono, Djoko, Deden Nuryadin, and Anung S Hadi. 2014. "Evaluasi Tim Penegasan Batas Daerah (Studi Kasus di Provinsi Lampung dan Kalimantan Timur)." *Jurnal Bina Praja: Journal of Home Affairs Governance* 6 (1):53-64.
- Sun, Weizeng, Yuming Fu, and Siqi Zheng. 2017. "Local Public Service Provision and Spatial Inequality in Chinese Cities: The Role of Residential Income Sorting and Land-Use Conditions." *Journal of Regional Science* 57 (4):547-567.
- Sururi, Ahmad. 2020. "Performance the Implementation of Public Services the Subdistrict Government in Cilegon City." *ARISTO* 8 (1):137-164.
- Suryaningtyas, Arthy, and Abdul Rohman. 2019. "Pengaruh Penerapan Corporate Governance Terhadap Nilai Perusahaan Dengan Kinerja Keuangan Sebagai Variabel Mediasi." *Diponegoro Journal of Accounting* 8 (4).
- Suwandi, Luthful Hakim, Ongki Wiratno, Aulia Azhar Abdurachman, and M Ade Supriyatna. 2017. *Statistik Lahan Pertanian Tahun 2012-2016*. Jakarta: The Ministry of Agriculture.
- Svara, James H. 2008. "Strengthening Local Government Leadership and Performance: Reexamining and Updating the Winter Commission Goals." *Public Administration Review* 68:S37-S49.
- Swank, Jacqueline M, and Patrick R Mullen. 2017. "Evaluating Evidence For Conceptually Related Constructs Using Bivariate Correlations." *Measurement and Evaluation in Counseling and Development* 50 (4):270-274.
- Syaukat, Yusman, Fitria Nur Arifah, and Fahma Minha. 2014. "Economic Value ad Service Fee of Irrigation Water in The Districts Bogor and Kudus, Indonesia." *Journal of the International Society for Southeast Asian Agricultural Sciences* 20 (2):157-172.
- Tabachnick, Barbara G., and Linda S. Fidell. 2014. Using Multivariate Statistics. Sixth, Pearson new international ed. Harlow: Pearson Education Limited.

- Tabassum, Naeem, and Satwinder Singh. 2020. Corporate Governance and Organisational Performance: The Impact of Board Structure. Switzerland: Palgrave Macmillan.
- Taber, Keith S. 2018. "The Use of Cronbach's Alpha When Developing and Reporting Research Instruments in Science Education." Research in Science Education 48 (6):1273-1296.
- Tacconi, Luca. 2007. "Decentralization, Forests and Livelihoods: Theory and Narrative." Global environmental change 17 (3-4):338-348.
- Tackie, George, Edward Marfo-Yiadom, and Sampson Oduro Achina. 2016. "Determinants of Internal Audit Effectiveness in Decentralized Local Government Administrative Systems." International Journal of Business and Management 11 (11):184-195.
- Takahashi, Hiroyuki. 1999. Fiscal Crises in Japan's Prefectures and The Debate on Corporate Tax Reform. Washington, DC: Japan Economic Institute of America, Inc.
- Tamawiwy, Julius, Jullie J Sondakh, and Jessy DL Warongan. 2016. "Pengaruh Kinerja Keuangan Pemerintah Daerah Terhadap Belanja Modal Untuk Pelayanan Publik (Studi pada kabupaten dan kota di Provinsi Sulawesi Utara)." Jurnal Riset Akuntansi dan Auditing 7 (2).
- Tambulasi, Richard I. C. 2007. "Who Is Fooling Who?: New Public Management-Oriented Management Accounting and Political Control in the Malawi's Local Governance." Journal of Accounting and Organizational Change 3 (3):302-328.
- Tampubolon, Endy Grade. 2019. "Pengaruh Belanja Modal Terhadap Kesejahteraan Masyarakat." Sosio e-Kons 11 (1):79-89.
- Tavares, António F. 2017. "Ten Years After: Revisiting the Determinants of The Adoption of Municipal Corporations for Local Service Delivery." Local Government Studies 43 (5):697-706.
- Tejedo-Romero, Francisca, and Joaquim Filipe Ferraz Esteves de Araujo. 2018. "Determinants of Local Governments' Transparency in Times of Crisis: Evidence From Municipality-Level Panel Data." Administration & Society 50 (4):527-554.
- The World Bank. 2003. The World Development Report 2004: Making Services Work for Poor People. Washington,: he International Bank for Reconstruction and Development.
- Thomas, Ian G. 2010. "Environmental Policy and Local Government in Australia." Local Environment 15 (2):121-136.
- Thompson, Paul N. 2017. "Effects of Fiscal Stress Labels on Municipal Government Finances, Housing Prices, and The Quality of Public Services: Evidence from Ohio." Regional Science and Urban Economics 64:98-116.
- Tibor, Gabriel J, Neil S Grigg, Otto J Helweg, Herve Plusquellec, Charles M Burt, John L Merriam, and Phillip Z Kirpich. 2000. "Problems of Irrigation in Developing Countries." Journal of Irrigation and Drainage Engineering 126 (3):195-202.
- Timoshenko, A. 2018. "Ensuring the Management Efficiency for the Budget Resources Allocation in Ukraine." *Modern Economics* 7. Tkachenko, Liudmila. 2020. "Public Finance Management: Challenges and
- Opportunities." Athens Journal of Business & Economics 6 (1):73-98.
- Todaro, Michael P., and Stephen C. Smith. 2006. Economic Development. 9th ed. Boston: Pearson Addison Wesley.

- Tolley, Sukma Selviany, Ridwan, and Muh.Yunus Kasim. 2017. "Pengaruh Sistem Pengendalian Intern Pemerintah Terhadap Kinerja Satuan Kerja Perangkat Daerah Kabupaten Sigi." *Katalogis* 5 (2):179-190.
- Torres, Lourdes, Vicente Pina, and Ana Yetano. 2011. "Performance Measurement in Spanish Local Governments. A Cross-Case Comparison Study." *Public administration* 89 (3):1081-1109.
- Tran, Liem. 2016. "An Interactive Method To Select A Set Of Sustainable Urban Development Indicators." *Ecological Indicators* 61:418-427.
- Tran, Quoc Trung. 2019. "Corporate Cash Holdings and Financial Crisis: New Evidence From an Emerging Market." *Eurasian Business Review*:1-15.
- Trinirmalaningrum, Nurdiyansah Dalidjo, Frans R Siahaan, Untung Widyanto, and Ivan Aulia Achsan. 2015. *Di Balik Tragedi Asap: Catatan Kebakaran Hutan dan Lahan*.
- Tripathi, Satya S. 2015. *Forest Ecosystem Valuation Study: Indonesia*. Jakarta: United Nations Office for REDDCoordination in Indonesia (UNORCID).
- Tsoutsoura, Margarita. 2004. "Corporate social responsibility and financial performance."
- Uche, Ada. 2014. "Analysis of Local Government Performance and Leadership in Nigeria." *Africa's Public Service Delivery & Performance Review* 2 (4):130-150.
- UCLA. 2006. What Does Cronbach's Alpha Mean.
- Udeh, Sergius Nwannebuike, and EO Nwadialor. 2016. "Evaluation of Effectiveness of Internal Audit in the Nigerian Public Sector." *European Journal of Business, Economics and Accountancy* 4 (3).
- Uhaib, Muhammad, and William Djani. 2018. "Indonesia Public Administration Reform in Supporting Good Governance." *Review of Public Administration and Management* 6 (3).
- Ujkani, Sead, and Nexhmie Berisha Vokshi. 2019. "An Overview on the Development of Internal Control in Public Sector Entities: Evidence from Kosovo." *International Journal of Economics and Business Administration* VII (4):320-335.
- UN-HABITAT. 2015. The Challenge of Local Government Financing in Developing Countries. Nairobi: United Nations Human Settlements Programme.
- UNDP. 2014. Human Development Report 2014 Sustaining Human Progress: Reducing Vulnerabilities and BuildingResilience. New York, USA.: UNDP.
- Usang, Obal Usang Edet, and Basariah Salim. 2018. "The Relationship Between Institutional Environment, Internal Audit and Performance of Local Governments in Nigeria." 4th ICAN International Academic Conference Proceedings, Nigeria.
- Utami, Sri. 2018. "Angka Kematian Ibu dan Bayi Indonesia, 10 Negara Tertinggi di Dunia." *Media Indonesia*.
- Van de Walle, Steven. 2006. "The State of The World's Bureaucracies." *Journal of Comparative Policy Analysis* 8 (4):437-448.
- Van de Walle, Steven, and Geert Bouckaert. 2003. "Public Service Performance And Trust In Government: The Problem Of Causality." *International Journal of Public Administration* 26 (8-9):891-913.
- Van der Waldt, Gerrit. 2004. *Managing Performance in the Public Sector: Concepts, Considerations and Challenges*: Juta and Company Ltd.
- Van Doeveren, Veerle. 2011. "Rethinking Good Governance: Identifying Common Principles." *Public Integrity* 13 (4):301-318.

- Veeraraghavan, K. 2018. "Effect of Financial Management Practices on Financial Performance of Small and Medium Enterprises in Pudhucherry, India." International Journal of Management Studies V (4).
- Verma, Pramit, and AS Raghubanshi. 2018. "Urban Sustainability Indicators: Challenges and Opportunities." *Ecological Indicators* 93:282-291.
- Vermillion, Douglas L, Madar Samad, Suprodjo Pusposutardjo, Sigit S Arif, and Saiful Rochdyanto. 2000. An Assessment of The Small-Scale Irrigation Management Turnover Program In Indonesia. Colombo: International Water Management Institute.
- Vijayakumar, AN, and N Nagaraja. 2012. "Internal Control Systems: Effectiveness of Internal Audit in Risk Management at Public Sector Enterprises." *BVIMR Management Edge* 5 (1).
- Viswanathan, S, Kapil Verma, Yancy Toh, and Mr Ong Boon Hwee. 2020. Understanding Corporate Stewardship, Governance and Sustainability. Singapore: Stewardship Asia Centre CLG Limited.
- Waddock, Sandra A, and Samuel B Graves. 1997. "The Corporate Social Performance–Financial Performance Link." *Strategic management journal* 18 (4):303-319.
- Wagner, Marcus, Nguyen Van Phu, Théophile Azomahou, and Walter Wehrmeyer. 2002. "The relationship between the environmental and economic performance of firms: an empirical analysis of the European paper industry." *Corporate Social Responsibility and Environmental Management* 9 (3):133-146.
- Wahba, Hayam, and Khaled Elsayed. 2015. "The Mediating Effect of Financial Performance on the Relationship Between Social Responsibility and Ownership Structure." *Future Business Journal* 1 (1-2):1-12.
- Wahid, Fathul, and Maung K Sein. 2013. "Institutional Entrepreneurs: The Driving Force in Institutionalization of Public Systems in Developing Countries." *Transforming Government: People, Process and Policy* 7 (1):76-92.
- Wang, Xiaohu. 2002. "Assessing Administrative Accountability: Results From A National Survey." *The American Review of Public Administration* 32 (3):350-370.
- Wang, Xiaohu, Lynda Dennis, and Yuan Sen Jeff Tu. 2007. "Measuring Financial Condition: A Study of US States." *Public Budgeting & Finance* 27 (2):1-21.
- Wang, Xiaohu, Montgomery Van Wart, and Nick Lebredo. 2014. "Sustainability Leadership in A Local Government Context: The Administrator's Role in the Process." *Public Performance & Management Review* 37 (3):339-364.
- Wanjau, Boniface Muriithi, Willy Mwangi Muturi, and Patrick Ngumi. 2018. "Effect of Financial Transparency on Financial Performance of Companies Listed in East Africa Securities Exchanges " Research Journal of Finance and Accounting 9 (4).
- Wardhani, Ratna, Hilda Rossieta, and Dwi Martani. 2017. "Good Governance and the Impact of Government Spending on Performance of Local Government in Indonesia." *International Journal of Public Sector Performance Management* 3 (1):77-102.
- Wardhono, Adhitya, Yulia Indrawati, and Ciplis Gema Qori`ah. 2013. "Kajian Pemetaan Dan Optimalisasi Potensi Pajak Dalam Rangka Meningkatkan Pendapatan Asli Daerah (Pad) Di Kabupaten Jember." *J@ti Undip* 7 (2):69-76.

- Wawanudin, Wawanudin, and Rohidin Sudarno. 2018. "Pelaksanaan Sistem Merit dalam UU ASN, Wewenang KASN dan Analisis Peraturan Perundang-Undangan yang Berpengaruh terhadap Wewenang KASN." *Mozaik* 10 (1):267959.
- Welch, Richard. 2002. "Legitimacy of Rural Local Government in the New Governance Environment." *Journal of rural studies* 18 (4):443-459.
- Welham, Bryn, Philipp Krause, and Edward Hedger. 2013. *Linking PFM Dimensions To Development Priorities*. London: Overseas Development Institute.
- Wendling, Zachary A, John W Emerson, Daniel C Esty, Marc A Levy, and Alex de Sherbinin. 2018. 2018 Environmental Performance Index. New Haven, CT: Yale Center for Environmental Law & Policy.
- Whitley, Bernard E, and Mary E Kite. 2013. *Principles of Research In Behavioral Science*. New York: Routledge.
- Wibowo, Galih. 2020. "Manajemen Inovasi: Promosi Dalam Mendukung Sistem Merit Di BKD Provinsi Jawa Barat." *Jurnal Manajemen Publik & Kebijakan Publik* (*JMPKP*) 2 (1):1-10.
- Wiharani, Annisa Paramita, and Ronald Holzhacker. 2016. "Environmental Governance by Transnational Municipal Networks: The Case of Indonesian Cities." In *Decentralization and Governance in Indonesia*, 201-229. Springer.
- Wijayati, Nureni, Niels Hermes, and Ronald Holzhacker. 2016. "Corporate Governance and Corruption: A Comparative Study of Southeast Asia." In *Decentralization and Governance In Indonesia*, edited by Ronald L Holzhacker, Rafael Wittek and Johan Woltjer, 259-292. Switzerland: Springer International Publishing.
- Williams, Belinda. 2015. "The Local Government Accountants' Perspective on Sustainability." *Sustainability Accounting, Management and Policy Journal.*
- Williams, Belinda, Trevor Wilmshurst, and Robert Clift. 2011. "Sustainability Reporting By Local Government in Australia: Current and Future Prospects." Accounting Forum.
- Williams, Brett, Andrys Onsman, and Ted Brown. 2010. "Exploratory Factor Analysis: A Five-Step Guide For Novices." *Australasian Journal of Paramedicine* 8 (3).
- Williams, Daniel W. 2004. "Evolution of Performance Measurement Until 1930." Administration & Society 36 (2):131-165.
- Winarni, Esti, Abdul Aziz Ahmad, and Suharno Suharno. 2019. "Community Participation of Development Planning in Purbalingga Regency." *ICORE* 5 (1).
- Wirawan, Ricky, Mardiyono, and Ratih Nurpratiwi. 2015. "Partisipasi Masyarakat Dalam Perencanaan Pembangunan Daerah." *JISIP: Jurnal Ilmu Sosial Dan Ilmu Politik* 4 (2).
- Witing, M. 2006. "Relations Between Organizational Identity, Identification and Organizational Objectives." *Journal of Managerial Psychology* 21 (7):432-452.
- Wollenberg, Eva, Moira Moeliono, Godwin Limberg, Ramses Iwan, Steve Rhee, and Made Sudana. 2006. "Between State and Society: Local Governance of Forests in Malinau, Indonesia." *Forest Policy and Economics* 8 (4):421-433.
- Wood, Anthony, and Natalya Brathwaite. 2013. "Internal Controls in the Retail Sector: A Case Study of a Leading Supermarket in Barbados." *International Journal of Arts and Commerce* 2 (11):1-16.
- Wood, Michael. 2005. "Bootstrapped Confidence Intervals as an Approach to Statistical Inference." *Organizational Research Methods* 8 (4):454-470.
World Bank. 2010. Laporan Penelitian Dana Transfer Pusat ke Daerah: Penyempurnaan Grand Design Desentralisasi Fiskal. Jakarta: World Bank.

- Wulandari, Eva, Muhamad Wahyudi, and Utpala Rani. 2018. "Effect of Original Local Government Revenues, General Allocation Funds, Special Allocation Funds, Share Funds, Other Legal Revenues, Budget Surplus/Deficit to Human Development Index Through Capital Expenditures: Case Study of Regencies/Municipalities in Indonesia." *Review of Integrative Business and Economics Research* 7:125-137.
- Yan, Xin, and Xiaogang Su. 2009. *Linear Regression Analysis: Theory and Computing*: World Scientific Publishing Company Pte Limited.
- Yasa, I Nyoman Mahaendra, and Made Suyana Utama. 2015. "Dampak Kinerja Keuangan Daerah Terhadap Kesejahteraan Masyarakat Kabupaten/Kota di Provinsi Bali." *E-Jurnal Ekonomi dan Bisnis Universitas Udayana* 4:44748.
- Yasin, Muhammad. 2020. "Analysis of Regional Original Revenues And Routine Expenditures on Regional Financial Performance in East Java Regencies and Cities." *Jurnal Mantik* 3 (4):64-69.
- Yemer, Mesfin, and Firew Chekol. 2017. "The Effect Of Internal Controls Systems On Hotels Revenue. A Case Of Hotels In Bahir Dar And Gondar Cities." *Arabian Journal of Business and Management Review (Oman Chapter)* 6 (6):19.
- Yin, Haitao, and Peter J. Schmeidler. 2009. "Why Do Standardized ISO 14001 Environmental Management Systems Lead To Heterogeneous Environmental Outcomes?" *Business Strategy and the Environment* 18 (7):469-486.
- York, Richard, Eugene A Rosa, and Thomas Dietz. 2003. "STIRPAT, IPAT and ImPACT: Analytic Tools for Unpacking the Driving Forces of Environmental Impacts." *Ecological Economics* 46 (3):351-365.
- Yung, Kenneth, DeQing Diane Li, and Yi Jian. 2015. "The Value of Corporate Financial Flexibility in Emerging Countries." *Journal of Multinational Financial Management* 32:25-41.
- Yurniwati, and Afdhal Rizaldi. 2015. "Control Environment Analysis at Government Internal Control System: Indonesia Case." *Procedia-Social and Behavioral Sciences* 211:844-850.
- Zafra-Gómez, José Luis, Antonio Manuel López-Hernández, and Agustín Hernández-Bastida. 2009a. "Developing A Model to Measure Financial Condition in Local Government: Evaluating Service Quality and Minimizing the Effects of the Socioeconomic Environment: An Application to Spanish Municipalities." *The American Review of Public Administration* 39 (4):425-449.
- Zafra-Gómez, José Luis, Antonio Manuel López-Hernández, and Agustín Hernández-Bastida. 2009b. "Evaluating Financial Performance In Local Government: Maximizing The Benchmarking Value." *International Review of Administrative Sciences* 75 (1):151-167.
- Zagorchev, Andrey, and Lei Gao. 2015. "Corporate Governance and Performance of Financial Institutions." *Journal of Economics and Business* 82:17-41.
- Zakerian, Seyed Abolfazl, and Indra Devi Subramaniam. 2009. "The Relationship Between Psychosocial Work Factors, Work Stress and Computer-Related Musculoskeletal Discomforts Among Computer Users in Malaysia." International Journal of Occupational Safety and Ergonomics 15 (4):425-434.
- Zeithaml, Valarie A, Mary Jo Bitner, and Dwayne D Gremler. 2010. "Services Marketing Strategy." Wiley International Encyclopedia of Marketing.

- Zeng, A.M., C. Zhang, and Z.H. Wei. 2013. "The Impact of Financial Crisis, Flexible Financial Reserve and Investment Behavior of Enterprises—Empirical Evidence from Listed Companies in China." *Management World* 4:107-120.
- Zhang, Haifeng, Zhuo Zhang, and Ekaterina Steklova. 2020. "Do Companies Need Financial Flexibility for Sustainable Development?" *Sustainability* 12 (5):1811.
- Zhang, Zhiyong. 2014. "Monte Carlo Based Statistical Power Analysis for Mediation Models: Methods and Software." *Behavior Research Methods* 46 (4):1184-1198.
- Zhou, Kuo, Baicheng Zhou, and Mengmeng Yu. 2020. "The Impacts of Fiscal Decentralization on Environmental Innovation in China." *Growth and Change* 00:1-21.

Appendix

Appendix Linearity and Homoscedasticity





Partial Regression Plot Dependent Variable: Service Performance









Financial Transparency

Partial Regression Plot



Partial Regression Plot Dependent Variable: Environmental Performance .400 G .200 Q Environmental Performance .000 G Ð 5 -.200 -.400 -.600 Internal Control -.400 -.200 .200 .400

Partial Regression Plot





